

An investigation into the influences of catchment nutrient export and climatic effects on the trophic status of a small inter-drumlin lake, Milltown Lake, Co. Monaghan

By

Vicky Veerkamp, B.Sc. (Hons) Analytical Science

2020

A thesis presented to the School of Health and Science, Dundalk Institute of Technology in fulfillment of the requirement of the degree of Ph. D.

School of Health and Science, Centre for Freshwater and Environmental Studies

Dundalk Institute of Technology

We, the undersigned declare that this thesis entitled, <u>"An investigation into the influences of catchment nutrient export and climatic effects on the trophic status of a small inter-drumlin lake, Milltown Lake, Co. Monaghan"</u>, is entirely the author's own work and has not been taken from the work of others, except as cited and acknowledged within the text.

The thesis has been prepared according to the regulations of Dundalk Institute of Technology and has not been submitted in whole or in part for an award in this or any other institution.

Author Na	me: Vicky	Veerkamp	
Author Sig	gnature:	Vicky Vecekomp	
Date:22	/08/2019		

Supervisor Name: Eleanor Jennings	
Supervisor Signature:	
Date:	

This research project is supervised by:

Dr Eleanor Jennings

Principal Investigator Department of Applied Science Dundalk Institute of Technology Dundalk, Co. Louth.

Dr Suzanne Linnane

Principal Investigator Department of Applied Science Dundalk Institute of Technology Dundalk, Co. Louth.

Acknowledgements

Firstly, I would like to thank my supervisors, Dr Eleanor Jennings and Dr Suzanne Linnane, for all their guidance, enthusiasm and endless support. I have gained valuable knowledge and experience from them both over the last number of years. Many thanks to my all my past and present colleagues in the Centre for Freshwater and Environmental Studies, for their help, expertise and assistance with both field and laboratory work. I would like to acknowledge the work carried out by bursary students Karen Doherty, Shannon Dixon and Thomas Curran under the supervision of Dr Valerie Mc Carthy, Centre for Freshwater and Environmental Studies, Dundalk Institute of Technology, for their help in the enumeration of phytoplankton and zooplankton in Chapter 6.

I would like to thank the following people and institutes for all their field and technical support particularly in the loan of equipment and with the installation of the thermistor chain and sensor platform in Milltown Lake; the Staff at Maine Institute, Furnace, Newport, Co. Mayo, Dr Valerie Mc Carthy, Mr Antoin Lawlor, Dr Norman Allott, Dr Joerg Arnscheidt (University of Ulster), Dr Timmothy Sullivan and Dr Fiona Reagan (Dublin City University). This research was supported by Dundalk Institute of Technology Masters Scholarship Fund and benefited from participation in the Global Lake Ecological Observatory Network (GLEON).

Finally, a heartfelt thanks to my wonderful family, Dad, Mum, Paddy, Robyn and Jake for your unending love and support throughout this process. I know there were many times when you were prepared to strangle me, but I am ever so grateful you didn't. Last but not least, thanks to my wonderful partner Patrick, your love, kindness and never-ending support and encouragement is what kept me going, I know without it I would never have gotten this far.

Abstract

Eutrophication is an on-going problem in lakes that are nutrient enriched. In addition, lakes are now subject to changes in the local weather, which is linked to climate change. Understanding the way that climate affects nutrient export and changes within lakes is critical if lakes are to be improved and protected in the future. This thesis has examined this by looking at number of different factors, which include: (1) the role of nutrients exported from the surrounding landscape and how agricultural mitigation measures (e.g. comparing a fenced stream with a vegetative buffer strips to an unfenced stream which had no fencing installed), can be used to decrease nutrient loss from the land to water, (2) examining local climatic factors and their role, either directly or indirectly, in influencing lake biota, particularly in lakes where nutrient supplies are not limiting, and finally (3) determining how phytoplankton richness is influenced by changes in water column stability during the onset of spring stratification. The study within the catchment found net retention of nutrients (TP and TN) occurred during the Grazing season (cattle grazing in the field) in the fenced stream reach, while during the Open season (slurry spreading permitted, but cattle still indoors) the reverse was seen for TP only, suggesting that different processes were controlling export of nutrients in both the fenced and unfenced reach. Despite the net TP release in the Open season, the estimated TP and TN loads on an annual basis were lower for the fenced stream, compared to that of the unfenced stream. Regarding local climatic factors, results showed that in the epilimnion of Milltown Lake, surface nutrient loading (mainly P) and light were more significant than internal lake nutrient resuspension, while N availability was more influential in controlling the standing crop of phytoplankton at the lower lake depths. In general, catchment management strategies are devised to reduce nutrient loads from the surrounding catchment in order to protect and improve catchment waterbodies, which this study has shown to be achieved through the implementation of streamside fencing with a vegetative buffer strip. However, this study has also shown that more attention is needed in understanding the internal lake sediment nutrient release, should catchment load be reduced, it is this legacy source of nutrients that is likely to play a significant role in determining a lake's trophic status.

Table of contents

Acknow	ledgements	iv
Abstrac	t	v
List of a	bbreviations	ix
List of t	ables	xi
List of f	igures	xii
Chapter	1: General introduction	1
1.1	Introduction	1
1.2	Aims and objectives	5
1.3	Knowledge gaps	6
1.4	Thesis structure	7
1.5	Hypotheses	7
Chapter	2: Literature review	9
2.1	Sources of phosphorus and nitrogen in lakes	9
2.1	1 Phosphorus loss from surrounding agricultural land	9
2.1	2 Nitrogen loss from surrounding agricultural land	12
2.1	.3 Internal nutrient loading from phosphorus and nitrogen to lakes	15
2.2	Stratification of lakes	17
2.3	The effects of wind on the breakdown of thermal stratification	19
2.4	Effects of light on phytoplankton groups	20
2.5	Effects of nutrients on phytoplankton groups	20
2.6	Chlorophyll <i>a</i> as a proxy for phytoplankton biomass	21
2.7	Combined effects of climate and nutrients on phytoplankton biomass	22
2.8	Policy and management	23
Chapter	3: Site description and methods	27
3.1	Introduction	27
3.2	Site description	27
3.3	Stream discharge and meteorological data	
3.4	Nutrient analysis	
3.5	Automatic sensor deployment in Milltown Lake and calibration	
3.6	Chlorophyll-a analysis	
3.7	Analysis of lake temperature profile data	
3.8	Statistical methods	
Chapter of Millt	4: The effects of streamside fencing on P and N concentrations in catchmer own Lake	nt streams

4.1	Abstract	
4.2	Introduction	
4.3	Aims	41
4.4	Methods and materials	41
4.4	4.1 Site selection	
4.4	4.2 Sample collection and nutrient analysis	
4.4	4.3 Data analysis	45
4.5	Results	
4.5	5.1 Meteorological and hydrology data	
4.5 trib	5.2 General nutrient concentrations at the lower sites of the fenced and unf butary	enced
4.5	5.3 Upstream to downstream nutrient concentration change	
4.5	5.4 Drivers of nutrient concentrations in the fenced and unfenced tributary	
4.5	5.5 Concentration-discharge relationships at a sub-daily scale	61
4.6	Discussion	71
4.7	Conclusion	
Chapter	er 5: Multiple drivers of phytoplankton biomass in Milltown Lake	
5.1	Abstract	
5.2	Introduction	
5.3	Aims	
5.4	Material and methods	
5.4	4.1 Examining the inter-annual thermal structure using Lake Analyzer	
5.4 nut	4.2 Sampling programme assessing the importance of both climatic stresso atrient availability in controlling the standing stock of phytoplankton biomass.	ors and83
5.4	4.3 Data analysis	
5.5	Results	
5.5	5.1 Meteorological and hydrology data	
5.5	5.2 Annual lake temperature and stratification profile (2010, 2011 and 201	2)90
5.5	5.3 Lake abiotic data (01 st of May to the 17 th of September 2012)	
5.5 201	5.4 Drivers of chlorophyll a concentration $(1^{st} \text{ of May to the } 17^{th} \text{ of Septen})$	1ber 99
5.6	Discussion	
5.7	Conclusion	
Chapter spring s	er 6: Influences of multiple drivers on phytoplankton diversity in Milltown Lal	ce during
6.1	Abstract	

6.2	Intro	oduction	111
6.2	.1	Phytoplankton	111
6.2	.2	Zooplankton	114
6.2	.3	Deep Chlorophyll Maximum (DCM)	115
6.2	.4	PEG model concept	116
6.3	Aim	ns of this study	118
6.4	Met	hods	118
6.4	.1	Sampling depth calculation and abiotic sampling protocol	118
6.4	.2	Collection of biotic parameters based on Secchi depth measurements	121
6.4	.3	Phytoplankton counting	121
6.4	.4	Zooplankton counting	122
6.4	.5	Other data	123
6.4	.6	Data analysis	124
6.5	Res	ults	124
6.5	.1	Meteorological and in-lake vertical temperature profile data	124
6.5	.2	Lake abiotic data	128
6.5	.3	Drivers of chlorophyll a concentration	132
6.5	.4	Lake biotic data: phytoplankton species composition and abundance	134
6.5	.5	Phytoplankton biomass	136
6.5	.6	Zooplankton species composition and abundance in terms of population	
den	sities		138
6.5	.7	Drivers of change in phytoplankton group assemblage	141
6.6	Disc	cussion	144
6.7	Con	clusion	151
Chapter	7: Fi	nal Discussion	152
7.1 General overview			152
7.2	Ove	erview of the trophic status of Milltown Lake and links to catchment nutrient	
expor	t		153
7.3	7.3 Drivers of phytoplankton biomass within small lakes15		155
7.4	Imp	lications of management	157
7.5	Ove	erall conclusion	159
Referen	ces		162

List of abbreviations

AEOS	Agri-Environmental Options Scheme
AIC	Akaike Information Criterion
CAP	Cattle Access Points
Chl-a	Chlorophyll <i>a</i>
DCM	Deep Chlorophyll Maxima
DO	Dissolved Oxygen
DS	Downstream
edf	estimated degrees of freedom
EEA	European Environment Agency
EPA	Environmental Protection Agency
EU	European Union
GAM	General Additive Model
GAP	Good Agricultural Practice
GLAS	Green Low-Carbon Agri-Environment Scheme
GLEON	Global Lakes Observatory Network
GWS	Group Water Scheme
LAWPRO	Local Authority Waters Programme
L.Flow	Log Transformed Flow
LU	Livestock Unit
NAP	National Action Programme
Ν	Nitrogen
N ₂ O	Nitrous Oxide
NaNO ₂	Sodium Nitrite
NaNO ₃	Sodium Nitrate
NH ₄	Ammonium
NH ₄ Cl	Sodium Chloride
(NH4)3 PO4	Ammonium Phosphate
NI	Northern Ireland
NO ₂	Nitrite
NO ₃	Nitrate
NSPPP	National Source Protection Pilot Project
NVZ	Nitrogen Vulnerable Zone
°C	Temperature
OSWTS	On-site wastewater treatment systems
Р	Phosphorus
PAA	Priority Area for Action
PAR	Photosynthetically Active Radiation
PCA	Principal Component Analysis
PEG	Plankton Ecology Group
PO4 ³⁻	Orthophosphate
РОМ	Programme of Measure
RBD	River Basin District
RBMP	River Basin Management Planning
REPS	Rural Environment Protection Scheme

ROI	Republic of Ireland
S	Schmidt stability
SRP	Soluble Reactive Phosphorus
SW	Shortwave radiation
TDN	Total Dissolved Nitrogen
TN	Total Nitrogen
ТР	Total Phosphorus
UK	United Kingdom
US	Upstream
USA	United States of America
WFD	Water Framework Directive
Wn	Wedderburn number

List of tables

Chapter 2:

Table 2.1 Summary of TP exports in small agricultural streams in Northern Europe.**Table 2.2** Summary of TN exports in small agricultural streams in Northern Europe.

Chapter 3:

- **Table 3.1** Characteristics of the western (fenced) and middle (unfenced) stream reaches, in the Milltown Lake Catchment.
- **Table 3.2** Long term average meteorological data at Met Éireann weather stations near Milltown Lake.
- Table 3.3 QuickChem® Methods used to determine phosphorus and nitrogen.

Chapter 4:

- **Table 4.1** Field sampling categories and dates for this study between the 20th of April 2011 to the 16th of April 2012.
- **Table 4.2** Summary of meteorological data (Ballyhaise and Coose Met Éireann weather station) and hydrological data (EPA staff gauge on the Drumleek River) for the main study periods. n denotes the number of days in each period.
- **Table 4.3** Median TP and TN concentrations for the fenced and unfenced, upstream (US) and downstream (DS) sites (mg L⁻¹) during the Total, Early and Late Grazing, Closed and Open time periods, together with 25th and 75th percentiles. Significance levels for the Mann Whitney U test are also presented.
- **Table 4.4** Generalized additive models for the lower sites on the fenced and unfenced tributary for TP and TN concentrations (mg L⁻¹), during the grazing period including extension period and TN concentration (mg L⁻¹) for the Open period.
- **Table 4.5** Summary of meteorological data (Coose Met Éireann weather station) and hydrological data (EPA staff gauge) for the four high discharge events sampled during the grazing season in 2010 and 2011.
- **Table 4.6** Phosphorus concentration-discharge relationship in the fenced and unfenced tributary during four high discharge events in the Grazing period of 2010 and 2011.
- **Table 4.7** Nitrogen concentration-discharge relationship in the fenced and unfenced tributary during four high discharge events in the Grazing period of 2010 and 2011.
- **Table 4.8** Summary of chapter findings.

Chapter 5:

- Table 5.1 Comparative summary table of meteorological data (Ballyhaise and Coose Met Éireann weather station) and hydrological data (EPA staff gauge) during the study period months (1st May to the 17th of September 2012) for the same period in the previous years 2010 and 2011.
- **Table 5.2** GAM model for the drivers of Chl-*a* concentrations (μg L⁻¹) at the surface, 2 m, 4 m, 6 m and 8 m depth intervals at the deepest point in Milltown Lake.

Chapter 6:

Table 6.1 GAM model for the drivers of chlorophyll *a* concentrations (μ g L⁻¹) at the surface, 2 m, 4 m, 6 m and 8 m depth intervals at the deepest point in Milltown Lake.

List of figures

<u>Chapter 2:</u> Figure 2.1 Phosphorus Cycle (Kumar et al., 2018). Figure 2.2 Nitrogen Cycle (ib.bioninja.com.au, 2016).

Chapter 3:

- Figure 3.1 (a) Milltown Lake Catchment, Co. Monaghan, Ireland showing the western (fenced) and middle (unfenced) sub-catchments. Autosampler locations indicated by red dots, for the upper (north) and lower (south) sites on the fenced and unfenced tributaries. EPA staff gauge location indicated by green dot. Milltown Lake is shown at the bottom of the figure.
 (b) Milltown Lake, the orange dot indicates the sampling point (deepest point) within Milltown Lake, and EPA staff gauge location indicated by green dot.
- Figure 3.2 (a) Milltown Lake and (b) drumlin landscape of Milltown Lake Catchment.
- Figure 3.3 (a) Before fencing installed (b) after fencing (1.5 m from the stream bank) installed as part of the NSPPP in the Milltown Lake Catchment.
- Figure 3.4 Location of the Ballyhaise, Clones and Coose *Met Éireann* weather stations in proximity to Milltown Lake.

Chapter 4:

Figure 4.1 Storm hydrograph (www. geographyaslevelaqa.wordpress.com, 2016).

- Figure 4.2 Distance from the source of the western, middle and eastern tributaries to Milltown Lake.
- Figure 4.3 (a) daily mean air temperature (°C) at Ballyhaise Met Éireann weather station, (b) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station and daily stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (c) Monthly rainfall (mm month⁻¹) (grey bars) and average monthly stream discharge data (m³ sec⁻¹ month⁻¹) (continuous black line) for the study period (April 2011 April 2012). r = Pearson correlation for April to September 2011 and October 2011 to April 2012.
- Figure 4.4 Daily TP concentration (mg TP L⁻¹) during the (a) Early Grazing, (c) Late Grazing, (e) Closed and (g) Open periods and daily TN concentration (mg TN L⁻¹) during the (b) Early Grazing, (d) Late Grazing, (f) Closed and (h) Open periods, for the lower sites on the fenced (black) and unfenced (grey) tributaries. Stream discharge from the EPA gauge is shown as grey bar. Note difference in scales between TP and TN figures.
- Figure 4.5 Median TP concentrations for the fenced (**a**, **b**, **c** and **d**) and unfenced (**e**, **f**, **g** and **h**) upstream (US) and downstream (DS) sites (mg TP L⁻¹) for the Early Grazing (EG, n = 50), Late Grazing including extension period (LG.ex, n = 126), Closed (C, n = 17), and Open (O, n = 63) periods. The whiskers indicate the maximum and minimum concentrations.
- Figure 4.6 Median TN concentrations for the fenced (a, b, c and d) and unfenced (e, f, g and h) upstream (US) and downstream (DS) sites (mg TN L⁻¹) for the Early Grazing (EG, n = 50), Late Grazing including extension period (LG.ex, n = 126), Closed (C, n = 17), and Open (O, n = 63) periods. The whiskers indicate the maximum and minimum concentrations.
- Figure 4.7 Estimated smoothing curves in the Grazing periods including extension period for TP concentrations (a) L.Flow fenced, (b) L.Flow unfenced and TN concentrations (c) L.Flow fenced, (d) L.Flow unfenced, lower reaches. Estimated smoothing curves in the Open period for TN concentrations (e) L.Flow fenced and (f) L.Flow unfenced, lower reaches. The solid line is the smoother and dotted lines indicate the 95% confidence intervals.

- **Figure 4.8** Estimated smoothing curves in the Grazing period including extension period for TP concentrations, for Air_Temp unfenced lower reach. The solid line is the smoother and dotted lines indicate the 95% confidence intervals.
- **Figure 4.9** Stream discharge (m³ sec⁻¹) from the EPA gauge is shown as grey line and daily rainfall (mm) from Coose Met Éireann weather station for (**a**) 2010 and (**b**) 2011. The events circled indicate the high discharge event sampled. The area between the dashed black lines indicates the Grazing period.
- **Figure 4.10** Phosphorus concentration-discharge relationships at the lower sites on the fenced and unfenced tributaries in July 2010. Red circle indicates the start of the event and the arrows indicates the direction and the end of the event. C indicates clockwise direction and Fig 8 indicates figure-of-eight direction.
- **Figure 4.11** Phosphorus concentration-discharge relationships at the lower sites on the fenced and unfenced tributary during high discharge events in September 2010 and October 2011. Red circle indicates the start of the event and the arrows indicates the direction. C indicates clockwise direction, Anti-C indicates anti-clockwise direction and Fig 8 indicates figure-of-eight direction.
- Figure 4.12 Nitrogen concentration-discharge relationships at the lower sites on the fenced and unfenced tributary during high discharge events in September 2010 and October 2011. Red circle indicates the start of the event and the arrows indicates the direction. C indicates clockwise direction, Anti-C indicates anti-clockwise direction and Fig-8 indicates figure-of-eight direction.
- Figure 4.13 Cattle access points and areas of bare soil exposure along the unfenced tributary in the Milltown Lake Catchment.

Chapter 5:

- Figure 5.1 2010 to 2012 data for (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station, plotted against daily stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (b) daily average wind speed and 30 day moving average wind speed (black line) for Milltown Lake, (c), daily average photosynthetically active radiation (PAR) and (d) daily mean air temperature (°C) at Ballyhaise Met Éireann weather station, between 1st of January 2010 and 31st of December 2012. The dotted black lines define each calendar year, while the area between the solid red lines define the sampling period in 2012.
- Figure 5.2 2012 data for (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station, plotted against stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (b) daily average wind speed and 30 day moving average wind speed (black line) for Milltown Lake, (c) daily average photosynthetically active radiation (PAR) from the Ballyhaise Met Éireann weather station and (d) daily average surface water temperature for Milltown Lake and air temperature from the Ballyhaise Met Éireann weather station, January to December 2012. The white area between the dashed lines represents the study period (1st May to the 17th of September 2012).
- Figure 5.3 16th of April 2010 and the 31st of December 2012 (a) contour plot of lake water temperature (°C), (b) Schmidt stability (J m⁻²), (c) buoyancy frequency (N s⁻¹) and (d) Wedderburn number (Wn). The dotted grey lines define each calendar year.
- Figure 5.4 1st of May to the 1st of October 2012, (a) stream discharge (m³ sec⁻¹) and nutrient loadings (g ha⁻¹ d⁻¹) from the lake inlet site, (b) dissolved oxygen (mg L⁻¹) at 1 m and 7 m depth (c) contour plot of lake temperature (°C) with thermocline depth (m) (superimposed black line), Secchi depth measurement on day of sampling (m) (superimposed white dots) and (d) Schmidt stability (J m⁻²). Note difference in scale for total N and total P loads.

- **Figure 5.5** Contour plots of (a) chlorophyll *a* (μg L⁻¹), (b) TP (mg L⁻¹) and (c) soluble reactive phosphorus (SRP) (mg L⁻¹), with thermocline depth superimposed (m), at the deepest point in Milltown Lake, between the 1st of May 2012 and the 17th of September 2012.
- **Figure 5.6** Contour plots of (a) TN (mg L⁻¹), (b) ammonium (NH₄-N) (mg L⁻¹) and (c) nitrate (NO₃-N) (mg L⁻¹), with thermocline depth at the deepest point in Milltown Lake, superimposed (m), between the 1st of May 2012 and the 17th of September 2012.
- Figure 5.7 Estimated smoothing curve for chlorophyll a concentrations (μg L⁻¹) at the surface
 (a) stream discharge, (b) WS (wind speed); 2 m (c) S (Schmidt stability), (d) WS (wind speed); 4 m (e) mean thermocline depth, (f) PAR (photosynthetically active radiation), (g) Lake NH4; 6 m (h) mean thermocline depth, (i), PAR (photosynthetically active radiation) (j) Lake NH4 and at the 8 m (k) mean thermocline depth, (l) PAR (photosynthetically active radiation) intervals at the deepest point in Milltown Lake. Grey shaded area indicates the 95% confidence bands.

Chapter 6:

Figure 6.1 Sampling depths in Milltown Lake in spring 2013.

- Figure 6.2 (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station and stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (b) daily average wind speed for Milltown Lake, (c) daily average photosynthetically active radiation (PAR) from the Ballyhaise Met Éireann weather station and (d) daily average surface water temperature for Milltown Lake and air temperature from the Ballyhaise Met Éireann weather station, for six month period (January to June 2013). The area between the dashed lines indicates the study period. The red line indicates the day on which the lake stratified (26th May 2013).
- Figure 6.3 (a) stream discharge (m³ sec⁻¹) and nutrient loadings (g ha⁻¹ d⁻¹) from the lake inlet site, (b) dissolved oxygen (mg L⁻¹) at 1 m and 7 m depth. The grey area between the grey dashed line indicates the range for DO at 1 m and the area between the black dashed line indicates the range for DO at 7 m. (c) contour plot of lake temperature (°C) with thermocline depth (m) superimposed and (d) Schmidt stability (J m⁻²), between the 18 April 2013 and the 20 June 2013. Note difference in scale for TN and TP loads, where TP concentrations are multiplied by a factor of 5. The red line indicates the start of thermal stratification in Milltown Lake.
- **Figure 6.4** Contour plots of (a) chlorophyll *a*, (b) total P, (c) soluble reactive phosphorus, (d) total N, (e) nitrate (NO₃-N) and (f) ammonium (NH₄-N) concentrations with Secci depth indicated by the solid white line and the bottom of the photic zone indicated by the dashed white line superimposed (m), at the deepest point in Milltown Lake, between the 18 April 2013 and the 20 June 2013. All units are in mg L⁻¹, with the exception of Chl-*a* which is in μ g L⁻¹. The white section at the lower righthand corner of the figure denotes a lowering in lake water level.
- Figure 6.5 Estimated smoothing curve for chlorophyll *a* concentration (μg L⁻¹) at the surface
 (a) PAR_3D, (b) Lake_TP, (c) Lake_SRP, 2 m (d) TN_catch_load, 4 m (e) TN_catch_load, 6 m (f) Lake_TN, (g), Lake_NH4 and 8 m (h) Lake_TN, depth intervals at the deepest point in Milltown Lake. Grey shaded area indicates the 95% confidence intervals. PAR = photosynthetically active radiation.
- **Figure 6.6 (a)** Phytoplankton abundance (Number L⁻¹) counted at Depth 1 (0.1 m below the surface) and **(b)** Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.
- Figure 6.7 (a) Percentage contribution of Phytoplankton counted at Depth 1 (0.1 m below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.

- **Figure 6.8 (a)** Phytoplankton biomass (pg C ml⁻¹) counted at Depth 1 (0.1 m below the surface) and **(b)** Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.
- **Figure 6.9 (a)** Zooplankton abundance (Number L⁻¹) counted at Depth 1 (10 cm below the surface) and **(b)** Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.
- Figure 6.10 (a) Percentage Contribution of Zooplankton counted at Depth 1 (10 cm below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.
- **Figure 6.11 (a)** Zooplankton biomass (pg C ml⁻¹) counted at Depth 1 (0.1 m below the surface) and **(b)** Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.
- Figure 6.12 Principal Component Analysis (PCA) of individual phytoplankton species (red) and potential drivers (black), before and after thermal stratification (dates in orange) in Milltown Lake at (a) Depth 1 and (b) Depth 2. WS = windspeed, PAR = photosynthetically active radiation, preD_cath_load = previous day catchment nutrient load, catch_load = catchment nutrient load on the day of sampling, Mean_thermo._depth.= mean lake thermocline depth, SS = Schmidt stability, zooplankton biomass (pg C ml⁻¹).

Chapter 1: General introduction

1.1 Introduction

Cultural eutrophication is a worldwide problem which is linked to increased availability of one or more limiting growth factors needed for photosynthesis (e.g. nutrients, carbon dioxide and sunlight), and can cause a substantial increase in the density of algal cells suspended in water (Smith and Schindler 2009; Jones and Brett, 2014). Algal blooms and enhanced growth of other aquatic vegetation disrupt the normal functioning of lake ecosystems. Algal growth on the surface of lake waters limits sunlight penetration, as well as photosynthesis by the algae, causing a reduction in the oxygen levels in the lake's waters. Levels of deoxygenation are also increased by microbial decomposition of dead algae. As algae die and decompose, the levels of dead and decaying organic matter increase, in the water column and sediment, depleting the water of available oxygen and causing the death of other organisms (Capella et al., 2014). Ultimately, these limitations in available sunlight and oxygen in lakes will affect the biodiversity of a lake system, such as anoxia, development of potentially toxic blue-green algal blooms (i.e. cyanobacteria), the reduction in plant diversity, deterioration in overall water quality and fish life (Søndergaard et al., 2001; Schindler, 2006; Istvánovics, 2009; May et al., 2009; Jeppesen et al., 2012). In addition, this can have a big impact on recreational opportunities which a lake can provide (Dodds et al., 2009; Chislock et al., 2013; Hjerppe et al., 2017).

Eutrophication of freshwaters remains a recognised challenge for governments throughout the developed and developing world, particularly the control and mitigation of the impacts of nutrient transfers from land to water (Aneja et al., 2008; Vaccari, 2009; Macintosh et al., 2011). Phosphorus (P) and nitrogen (N) are the key nutrients linked to eutrophication, with excess loading increasingly being associated with agricultural practices and, to a lesser degree, point sources within Europe (Smith and Schindler, 2009; Dupas et al., 2015; EEA, 2016). In Rural areas, the point sources include on-site wastewater treatment systems (OSWTS) that service the

local population, as well as farmyards and cattle access points (CAP) to streams and lakes (Deasy et al., 2010; Neal et al., 2010; Macintosh et al., 2011). These local point sources within agricultural catchments can contribute to the overall nutrient loss, however, it is generally considered that wider, hydrology-driven diffuse sources are predominantly linked to incidental losses of nutrient stores from soils to waterbodies within these catchments (Withers et al., 2001; Jordan et al., 2005; Sharpley et al., 2013).

Throughout Europe, diffuse agricultural pollution remains an issue, regardless of improvements in the control of point sources implemented in recent decades (Smith et al., 2005; Schindler, 2006; Ulén et al., 2007; Heckrath et al., 2008; Schulte et al., 2010; Dupas et al., 2015). Mitigation measures contained within the Nitrates Directive (91/676/EEC) and the European Union (EU) Water Framework Directive (WFD) (2000/60/EC) are being used to curtail diffuse pollution within agricultural catchments. These measures include constraints on the timing and magnitude of P and N in organic and inorganic fertiliser applications, ensuring that the optimum P fertility is maintained and incidental P transfer during hydrological active periods is avoided (Fealy et al., 2010; Macintosh et al., 2011; Jordan et al., 2012). They also include fencing to exclude cattle access to streams (Galeone, 2000; Vidon et al., 2008; Miller et al., 2010).

In Ireland, agriculture accounts for 65% of the total land use by area with 81% of this area devoted to dairy and beef cattle production on managed grassland (CSO, 2015; Department of Agriculture, Food and the Marine, 2016). The total number of cattle in the Republic of Ireland (ROI) at the end of 2017 was 6,673,600 (80% beef and 20% dairy cattle), compared to an overall human population of 4,749,153 (CSO, 2018). Agriculture in Ireland mainly comprises of grassland agriculture with some 80-90% of the diet of dairy and beef animals being attributed to grass or silage that is grown on the farm (Department of Agriculture, Food and the Marine, 2016). Annual TP and TN excretion rates for cattle is 13 kg TP yr⁻¹ and 65 kg TN yr⁻¹ (Ulén et al.,

2007; Stoate et al., 2009; Lewis et al., 2013; S.I. No. 31 of 2014). Spreading of slurry is prohibited during a set period in Ireland to comply with the Nitrates Directive (91/676/EEC) and depending on where in the country you live, the open dates for land spreading are from January to September and usually the closed period is October to January (Teagasc, 2011).

Problems related to eutrophication have resulted in a large body of research on both the export of nutrients to lakes (Bosch et al., 2014; Soranno et al., 2015; Vystavna et al., 2017) and possible mitigation measures (Cherry et al., 2008; Ranalli and Macalady, 2010; Miller et al., 2011). In addition, there is a substantial body of literature concerning the relationships between phytoplankton in lakes and nutrients (Hayes et al., 2015; Søndergaard et al., 2017). With lakes being under additional pressures from the effects of climate change, both past and present (Bosch et al., 2014; Hayes et al., 2015; Verma et al., 2015; Wang et al., 2018), it is important to gain an understanding of how changes in climate will affect nutrient export (e.g. during increased and more persistent rainfall events) and responses in lakes during critical times in order for lakes to be adequately managed in the future.

External nutrient sources from the wider catchment area have been the main focus of management in eutrophic lakes. However, contributions from internal lake nutrient sources, such as from lake sediments to the overlying waters, are being increasingly examined (Søndergaard et al., 2007), with a renewed emphasis on what is described as 'legacy P' (Sharpley et al., 2013). Legacy P refers to the accumulation of P in sediments found in down slope areas such as rivers and lakes which can be remobilised or recycled, thus acting as a continuing source of P to downstream waterbodies for years, decades, or even centuries (McDowell et al., 2002; Kleinman et al., 2011; Sharpley et al., 2013). In lakes, internal nutrient loading is related to high concentrations of organic P and N building up in lake sediments which is being released back into the overlying water, usually under low dissolved oxygen (DO)

conditions (Søndergaard et al., 1999; Burger et al., 2008). These nutrients can promote algal growth and impair water quality, significantly delaying remedial efforts on eutrophic lakes, even following load reduction (Meals et al., 2010; Spears et al., 2012; Sun Loh et al., 2013; Zhu et al., 2013; Wu et al., 2014; Zamparas and Zacharias, 2014).

In addition to the key role that excess nutrients play in eutrophication, it is recognised that changes in local weather conditions may also cause qualitative and quantitative changes in phytoplankton community dynamics (Paerl and Fulton, 2006; Stella et al., 2007; James et al., 2009; Cai et al., 2012). Such weather related effects are now receiving closer attention due to the long term trends attributed to climate change. Increasing water temperatures are having a direct effect on algal growth, for example, with harmful phytoplankton groups such as cyanobacteria being promoted by rising water temperatures (O'Neil et al., 2012; Deng et al., 2014; Rigosi et al., 2014). Furthermore, increases in rainfall are leading to more frequent and larger pulses of nutrient runoff from surrounding land, intensifying the symptoms of eutrophication in many freshwater lakes particularly in agricultural catchments (Jeppesen et al., 2009; Moss et al., 2011; Deng et al., 2014).

In Ireland, small lakes are prevalent, particularly in the drumlin belt, which traverses the island from County Down in the east, to Counties Sligo and Donegal on the Atlantic coast (Aalen et al., 1997; Carson, 2010). These lakes interlinked through a network of rivers and streams, and are typically small in area, with short retention times, and often serve as local drinking water sources for the surrounding communities (National Federation of Group Water Schemes, 2016). Much research has been carried out internationally that highlights the various issues affecting small, and very shallow, freshwater lakes. Previous studies by Krovang et al. (2005) and Søndergaard et al. (1999) carried out in Denmark have highlighted the delays in lake recovery following the reduction of external catchment nutrient loads, due to re-suspension of internal nutrients from the lake sediment. Jeppesen et al. (2014) highlighted that climate change may have both direct and indirect effects on the plankton community structure. These studies highlight the complexity involved in implementing remediation measures that are required to tackle eutrophication problems in freshwater lakes.

Despite the large body of literature on nutrient mitigation measures, these studies have tended to focus on systems in the USA, Canada and Australia (McKergow et al., 2003; Galeone, 2006; Vidon et al., 2008; Bartley et al., 2010; Miller at al., 2010; Vidon et al., 2010; Miller at al., 2011), with few that relate to livestock agriculture in western Europe (Conroy et al., 2016; Bragina et al., 2017). In addition, while there is a large body of literature on small shallow lakes from mainland Europe, particularly Denmark (Trolle et al., 2015; Gutierrez et al., 2016; Søndergaard et al., 2017; Molina-Navarro et al., 2017), those lakes tend to be more shallow than Irish inter-drumlin lakes. They, therefore, tend to be more polymictic (Søndergaard et al., 2017) and more affected by sediment nutrient release (Kragh et al., 2017). There have been few studies to date on the combined effects of nutrients and weather on inter-drumlin lakes. The current study includes assessing the effectiveness of streamside fencing as a mitigation measure to reduce nutrient load from the catchment, and both internal nutrient sources and local climatic pressures, to gain an overall insight into how future catchment management strategies can be implicated for shallow drumlin lakes.

1.2 Aims and objectives

The overall aim of this study was to explore the relative effects of nutrient export from a cattle based agricultural catchment and local weather conditions as drivers of change in phytoplankton species in a small inter-drumlin lake, and therefore, to inform future management of these systems with an aim of achieving a minimum of good ecological status as set out in the WFD. Specific objectives included:

- **Chapter 4:** Investigating the effects of agricultural management of extensive livestock farms on P and N levels in catchment streams, including the effect of excluding cattle from stream waters, using fencing as a mitigation measure.
- Chapter 5: Investigating the relative importance of (1.) local weather changes and their effects on lake physical characteristics, including stratification, and (2.) nutrient (P and N) levels on the standing stock of phytoplankton biomass as measured by Chlorophyll *a* (Chl-*a*).
- **Chapter 6:** Studying the relative importance of changes in water column stability, nutrient availability, and zooplankton grazing in determining plankton diversity in a small lake at the surface, and at the photic depth, during the onset of spring thermal stratification.

1.3 Knowledge gaps

- The exact timings and quantity of slurry and fertilisers application within the catchment during this study were not exactly known. This information would be difficult to obtain and require more resources. However, this information would be beneficial if a similar study were to be carried out in the future.
- There are no arterial drainage schemes within the catchment, however, the history of land improvement works carried out by farmers within the catchment is not known. Although this information is not known, an extensive stream walk of both the fenced and unfenced tributaries was carried out with no potential problematic drains identified.
- The installation of a photosynthetically active radiation (PAR) sensor within the lake and a remote weather station were not feasible for this study due to budget constraints. However, due to the proximity of two Met Éireann weather stations one in Ballyhaise, Co. Cavan and the second in Coose, Castleblayney, Co Monaghan, it is believed that the data obtained at both stations would still be sufficient in answering the questions asked in this study.

1.4 Thesis structure

The thesis is presented in seven chapters. Chapter one provides the study background and detailed description of the study objectives. A detailed literature review is presented in Chapter two, focusing on the role that climate and nutrients (P and N) play in controlling phytoplankton biomass. Chapter three describes the location of Milltown Lake and its catchment, along with the general material and methods used during this research. Chapter four presents the results of a study which focused on the effects of streamside fencing within the Milltown Lake Catchment and its ability to reduce nutrient loss to surface water . This was assessed by comparing a fenced and an unfenced tributary stream reach with similar topography and land management practices during three contrasting management periods in the Irish agricultural calendar. Chapter five looks at the long-term thermal structure in Milltown Lake using high frequency weather and *in-situ* temperature sensor data between April 2010 and September 2012. This chapter then particularly focuses on the importance of climate and nutrients as drivers in controlling the standing stock of phytoplankton biomass as measured by Chl-a during the growing season in 2012. Chapter six further investigates the influence that thermal dynamics play in plankton structure during the onset of stratification. This was assessed by looking at the physical parameters, nutrient concentrations, and plankton communities during the spring of 2013. Chapters four to six each contain an abstract, detailed introduction, methods, results, discussion and conclusion sections. Chapter seven discusses the findings of the previous chapters, presenting the final conclusions and highlighting the potential application of this research to other eutrophic lakes and catchments.

1.5 Hypotheses

• Chapter 4: Streamside fencing and thus cattle exclusion will reduce the overall P and N concentrations in streams within agricultural catchments.

- Chapter 5: Local climatic factors (wind and rainfall), physical drivers (thermal stratification) and nutrient (P and N) concentrations does influence phytoplankton biomass within a small lake.
- **Chapter 6:** Water column stability, nutrient (P and N) availability and zooplankton grazing are drivers in determining plankton diversity at both the subsurface and photic zone depth during the onset of spring thermal stratification.

Chapter 2: Literature review

2.1 Sources of phosphorus and nitrogen in lakes

Although natural eutrophication is a slow-aging process for a waterbody, human activities speed up the process greatly (Art, 1993). Within a lake's catchment, it is geology and land use that generally determine the amount of nutrients entering the lake either *via* surface water or groundwater channels. Water transported by the rivers and streams draining into a lake carry with them much of the suspended sediment and nutrients which have runoff from the surrounding landscape. When river water enters a lake, its speed diminishes rapidly, bed-load transport ceases, and the suspended load of sediments and nutrients begins to settle to the bottom. However, external nutrient loads from the surrounding catchment are not the only sources of elevated nutrient concentrations in lakes. Internal lake sediment can play an important role in the retention and release of nutrients depending on lake depth, due to physical (wind, temperature and stream discharge), chemical (oxygen depletion in the hypolimnion, resulting in nutrient release from the lake sediment due to anoxia) and biological factors (decomposition of organic matter) (Egemose et al., 2011; Palmer-Felgate et al., 2011; Søndergaard et al., 2013).

2.1.1 Phosphorus loss from surrounding agricultural land

Phosphorus losses from land are mainly associated with the erosion of P enriched soils, especially from bare or poached soils, with excess manure/fertiliser applications and with surface runoff of dissolved forms occurring during heavy rainfall (Haygrath and Jarvis, 1999; Hooda et al., 2000; Jennings et al., 2003; Kiely et al., 2007; Palmer-Felgate et al., 2009; Macintosh et al., 2011). Concentrations of P in runoff, particularly from agricultural land, are dependent on the quantities of P present in, or on, the soil and the extent to which water moving through or over the soil captures these sources (Jennings et al., 2003; Withers and Hodgkinson, 2009). The form of P transported in runoff from organic fertiliser spreading on grassland is

variable, depending on specific farming practices. Organic P is the principal form of P in slurry and animal manures, comprising approximately two-thirds of the P in fresh manure (Jordan and Rippey, 2003; Kurtz et al., 2005; Withers and Hodgkinson, 2009). In contrast, inorganic P is the main component of chemical fertilisers (Dils et al., 1999) and can be found in either soluble or bound forms.

Soluble inorganic P is a strong anion that is highly reactive and, therefore, attaches to exchange sites in the soil (Daly et al., 2002). This is the form of P that is taken up by plants (Cordell et al., 2009) (Figure 2.1). Over time, soluble inorganic P will react with cations such as calcium, iron, aluminium and manganese, slowly converting to a less soluble form known as unavailable inorganic (bound) P (Figure 2.1). This limits its availability to plants (Wiederholt and Johnson, 2005). A large fraction of soil P exists as bound inorganic P and can either be attached loosely or tightly to cations. Loosely bound P particles can remain in equilibrium with soluble P and become available in soil pore waters (Figure 2.1). This enables bound inorganic P to be converted into soluble inorganic P when plant removal reduces the concentration of soluble P, thus maintaining the equilibrium (Wiederholt and Johnson, 2005).



Figure 2.1 Phosphorus Cycle (Kumar et al., 2018).

The export of nutrient in overland flow related to fertilisers and manures is more prevalent after periods of heavy rainfall, with areas of land that have a high soil test P being the more susceptible (Galeone, 2006; Reichenberger et al., 2007; Buckley, 2012). Soil erosion by water includes the processes of detachment, transportation and deposition of the soil particles by raindrops or surface flow (Foster et al., 1985; Hatfield and Follett, 2008). High rainfall events can be associated with higher losses of particulate P and dissolved P. However, particulate material will only be transported during high rainfall events (Macintosh et al., 2011). In some agricultural areas, artificial land drains can represent an important pathway of particulate P, particularly during periods of heavy rainfall, as they tend to have a direct pathway to the watercourse (Chapman et al., 2001; Jennings et al., 2003; Montagne et al., 2009; Vidon et al., 2012). A further source of particulate P loss from agricultural land is through increased animal activities around and within streams, resulting in bankside erosion of soil and disturbances of the streambed sediment (Clew et al., 2010; Bond, 2012).

In Ireland, incidental P transfers from slurry and manure management, fertiliser spreading, and diffuse losses from high nutrient status soils all contribute substantially to agricultural P losses (Schulte et al., 2010). In European countries such as Norway and Sweden, where arable agriculture predominates and plant cover may be absent for long periods, sediment P losses to water are a particular problem (Ulén et al., 2007). A summary of TP exports in small agricultural streams reported in studies carried out in Northern Europe and Ireland is summarised in Table 2.1.

Study	Export (kg TP ha ⁻¹ yr ⁻¹)	Description of catchments
		Estimation of TP export in north-western
Ulen et al. (2007) & Stoate	03 60	European countries (Norway, Sweden,
et al. (2009)	0.5 0.0	Ireland and the UK), with the highest
		losses in Norway and UK.
Iordan et al. (2007)	0.2 - 3.1	Three different grassland catchments in
	0.2 - 5.1	Ireland.
Green et al (2001)	0.1 - 1.0	Lough Sheelin a drumlin grassland
		catchment in central Northern Ireland.
Macintosh et al. (2011) &	1.8 – 3.8	Cross border grassland catchment in
Cassidy and Jordan (2011)		between Northern Ireland and Ireland
		over a four year study period.
		Lough Leane, a large lake with both
Jennings et al. (2013)	0.55	agriculture and peatland sub catchment in
		Co. Kerry, Ireland.
Mellander et al. (2012)	0.54 0.70	Four agricultural catchments (arable and
	0.54 - 0.70	grassland) across Ireland.

Table 2.1 Summary of TP exports in small agricultural streams in Northern Europe.

2.1.2 Nitrogen loss from surrounding agricultural land

Nitrogen losses to waterways are mainly attributed to over-fertilisation, excessive manure applications, autumn ploughing, intensive stocking of pastures, cattle access to waterways and mineralisation of soil organic material (Shepherd et al., 2001; Follett, 2008; Zaied, 2009). The predominant forms of N in the soil are ammonium (NH4⁺) and nitrate (NO3⁻), with nitrite (NO2⁻) and nitrous oxide (N2O) being present to a lesser extent (Dalgaard et al., 2011) (Figure 2.2). Nitrogen is required for the survival of all living things and is an essential nutrient for the production of crop plants (Follett, 2008). Plants tend to take up N as NH4⁺ and NO3⁻, whereas NO2 is essentially toxic to plants (Taylor and Klepper, 1975). The main processes in the nitrogen cycle involve nitrogen fixation, ammonification, nitrification and denitrification (Figure 2.2). The initial process nitrogen fixation is where N2 is transformed to organic nitrogen in order for the consumption by living organisms (Figure 2.2). This process is considered the most dominant process by which N enters the soils biological pool (Robertson and Groffman, 2007). Once N has entered the soil or water, it undergoes the process of ammonification whereby bacteria convert N or nitrogen compounds into NH4⁺. Ammonium, with its positive charge, bonds to the negatively charged surfaces of soil particles and has a strong affinity to

clay in particular (Seybold, 1994). Soil concentrations are generally quite low, because the biological conversions from NH_4^+ to NO_3^- , which are favourable for mineralisation, tend to be rapid (Brown, 1993). Nitrite is an intermediate species in the conversion from NH_4^+ to NO_3^- and tends not to accumulate in soil because the conversion from NO_2^- to NO_3^- is much faster than from NH_4^+ to NO_2^- (Burns et al., 1996) (Figure 2.2). Nitrate is a weak anion and a very soluble compound; therefore, it does not have a strong affinity to bind to the soil or to other insoluble compounds present in the soil (Krammer et al., 2006).



Figure 2.2 Nitrogen Cycle (ib.bioninja.com.au, 2016).

Nitrogen is typically leached from a surface source to groundwater as NO_3^- and usually reaches surface waters via subsurface groundwater channels or artificial drains (Kc, 2007; Robertson and Vitousek, 2009; Follet et al., 2010; Mellander et al., 2012). Elevated N in groundwater often results in poor drinking water quality, with concentrations reaching levels above those acceptable for drinking water supply (i.e. < 50 mg NO₃ L⁻¹) (Directive 91/676/EEC). Losses of N may, however, also occur via surface runoff (Jiao et al., 2006). Surface runoff occurs when rainfall exceeds the maximum infiltration level of the soil (Heathwaite et al., 1998). Movement of N into surface water from soils is mainly associated with soil erosion during heavy rainfall and, to a lesser degree, through wind deposition (Cameron and Haynes, 1986). Some sediment particles may travel only a few millimetres, while others may be transported long distances before they are deposited or reach a surface water receptor (Follet et al., 2010). Nitrogen is primarily transported as organic N in soil or sediment organic matter, or as NO₃⁻, a form that is completely water soluble (Follet et al., 2010; Yang et al., 2013). However, agricultural soils that have recently received fertiliser and manure applications containing high NH4⁺ concentrations can result in NH4⁺ being washed from the land to nearby surface waters following episodes of high rainfall (Burns et al., 1996).

In the 2010 agricultural census, cattle density in Ireland had increased from the 0.76 livestock units (LU) ha⁻¹ reported in 1970 to 1.54 LU ha⁻¹ (CS0, 2010). Animals, in particular cattle, defecate on the fields and, in some cases directly into watercourses where access is permitted (Bond et al., 2012). High densities of dung and increased levels of soil-profile NO_3^- are frequently observed in areas near watering and shade points (Wilkinson et al., 1989; Haynes and Williams, 1993; Follett, 2008). The levels of nutrients, especially NH_4^+ , associated with fresh animal faeces or urine increase immediately in stream waters when in-stream livestock activity occurs (Demal, 1983; Sheffield et al., 1997; Miller et al., 2011).

In Europe, N from mineral fertilisers is the more dominant source of N, with animal manure, particularly chicken manure, being prominent in regions with high stocking density (McKauge et al., 2005). There are large differences in total nitrogen (TN) input, output and surplus for agricultural land between countries (Velthof et al., 2009). A summary of TN exports in small agricultural streams reported in studies carried out in Northern Europe and Ireland is summarised in Table 2.2.

Study	Export (kg TN ha ⁻¹ yr ⁻¹)	Description of catchments
	<50	Central Europe (Bulgaria, Estonia, Latvia
Velthoh et al. (2009) & EEA		and Romania)
(2010)		
	170	Denmark, Belgium and the Netherlands
		Lough Mask Catchment spanning
Venhor et al. (2005)	10.8	between Co. Mayo and Co. Galway,
		Ireland.
Fealy et al. (2010) & Wall et	8.9 - 28.8	Six agricultural catchments (arable and
al. (2011)		grassland) across Ireland
	24.4 & 13.3	Grassland and unimproved grassland
Jordan and Smith (2005)		catchments in the sub-catchments of the
Jordan and Siniti (2003)		River Upper Bann and the River
		Colebrooke

Table 2.2 Summary of TN exports in small agricultural streams in Northern Europe.

2.1.3 Internal nutrient loading from phosphorus and nitrogen to lakes

Internal cycling of nutrients is an important factor that supports to plankton productivity (Gardner et al., 1998), with lake sediments being an important source, particularly in freshwater lakes (Zhu et al., 2012). Phosphorus plays a key role in lake ecosystems, having a longer residence time and higher retention efficiency than N (Saunders and Kalff, 2001; Jeppesen et al., 2005; Cook et al., 2010; Grantz et al., 2014; Yan et al., 2016). Recently, however, besides the role of P, the importance of N limitation for lake productivity has been more intensively studied and discussed (Sterner, 2008; Conley et al., 2009; Moss et al., 2013; Olsen et al., 2015a; Søndergaard et al., 2017).

Phosphorus, once introduced into a lake's ecosystem, accumulates in the bottom sediments, which are considered net sinks and tend to have higher P concentrations than the overlying waters (Wetzel, 2001). Sedimentary P can act as a large internal load to the overlying water column, contributing to the growth of algal blooms, impairing water quality and significantly delaying remedial efforts in eutrophic lakes, even following external load reduction (Sun Loh et al., 2013; Zhu et al., 2013; Wu et al., 2014). The release of sedimentary P is influenced by various environmental factors, including temperature, pH, redox potential, DO, hydrological

conditions and biological activity (Kleeberg et al., 1998; Miao et al., 2006; Wu et al., 2014). The main P fraction released from the sediment is highly biologically available orthophosphate (unbound PO_4^{3-}). This form of P, when released during the period of thermal stratification of a monomictic or dimictic lake, can contribute to an increase in phytoplankton biomass during the summer and autumn months (Nürnberg, 2009; Sun Loh et al., 2013). Internal P loading during the winter can also increase spring productivity unless limited by light (Sun Loh et al., 2013). Studies have shown that sedimentary P release is mainly from the surface layer of the benthic sediment, therefore, the potential influence of sediment P in shallow lakes on water concentrations is stronger than that of deeper lakes, due to the ability for wind to mix up the benthic sediment in shallower lake systems (Søndergaard et al., 2003).

The N cycle is a complex biochemical process in which the various forms of N are altered by N fixation, assimilation and reduction of NO₃ to N₂ by denitrification (Wetzel, 2001). Wetzel (2001) described the simplified N cycle of lakes to be microbial in nature, with bacterial oxidation and reduction of N compounds being coupled with photosynthetic assimilation and utilisation by algae, photosynthetic bacteria and larger aquatic plants. In freshwater lake ecosystems, NH₄ can stimulate phytoplankton growth, exhibit toxicity to aquatic biota, and exert an oxygen demand on the surface waters (Beutel, 2006). In productive lakes, NH₄ concentrations accumulate from decomposition of organic matter and sediment release of NH₄ to the overlying waters can occur under anoxic conditions (Wetzel, 2001). Increases in N concentrations can negatively affect and degrade the ecological quality of a lake in several ways, for example by contributing to an increase in cyanobacterial biomass and dominance (Xu et al., 2010; Jeppesen et al., 2011; Davis et al., 2015). In some shallow lakes, the growth and species diversity of submerged macrophytes has been shown to be negatively impacted due to enhanced shading by phytoplankton and/or periphyton or effects of toxic stress on plant metabolism (Barker et al., 2008; Yu et al., 2015; Olsen et al., 2015); Søndergaard et al., 2017).

For instance, Olsen et al. (2015) showed, in a controlled experiment carried out on Lake Donghu (surface area 32 km², mean water depth 2.5 m) how the macrophyte population is partially resilient to abrupt increases in N loading, but, after prolonged exposure, a complete collapse occurs. Furthermore, Olsen et al. (2015) showed that macrophyte loss is intensified by the shading of filamentous algae during the winter months and by phytoplankton in the summer months.

2.2 Stratification of lakes

Thermal stratification is a natural phenomenon that occurs in lakes as a result of the thermal expansion properties of water (Read et al., 2011; Woolway et al., 2017). It is determined by the balance between turbulence, which acts to enhance mixing, and buoyancy forces, which act to suppress turbulence, resulting in vertical layering (Boehrer and Schultze, 2008; Woolway et al., 2014). The epilimnion is defined as that part of the water column that is immediately below the water surface, and directly influenced by atmospheric forcing, while the hypolimnion, generally the coolest and thus the densest layer, lies in contact with the lakebed. Both of these layers are separated by a temperature-driven density gradient known as the thermocline (Woolway et al., 2017). This vertical partitioning of the water column has important implications for the transport of nutrients and oxygen between the surface and deep water, or the light environment experienced by phytoplankton cells entrained in the mixed layer (Foley et al., 2012; Kirillin and Shatwell, 2016).

Stratification can be temporary or persistent, and its duration can vary from hours to months or even decades in some lakes (Jellison et al., 1998; Verburg et al., 2003; Rueda and Schladow, 2009). Shallow lakes, which mix to the bottom regularly, behave quite differently to deeper lakes, which tend to stratify as the weather warms causing a warmer, well-mixed layer to develop above a cooler poorly mixed layer, separating the surface water from deep, nutrient

rich waters and sediment (Kirillin and Shatwell, 2016). Shallow lakes, which do not develop a full seasonal stratification but tend to alternate between periods of mixing and stratification, tend to be strongly driven by meteorological conditions (Woolway et al., 2014; Woolway et al., 2017). These lakes are often termed polymictic according to Hutchinson's mixing classification and tend to be most abundant at mid-latitudes (Hutchinson, 1957; Woolway et al., 2017). In contrast, lakes that stratify continuously for a large part of the year are classified as dimictic (stratification in the summer and inverse stratification in winter), or monomictic if they stratify only once a year, either in the summer or winter months (Hutchinson and Löffler, 1956; Hutchinson, 1957; Lewis, 1983; Kirillin and Shatwell, 2016).

The main external stressors determining the mixing regime of lakes are meteorological conditions, changes in inflow discharge, and differences in lake morphology (Fee et al., 1996; Rimmer et al., 2011). Wind stresses that occur at critical time windows can result in the interruption or even breakdown of thermal stratification (Adrian et al., 2012; Jennings et al., 2012; Sommer et al., 2012). In addition, water transparency, or light extinction also play a major role in determining the mixing regime between polymictic and seasonally stratified lakes (Snucins and Gunn, 2000; Kirillin, 2010; Shatwell et al., 2016). For instance, transparency reduces the penetration of solar radiation to deeper layers, decreasing water temperatures and increasing vertical temperature gradients and water column stability (Kirillin and Shatwell, 2016).

Traditionally, lake depth temperature measurements have typically been collected on a weekly or fortnightly basis, although high-frequency monitoring of surface temperatures have been made (Maberly, 1996; Livingstone and Kernan, 2009). However, with the development of automatic in lake aquatic depth sensor technology, a large volume of data is rapidly being accumulated for lakes across the world (Porter et al., 2009; Read et al., 2011). This has allowed

the frequent monitoring of lake vertical temperature profiles from in-situ lake sensor platforms. To process such information more easily, the Lake Analyzer programme was created for the analysis of high-frequency data collected from instrumented lake buoys (Read et al., 2011).

2.3 The effects of wind on the breakdown of thermal stratification

The effect of wind stresses on a lake ecosystem is generally not associated with any seasonal or annual pattern. However, increased wind speeds can interrupt or even break down thermal stratification during a crucial time for a lake's ecosystem (Adrian et al., 2012; Jennings et al., 2012; Sommer et al., 2012). Moreno-Osten et al. (2009) showed that phytoplankton in the epilimnion responded rapidly to wind stresses in a Spanish lake. Wind induced mixing facilitates internal nutrient mixing, helps phytoplankton species to remain in suspension, facilitates species replacement and transports deeper species into the upper water column, therefore, changing the competitive balance between species (Huisman et al., 2004). Monero-Osten et al. (2009) studied the impact of short-term wind induced mixing on phytoplankton, specifically diatoms (negatively buoyant) and cyanobacteria (positively buoyant) and found that a higher biomass of diatoms occurred in the upper epilimnion in the winter during higher wind speeds, while a decline was evident in diatom concentrations during lower wind speeds. Regarding cyanobacterial behaviour, higher biomass in the upper epilimnion followed periods of lower windspeed, while lower biomass in the epilimnion occurred with increasing wind speeds. Huisman et al. (2004) reported similar findings with buoyant phytoplankton species, such as cyanobacteria, dominating in low turbulence conditions in a hypertrophic lake in the Netherlands, while sinking diatoms and green algae dominated in high turbulence conditions. Ibelings et al. (1994) suggested that lake stability is important for the formation of cyanobacteria blooms and that mixed conditions in lakes, associated with high wind speeds, inhibit the formation of cyanobacterial blooms.

2.4 Effects of light on phytoplankton groups

Phytoplankton are photoautotrophic, capturing and using solar energy in photosynthesis (Wetzel, 2001; Reynolds, 2006; Litchman and Klasmeijer, 2008). Phytoplankton must cope with a wide range of incident irradiance (Suggett and Borowitzka, 2010). Even without the fluctuation due to clouds, surface wave focusing or vertical mixing, phytoplankton need to remain in the upper layers of the water column for survival, where they have adapted to diel changes in light intensities (Reynolds, 2006; Suggett and Borowitzka, 2010). Three processes allow phytoplankton to adapt to changing light regimes, which all work at different time scales: adaption, acclimation and regulation (Huot and Bain, 2010). Due to changes in the genetic make-up of phytoplankton, different species are best suited for growth in different light environments, known as photoadaptation (Suggett and Borowitzka, 2010). Photoacclimation is when phytoplankton adjust their capacity to harvest and utilise light (e.g. add and remove pigments) to maintain photosynthetic efficiency under a variety of light intensities due to a change in environmental conditions (Geider et al., 2009). Finally, when phytoplankton cells respond to rapid changes in light intensity on a diel scale, and also accommodate changes in cloud cover, this is known as photoregulation (Huot and Bain, 2010). Photoregulation strategies of phytoplankton can reduce the capacity of photosynthesis to avoid cell damage at high light intensities, normally occurring under calm low wind conditions. The recovery from photoinhibition activities can occur within hours (Hiriart-Baer and Smith, 2005). Phytoplankton at the surface of lakes that are exposed to high UV-light can result in damage to the photosynthetic pigments and DNA of the plankton cells. To overcome this, cells exposed to high light irradiance generally have a lower content of Chl-a per cell in comparison to phytoplankton adapted to lower irradiance levels (Wetzel, 2001).

2.5 Effects of nutrients on phytoplankton groups

The controls on the community structure of phytoplankton have been the subject of intense study for many decades (Lynch and Shapiro, 1981; McQueen et al., 1989; Agawin et al., 2000;

Irwin et al., 2006; Zhu et al., 2010; Winder and Sommer, 2012; Deng et al., 2014; Carey et al., 2017). Generally, increased nutrient loads are thought to drive increases in cyanobacterial blooms in freshwater lakes (Brookes and Carey, 2011). However, under excessive nutrient loading conditions, a shift from cyanobacteria to chlorophytes has also been reported by Jensen et al. (1994), who studied approximately 200 shallow Danish lakes with a mean depth of < 3 m. This shift is largely due to the high growth and loss rates and, thus, higher demand for nutrients by chlorophytes in temperate lakes, whereas cyanobacteria have lower growth and loss rates and a lower demand for nutrients (Reynolds, 1984; Deng et al., 2014).

Seasonal variation in nutrient availability plays an important role in driving the change in phytoplankton community structure during different periods, with certain nutrients being limiting in the spring and different nutrients dominating the nutrient limitation process in the summer (Moss et al., 2013). For instance, during spring, increased losses from the catchment to the lakes, due to agricultural practices, can provide an excess loss of N over P. Therefore, as growth ends, P will have become depleted, while N will still be available (Lund, 1967; Moss et al., 2013). However, as lake water temperatures increase, resulting in stratification and deoxygenation of the hypolimnion, or as the sediments in a shallow lake become warmer and bacterial activity intensifies, releases of sediment P tends to increase, whilst N tends to denitrify as the water warms (Moss et al., 2013). This is particularly apparent in small shallow eutrophic lakes, as shown by Søndergaard et al. (2005), who found that, as P increased in the overlying water, a depletion of N occurred during the summer months, a process that may encourage the growth of N fixing cyanobacteria.

2.6 Chlorophyll *a* as a proxy for phytoplankton biomass

The measurement of phytoplankton biomass is a fundamental part of limnological assessments, although direct measurements are difficult. It is generally estimated either from cell counts or
from the concentration of the photosynthetic pigment Chl-*a* (Billington, 1991; Wetzel, 2001). Historically, measurements of Chl-*a* have, provided a useful estimate of algal biomass and its spatial and temporal variability. The pigment is present in all species of phytoplankton, including eukaryotic (algae) and prokaryotic organisms (cyanobacteria) (Gregor and Maršálek, 2004). The concentration of Chl-*a* present in a water sample is usually directly related to the number of algae living in the lake, therefore, acting as an indicator of the trophic state of a waterbody, with high levels indicating eutrophication. Chlorophyll *a* concentration is relatively easy and rapid to quantify, thus making it a reliable and popular proxy for total phytoplankton biomass (Felip and Catalan, 2000). Measurements are carried out by extracting the Chl-*a* pigment and quantifying the concentrations present using a spectrophotometer (Marra, 1997; Felip and Catalan, 2000; Mignot et al., 2011). Alternatively, Chl-*a* fluorescence can be used (Marra, 1997; Mignot et al., 2011).

Of the methods used to measure phytoplankton biomass there are different difficulties associated with each. The counting of phytoplankton cells can be time consuming and requires consistency to reduce errors associated with biomass calculations (Beardall et al., 2009). Additionally, the time required to count the cells, generally mean that the data obtained are at low resolution. With Chl-*a* extraction methods, all samples are measured at a specific wavelength which does not differentiate among the different phytoplankton species. In addition, Chl-*a* fluorescence measurements are also not a perfect proxy either, due to problems associated with converting the fluorescence signal into an accurate Chl-*a* concentration and issues relating to fluorescence quenching (Marra, 1997; Mignot et al., 2011).

2.7 Combined effects of climate and nutrients on phytoplankton biomass

It has long been recognised that combinations of nutrients (P and N) play an important role in the amplification of algal blooms in lakes (Paerl and Fulton, 2006; James et al., 2009; Cai et

al., 2012). This has been illustrated by various modelling studies (Elliott, 2005; Markensten et al., 2010), historical data analyses (Wagner and Adrian, 2009; Stich and Brinker, 2010) and experimental studies (Moss et al., 2003; McKee et al., 2003). However, it is also apparent that changes in local weather may also cause qualitative changes in phytoplankton community dynamics, with water temperatures playing a direct role in influencing algal growth (Stella et al., 2007). For instance, harmful algal blooms react positively to rising water temperatures (Paerl and Huisman, 2008; Cai et al., 2012; Deng et al., 2014). Many studies have reported that rising temperatures enhance cyanobacterial biomass compared to other phytoplankton species, such as diatoms and green algae (O'Neil et al., 2012; Deng et al., 2014; Rigosi et al., 2014). In addition to rising air temperatures, other aspects such as water temperatures increasing due to light absorption by algal blooms and increases in rainfall leading to nutrient runoff from surrounding catchments, also intensify symptoms of eutrophication in many freshwater lakes (Jeppesen et al., 2009; Moss et al., 2010; Deng et al., 2014).

2.8 Policy and management

In December 2000, the publication of the EU WFD (2000/60/EC) marked a turning point in the perception and management of water quality in Europe, establishing a framework for improving both surface and groundwater quality (Free et al., 2016). It is an important piece of environmental legislation outlining the requirements needed to carry out water protection and water management measures, which are integrated and targeted and which initially aimed to achieve an overall objective of at least 'good status' for Irish waters by 2015 or no further deterioration by 2027 (Cherry et al., 2008). The WFD was transposed into Irish Law by the European Communities (Water Policy) Regulations 2003 (SI No. 722 of 2003) in December 2003 (Earle and Blacklocke, 2008). The WFD runs in 6 year cycles up until the end of 2027 and is being implemented through river basin management planning (RBMP) which includes measures to address pollution from agricultural sources (WFD, 2000/60/EC). In the first six

year cycle of the RBMP which ran from 2009 and finished at the end of 2015, Ireland chose to undertake activities associated with the WFD through the creation of seven River Basin Districts (RBDs) (Earle and Blacklocke, 2008). Each RBD had individual RBMPs, which aimed to achieve the objectives of the WFD. It is acknowledged, however, that sufficient progress was not made in developing and implementing supporting measures during the first cycle of the RBMP (Department of Housing, Planning, Community and Local Government, 2018). According to the figures detailed in the RBMP for Ireland 2018 – 2021, 43% of river and 45% of lakes in the ROI did not meet their environmental objective in the first RBMP cycle (2009-2015) with 32% of rivers and lakes in Northern Ireland (NI) not achieving good ecological status (DAERA, 2018). Of those in the ROI, 53% of failures are suspected to be due to agricultural sources of pollution (Bradley et al., 2015; Deakin et al., 2016; Department of Housing, Planning, Community and Local Government, 2018). However, key measures which were implemented during the first cycle included the licensing of urban waste-water discharges (with an associated investment in urban waste-water treatment) and the implementation of the Nitrates Action Programme (Good Agricultural Practice Regulations) (Department of Housing, Planning, Community and Local Government, 2018). This current second cycle of the RBMP which is running from 2016 to the end of 2021, aims to build on the progress made and lessons learned from the first cycle (Department of Housing, Planning, Community and Local Government, 2018). Firstly, the structure of multiple RBDs did not prove effective, either in terms of developing the plans or the implementation of those plans. Secondly, the governance and delivery structures in place for the first cycle were not as effective as expected, and thirdly, the targets set for the first cycle were too ambitious and were not grounded on a sufficiently developed evidence base (Department of Housing, Planning, Community and Local Government, 2018). The current RBMP plan (2016 – 2021) has focused on combining the seven RBD's which were established in the first cycle to form a single RBD. In addition, the current RBMP has lists 190 Priority Areas for Action (PAA's), which include 726 waterbodies, where

collaborative efforts will be focussed to improve water quality, which will be assisted by the newly established Local Authority Waters Programme (LAWPRO). The LAWPRO Catchment Scientist will carry out as part of their work 'on the ground' actions and stream walks to gain a better understanding of the issues impacting on water quality in these PAA's with an aim to develop solutions that are relevant and workable locally.

In 1991, the EU introduced the Nitrates Directive (91/676/EEC). This is a precursor of the WFD (2000/60/EC) and aims to minimise surplus P and N losses from agricultural practices to the aquatic environment (Buckley, 2012). The Directive requires member states to designate all areas of land which drain into surface or groundwater's where NO₃⁻ exceeds 50 mg N L⁻¹, where nitrogen concentrations are showing a rising trend or where evidence of eutrophication and a significant amount of N present come from agricultural sources such as Nitrate Vulnerable Zones (NVZ) (Jordan and Smith, 2005). Giving effect to the Nitrates Directive in Ireland and supported by successive national regulations, the Nitrates Action Programme (NAP) was designed to prevent pollution of surface waters and groundwater from agricultural sources and to protect and improve water quality (Buckley, 2012; S.I. No. 605 of 2017). Subjected to a four year review cycle, the NAP's are now more commonly referred to as the Good Agricultural Practice (GAP) Regulations (Buckley, 2012).

One of the main focuses covered within the GAP Regulations, is outlining the minimum slurry storage requirements for housing livestock over the winter. This then aids in facilitating a closed slurry spreading period during the autumn and winter months and prevent incidental diffuse transfer of nutrients during heavy rainfall periods (Buckley, 2012). In Ireland, the voluntary agri-environmental scheme, Rural Environmental Protection Scheme (REPS), was introduced in 1992 and had a whole-farm focus. It evolved through four iterations (REPS 1, 2, 3 and 4) (Department of Agriculture, Food and Marine, 2017). The main purpose of the scheme was to

encourage farming in an environmentally friendly manner, placing much of the emphasis on biodiversity (Emerson and Gillmor, 1999; Feehan et al., 2005; Department of Agriculture, Food and Marine, 2017). Partial fencing of waterbodies was a requirement of this scheme, although cattle access to some stream segments for drinking water purposes was permitted (Department of Agriculture, Food and Rural Development, 2000; Feehan et al., 2005, VanRensburg et al., 2009). Initial uptake of the REPS scheme was slow, however, when it closed to new applicants in 2009, more than 60,000 farms were receiving payment, which represented 45% of farms in the ROI (Finn and Ó hUallacháin, 2012). It was succeeded, in 2010, by a smaller Agri-Environment Options Scheme (AEOS), which ran for three funding cycles until the end of 2013, taking a more targeted approach by focussing on part-farm actions (Department of Agriculture, Food and Marine, 2016). The current agri-environmental scheme introduced in the ROI is called Green Low-Carbon Agri-Environment Scheme (GLAS) (2014-2020). This scheme builds on the previous schemes primarily through encouraging farmers to protect water bodies from cattle by prohibiting cattle access to the waterways by implementing streamside fencing (Department of Agriculture, Food and Marine, 2015). Agri-environmental schemes similar to those run in the ROI have been adopted throughout the EU to help achieve the aims of the WFD (Batáry et al., 2015).

Chapter 3: Site description and methods

3.1 Introduction

This chapter provides a detailed description of the Milltown Lake Catchment study site. A general overview of materials and methods is given in this chapter, with greater detail outlined where applicable in their respective subsequent chapters. Methods which are described in this chapter include (1) where stream discharge and meteorological data was obtained, (2) how nutrients (TP, TN, soluble reactive phosphorus (SRP), NO₃-N, NO₂-N, NH₄-N and TDN) were analysed, (3) the deployment of the automatic sensor in Milltown Lake and (4) how chlorophyll *a* analysis was determined.

3.2 Site description

Milltown Lake (54°8'43"N; 6°42'35"W), sometimes referred to as Muchno Mills, has an area of approximately 0.14 km² (Figure 3.1; Figure 3.2a). The lake is relatively shallow, with a mean depth of 5.5 m and a maximum depth of approximately 10 m and has an estimated retention time of 25 days. The lake has one main inflow, the Drumleek River, and a single outflow, the Frankford River. The Drumleek River branches into three inflowing tributaries (440 m above lake) which are not named on the Ordnance Survey maps or locally (Figure 3.1a). Two of the larger tributaries of the Drumleek River are fed by small lakes. These include a small lake on the western tributary in Carnagh Forest, Tievenamara (54°12'4"N; 6°43'49"W) which is unnamed on the Ordnance Survey maps and Gentle Owen's Lake on the middle tributary (54°12'29"N; 6°42'45"W) (Carson, 2010). The lake is used as a domestic water supply for the local communities (abstracting approximately 286, 290 m³ per annum), which is typical of the small drumlin lakes that are found across the region. It is part of the Churchill and Oram Group Water Scheme (GWS) serving approximately 748 premises (Linnane et al., 2011). Recreational activities include angling, with the lake being stocked (rainbow trout and brown trout) on an annual basis for the local angling club.

Milltown Lake Catchment (54°8'44"N; 6°42'44"W) is situated in County Monaghan, northcentral Ireland (Figure 3.1a, b). The catchment is a sub-catchment of the River Fane, which is situated in the Neagh-Bann International RBD. The total area of the catchment including Milltown Lake is 30.6 km². Of this approximately 8 km² lies within Northern Ireland, providing a North-South cross-border element to the study (Linnane et al., 2011). The lake is an interdrumlin lake and is located in the drumlin belt of Ireland, which stretches from Strangford Lough in County Down to Clew Bay in County Mayo. The drumlin belt consists of a tightly packed hilly landscape, with many bogs and lakes found between the hills (Figure 3.2a, b). Soils within the catchment are mainly poorly drained gleys, with peat widely distributed throughout the northern section (Carson 2010). Land cover (based on CORINE data from 2006) is predominantly unimproved (62.5%) and improved (34.5%) pasture; however, small areas of forestry and peat bogs (3.0%) are also present in the upper section (EPA, 2016).





Figure 3.1 (a) Milltown Lake Catchment, Co. Monaghan, Ireland showing the western (fenced) and middle (unfenced) sub-catchments. Autosampler locations indicated by red dots, for the upper (north) and lower (south) sites on the fenced and unfenced tributaries. EPA staff gauge location indicated by green dot. Milltown Lake is shown at the bottom of the figure. (b) Milltown Lake, the orange dot indicates the sampling point (deepest point) within Milltown Lake, and EPA staff gauge location indicated by green dot.



Figure 3.2 (a) Milltown Lake and (b) drumlin landscape of Milltown Lake Catchment.

The main land use within the catchment is grassland agriculture, including intensive beef and dairy farming. According to the 2010 Irish census, 60% of farming in the county is dedicated to beef rearing and 24% to dairy farming (Monaghan Co. Council, 2012). The farming enterprises in the Milltown Lake Catchment are comprised of 67% beef, 20% mixed (beef and sheep) and 7.8% dairy (Carson, 2010). A farm survey carried out in the Milltown Lake Catchment in 2008 reported that 74% of farmers were farming land that bordered a tributary on the Drumleek River system or Milltown Lake itself (Linnane et al., 2011). The survey also asked farmers in the catchment specific questions regarding fertiliser and slurry spreading practices. Of the farmers that responded, 88% reported leaving a buffer or margin when spreading near watercourses, although the size of the buffer varied widely from field to field.

The use of this study location builds on previous work carried out by the National Source Protection Pilot Project (NSPPP) (2005-2010), which focused on source protection for the water supply for the local GWS (Linnane et al., 2011). As part of the NSPPP, the western tributary had fencing installed at a distance of 1.5 m from the stream bank in 2008 with the aim of excluding livestock from having direct access to the river (Figure 3.3a, b). As is typical with such fencing, this also allowed the re-establishment of streamside vegetation, creating a vegetative strip. This width of 1.5 m was the minimum distance required for fencing under the 2009 Good Agricultural Practice for Protection of Waters Regulations guidelines (S.I. 101 of 2009) in Ireland and was also the maximum width that farmers in the catchment would agree to (Linnane et al., 2011). The other two tributaries remained unfenced. The middle tributary was chosen as an unfenced comparison to the fenced tributary (see Chapter 4), due to its similar characteristics, farming practices and land uses (Table 3.1; Figure 3.1). A survey of the fencing prior to the commencement of the current study in 2010 identified one unfenced cattle access point (CAP) in the upper headwaters of the western tributary. Eighteen CAPs were identified on the middle unfenced tributary.



Figure 3.3 (a) Before fencing installed (b) after fencing (1.5 m from the stream bank) installed as part of the NSPPP in the Milltown Lake Catchment.

Characteristics	Catchment	Western tributary (fenced)	Middle tributary (unfenced)
Total area (km ²) excluding Milltown Lake	28.8	9.6	10.1
Tributary length (km)	24.6	8.4	6.3
Dwellings per tributary area (2010)	748	232	247
Cattle no. per tributary area (2010)	5554	1722	1833
*Upstream area above reach (km ²)		6.0	6.1
Upstream lake present		Yes	Yes
*Reach area (km ²)		3.6	4.0
*No. cattle access points in study reach		0	14
*Total length of cattle access points on reach (m)		0	48

Table 3.1 Characteristics of the western (fenced) and middle (unfenced) stream reaches, in the Milltown Lake Catchment.

* Reach refers to area between the upper and lower automated sampling sites on the tributaries as outlined in Figure 3.1a.

3.3 Stream discharge and meteorological data

In September 2005, an EPA staff gauge measuring stream discharge and water level on the Drumleek River (with an OTT Thalimede water level sensor) was installed in conjunction with the NSPPP and Monaghan County Council, approximately 500 m upstream (54°8'48"N; 6°42'15"W) from the inflow to Milltown Lake (Figure 3.1). Continuous stream discharge data (15 minute intervals) were available for this site from the Irish Environmental Protection Agency (EPA) Hydronet website (hydronet.epa.ie). The site was maintained by the EPA. Precipitation data for this study were obtained from the Coose weather station (10.4 km southwest for the Milltown Lake Catchment) and the Ballyhaise weather station (52 km south-west Milltown Lake Catchment) through Met Éireann (Table 3.2; Figure 3.4). Air temperature data were also available from Ballyhaise weather station. Additional data were obtained for the Clones Met Éireann station (42 km north-west) (Table 3.2; Figure 3.4). It should be noted that the Clones Met Éireann station closed in 2007.



Figure 3.4 Location of the Ballyhaise, Clones and Coose Met Éireann weather stations in proximity to Milltown Lake.

Table 3.2 Long term average meteorological data at Met Éireann weather stations near

 Milltown Lake.

Long term averages	Coose (2009 -2013)	Ballyhaise (2009 – 2013)
Air temperature	not measured	9.0 °C
Rainfall	1041 mm yr ⁻¹	1001 mm yr ⁻¹
Kaiiiiaii	1041 IIIII yi	

3.4 Nutrient analysis

All nutrient analyses (TP, TN, NO₃-N, NO₂-N, NO₄-N and SRP) were carried out colormetrically using a flow injection auto-analyser (Lachat Quickchem®, Lachat 7 Instruments, Loveland, Colorado, USA). Samples analysed for dissolved nutrients were filtered (GF/C filters nominal pore size 1.2 µm) on the day of sample collection and stored in pre-acid washed and Milli-Q water rinsed 100 ml polyethylene bottles. Total P and TN were analysed in unfiltered samples using the persulfate method (4500-P-J) for simultaneous determination of TP and TN (APHA, 2005). The QuickChem® methods used to determine TP, TN, SRP, NH₄-N, NO₃-N and NO₂-N concentrations are outlined in Table 3.3. Total dissolved nitrogen (TDN) was calculated as the sum of NH₄-N, NO₃-N and NO₂-N.

Parameter	QuickChem® Method	Detection Limit
ТР	10-107-04-4-B	0.007 mg P L ⁻¹
TN	10-115-01-4-B	$0.046 \text{ mg N L}^{-1}$
SRP	10-107-06-2-L	0.007 mg P L ⁻¹
NO4-N	10-107-04-1-R	$0.008 \text{ mg N L}^{-1}$
NO ₃ -N	10-115-01-1-V	$0.016 \text{ mg N L}^{-1}$
NO ₂ -N	10-115-01-1-V	0.016 mg N L ⁻¹

Table 3.3 QuickChem® Methods used to determine phosphorus and nitrogen.

Calibration standards for all of the above methods were prepared from separate 1000 mg L⁻¹ stock standard solutions from Inorganic[™] Ventures. These were sodium nitrate (NaNO₃) for TN and NO₃-N, sodium nitrite (NaNO₂) for NO₂-N, ammonium chloride (NH₄Cl) for NH₄-N and ammonium phosphate ((NH₄)₃PO₄) for TP and SRP analysis. In-house quality control check standards were prepared from analytical grade chemicals that were made from the same compound used in the calibration standards but purchased from a different manufacturer, i.e. Acros Organics[™]. Quality control check standards and laboratory and field blanks were included at the start of an analytical run and periodically, thereafter, to assess for instrumental drift. In the case of TP and TN analysis, the calibration standard, quality control standard, field and laboratory blanks were digested with potassium persulphate in the same manner as the samples prior to analysis on the Lachat flow injection auto-analyser.

3.5 Automatic sensor deployment in Milltown Lake and calibration

Additional data for water quality parameters were collected from Milltown Lake (Chapter 5 and 6) using automatic sensors located on permanent monitoring platforms. The Milltown Lake monitoring platform (deployed in November 2011) was anchored at a fixed position at the deepest point of the lake (~ 10 m) (Figure 3.1b). The instrumentation installed on the platform included two dissolved oxygen (D-Opto) sensors, at depths of 1 m and 7 m, which recorded DO data (mg L⁻¹ and % saturation) at 10 minute intervals. Data were downloaded from the DO sensors every 2 months. In addition, a chain of nine TidbiT temperature loggers (ONSET, Tidbit[®] v2 Temp Logger) were installed in Milltown Lake in April 2010 to allow for the

assessment of changes in physical structure. These sensors were mounted at 1 m depth intervals (1, 2, 3, 4, 5, 6, 7, 8 and 9 m) and recorded temperature data (°C) at 10 minute intervals. Data from the temperature loggers were collected using a HOBO[®] Shuttle-U-DTW-1 on a quarterly basis. All sensors were cleaned monthly during the winter, bi-weekly during spring and autumn and on a weekly basis during the summer, using a non-abrasive cloth and distilled water.

3.6 Chlorophyll-*a* analysis

Chlorophyll-*a* samples (1 L) were filtered through GF/C filters, nominal pore size 1.2 μ m. The filter papers were placed individually into 10 ml centrifuge tube and analysed on the day of sample collection. Extraction of Chl-*a* was carried out in the laboratory using a methanol (Emsure 96%) extraction method. Methanol (10 ml) was added to the centrifuge tube and heated at 75 °C for five minutes. Samples were allowed to cool to room temperature and filter papers were removed from the tubes. The remaining methanol was then centrifuged at 3500 rpm for 8 minutes. The clear extract was transferred to a cuvette and analysed using a Shimadzu Spectrophotometer (UV-1800), set to measure absorbance at 665 nm and at 750 nm against a reference cell filled with methanol. The Chl-*a* concentrations (mg L⁻¹) were calculated according to the following equation (Standing Committee of Analysts, 1983).

$$Chl-a = 11.9 \text{ x A x V}_e/V_s \text{ x d}$$
 (Equation 3.1)

Where A is the absorbance at 665 nm minus the absorbance at 750 nm of the extract; V_e is the volume of the extract (ml); V_s is the volume of the filtered sample (L); d is the path length of the optical cell (cm); and 11.9 is a coefficient. The factor 11.9 approximates to the reciprocal of the specific absorption coefficient at 665 nm for Chl-*a* in acetone (Standing Committee of Analysts, 1983). It includes an arbitrary assessment of absorbance contributed by other pigments (Standing Committee of Analysts, 1983).

3.7 Analysis of lake temperature profile data

Lake Analyzer is a visualisation tool for determining indices of mixing and stratification in lakes, providing standardised lake physical calculations including water density based upon temperature, thermal layers, thermocline depth, metalimnion depth, lake number, buoyancy frequency, Wedderburn number (Wn) and Schmidt stability (S) (Read et al., 2011) (http://lakeanalyzer.gleon.org/). Input data files required to run the programme include data on the lake hypsographic curve, water temperature profile, wind speed and a lake specific configuration file. The latter includes details of the lakes features such as maximum lake depth, height of wind speed instrument above the lake, and details of outliers for wind speed and temperature of the lake. The configuration file also outlines the data output resolution (daily/hourly) and an option on which figures to generate. For the current study lake temperature profile, S, buoyancy frequency, thermocline depth and Wn were used (Chapters 5 and 6).

3.8 Statistical methods

Detailed statistical analysis are outline in the subsequent chapters. Chapter 4 focuses on the nutrient export from a cattle based agricultural catchment and the effects of streamside fencing as a mitigation measure for the Milltown Lake Catchment. While Chapter 5 and Chapter 6 focuses on the local drivers of phytoplankton biomass during the spring/summer of 2012 and the spring of 2013 in Milltown Lake. Statistical analysis were carried out using the R statistical package version 3.2.2 (R Core Tea, 2015).

Chapter 4: The effects of streamside fencing on P and N concentrations in catchment streams of Milltown Lake

4.1 Abstract

Streamside fencing with vegetative buffer strips can be used as a mitigation measure to decrease nutrient and sediment loss from agricultural land to water and have been included in many European agri-environmental schemes. To date, limited research has been carried out within Europe on their effectiveness, particularly in western Europe which has a high-rainfall, oceanic climate. The work presented in this chapter investigated the effects of streamside fencing on stream water TP and TN concentrations in an intensive cattle-based agricultural landscape in north-east Ireland. The study was based on an upstream/downstream design and included a stream reach that had been fenced to exclude cattle five years before the study commenced (referred to as the fenced stream) and a separate unfenced stream reach on an adjacent tributary. The results indicated that exclusion fencing was effective in reducing TP and TN concentrations in the stream during the spring to autumn grazing period in 2011 (when slurry spreading was also permitted by Regulation). However, there was a significant increase in TP concentrations from upstream to downstream on this fenced reach during the following late winter/early spring, a period when slurry spreading was permitted by regulation, but cattle were still in winter housing. This behavioural difference was further evident in the differing concentrationdischarge relationships for TP for both catchments in the Grazing season and TN during the Open slurry spreading period. The difference between the two study periods may reflect interaction with buffer strip vegetation, which would be greatest in the grazing season but lower in the second study period.

4.2 Introduction

Identifying the areas in the landscape that pose the greatest risk of nutrient losses to surface waters is a critical component of nutrient management (Sharpley et al., 1993; Shore et al., 2014). These small defined areas, which contribute much of the catchment's sediment, P and N, are

known as critical source areas (Jorgensen, 2009) and can include sites where livestock have direct access to watercourses (McDowell and Srinivasan, 2009). Streamside fencing is used as a method to exclude livestock from waterways (Galeone, 2000; Vidon et al., 2008; Miller et al., 2010). Fencing is often used in conjunction with other agricultural management practices such as riparian buffers, and off-stream watering for livestock. It is mainly used to target and reduce suspended sediment and nutrient inputs to surface waterbodies from un-vegetated or sparsely vegetated stream banks, caused by livestock tramping and excess grazing (Belsky et al., 1999; Centner et al., 1999; Galeone, 2006; Miller et al., 2010; Miller et al., 2011). The main effect of fencing is the reduction of nutrients from livestock directly defecating and urinating within watercourses (Thomas, 2002; Miller et al., 2010). However, fencing also allows for the establishment of vegetation in riparian buffer areas, which further aids in reducing the nutrient and sediment input to streams through filtering overland discharge and retention of nutrients in biomass (Galeone, 2006; Melland et al., 2014; Haddaway et al., 2016). Riparian buffer zones are permanently vegetated areas located between agricultural fields and the watercourse to which they drain (Heathwaite et al., 1996; Haddaway et al., 2016). Dorioz et al. (2006) analysed the mechanism of typical grassland buffer strips on the specific behaviour of P and concluded that buffers have the capacity to decrease particulate P transfer from surface water discharge paths, increasing residence time and allowing for further biochemical processes to develop. However, Stoate et al. (2009) showed that in temperate northern climates (The Netherlands, Finland, Sweden and the UK), grassy buffer strips were ineffective during spring when vegetation was underdeveloped, and rain was most severe. Reichenberger et al. (2007) also showed that, during storm discharge events, buffer strips tend to be less effective as they are more effective when dealing with slow, shallow moving overland discharge. Shallow moving discharge provides maximum contact time for the removal of pollutants by several physical processes, including deposition and infiltration (Centner et al., 1999).

To date, there have been limited studies focusing on the effect of exclusion fencing in areas with intensive, cattle-based agriculture in western Europe, where management practices and climate differ from those in the US, Canada and Australia (Conroy et al., 2016; Bragina et al., 2017). In Ireland, these differences include longer periods when cattle graze in fields (van den Pol van Daselaar et al., 2015; Chen and Holden, 2016), land application of slurry for most of the year and high rainfall from south-westerly air-masses from the Atlantic Ocean (Nolan et al., 2010). As part of the consultation process with specific environmental authorities for the new RBMP 2018-2021 running in Ireland, the requirement of specific measures needed to target nutrient pollution from agriculture were highlighted (Department of Housing, Planning, Community and Local Government, 2018). Measures put forward included raising awareness and implementing streambank fencing with bankside vegetation as a viable option for restricting cattle access to water courses and reducing nutrient pollution from diffuse agricultural sources.

The bulk of annual P transfers are associated with sudden and infrequent high discharge events (Jordan et al., 2005; Jordan et al., 2007). However, Jordan et al. (2005; 2007) also noted that some rivers, particularly in rural Ireland, may also be maintained in a eutrophic state between storm events by low-level nutrient inputs from point sources, e.g. OSWTS. In the current study, in addition to determining the viability of streamside fencing as a management option for nutrient level reduction, nutrient concentration responses during infrequent high discharge events were examined to gain insight into the processes that were operating in both a fenced and an unfenced stream reach. This included the overall nutrient-discharge relationships, and a small number of targeted sampling campaigns for individual events. Previous studies examining concentration-discharge relationships have provided valuable information on the likely P and N sources (Ide et al., 2008; Stutter et al., 2008; Bowes et al., 2009; Abell et al., 2013). This concept is based on examining the difference in nutrient concentrations on the rising

and falling limb of the storm hydrograph (Bowes et al., 2015) (Figure 4.1). If the concentration of nutrients increases on the rising limb and decreases on the falling limb, the hysteresis plot produces a clockwise loop in the concentration-discharge relationship (Bowes et al., 2015) (Figure 4.1). This is considered to be caused by a quick flushing of sediment or nutrients to the waterbody which may become exhausted by the end of the storm event and therefore, indicates a source close to the stream (Williams 1989; Bowes et al., 2015). An anti-clockwise loop is indicative of high particulate or dissolved nutrient concentrations in the falling limb of the storm hydrograph, signifying the sediment peak is lagging the discharge, providing evidence of longer transit times of water and sediment (Williams 1989; Bowes et al., 2015) (Figure 4.1). This suggests that the nutrient source may be further from the stream. Finally, a figure-of-eight configuration combines both the clockwise and anti-clockwise configuration, and is considered to be related to a shift in the form of the relationship between discharge and sediment or nutrients, and therefore the source, during the course of a single flood event (Williams 1989; Bowes et al., 2015). However, Lloyd et al., 2016 noted that many hysteresis loops may be difficult to classify easily into these groups, and care should be taken with interpretation as the same type of loop could occur for different reasons. Nevertheless, the study of concentrationdischarge relationships during high discharge events using hysteresis can provide a valuable research tool to infer the likely contributing source areas and discharge pathways contributing to nutrient and sediment transport in catchments (Lloyd et al., 2016).



Figure 4.1 Storm hydrograph (www. geographyaslevelaqa.wordpress.com, 2016).

4.3 Aims

The primary aims of this chapter were to investigate the effects of agricultural management on P and N levels in catchment streams, including the effect of excluding cattle from stream waters, using fencing, as a mitigation measure. In addition, the chapter explored nutrient concentration/discharge relationships during different selected study time periods and selected discharge events to gain an insight into the processes operating in both the fenced and unfenced stream.

4.4 Methods and materials

Daily stream water sampling took place during three time periods between the 20th of April 2011 and the 16th of April 2012. Daily water samples were collected at four sampling points over three time periods that represented three main land management changes in the Irish agricultural calendar (Table 4.1). The main grazing period was further sub-divided into the Early Grazing (20th of April to 8th of June 2011) and Late Grazing period (9th of June to 31st of

October 2011), based on the expected time frame when grass for silage is normally cut in Ireland at the beginning of June. Best advice from Teagasc, the Irish agricultural authority, is that slurry spreading should take place immediately after silage is cut but before cattle are allowed to graze the grass (Teagasc, 2011) (Table 4.1). It is also important to note that in 2011, the period for slurry spreading was officially extended for two weeks beyond the GAP Regulations until the 31st of October due to abnormally high rainfall (Oireachtas, 2011).

the to of A	pm 2012.				
Study Period	Dates	n (days)	Agricultural Practices		
Grazing	20 th April to 31 st October 2011	176			
Early Grazing	20 th April to 8 th June 2011	50	Cattle grazing the land and slurry spreading permitted under the GAP regulations		
Late Grazing	9 th June to 31 st October 2011	126			
Closed	1 st November to 17 th November 2011	17	Slurry spreading prohibited under the GAP regulations and cattle still in housing		
Open	4 th February to 16 th April 2012	63	Slurry spreading permitted under the GAP regulations and cattle still in housing		

Table 4.1 Field sampling categories and dates for this study between the 20th of April 2011 to the 16th of April 2012.

4.4.1 Site selection

An initial scoping study was carried out on all three tributary streams between the 11^{th} of May and 2^{nd} of September 2010. This study determined which of the two unfenced (middle and eastern) streams would be suitable for comparing with the previously fenced (western tributary) (Figure 3.1a). The study involved an initial water sampling programme and a catchment inspection. The fenced tributary was also walked to inspect for any CAPs. One potential CAP (~1.5 m), was identified, where a farmer had not hung up a gate, thus providing a potential CAP. Both stream reaches were also surveyed to ensure that there were no major inflows to the reach. During the scoping study, weekly grab samples were collected on a Tuesday (17 sampling trips) at sites that had previously been used in the NSPPP (Linnane et al., 2011) along the western, middle and eastern tributaries (Figure 3.1a). The eastern tributary flowed directly through one farmyard area, a potential point source of nutrients, and was therefore, considered unsuitable for the current study. Based on the findings of the scoping study, an upper and lower site were selected on both the western fenced and the middle unfenced tributary (Figure 3.1a). The upper site on the fenced tributary was approximately 950 m downstream from a single CAP identified and a small area of forestry, elevated concentrations of TP were identified directly downstream from this location. Therefore, the study focused downstream from this location. These final two selected stream reaches had similar land practices, along with similar slope profiles, therefore, allowing for the best comparison of the effects of fencing on the western tributary (Table 3.1; Figure 4.2). Automated samplers were placed at the selected sites. The percentage of subcatchment area draining into the stream was calculated for each stream reach (fenced: 3.6 km²; unfenced: 4.0 km²) (Table 3.1).



Figure 4.2 Distance from the source of the western, middle and eastern tributaries to Milltown Lake.

4.4.2 Sample collection and nutrient analysis

Automated samplers (ISCO® 6700, HACH® Sigma 900 Max and HACH® Sigma Streamline 800SL) were placed at the four selected sampling sites. Samplers were programmed to collect

six 50 ml samples per day (at 4 hour intervals), giving a final daily sample volume of 300 ml. The polyethylene sample bottles were pre-acid washed and rinsed with Milli-Q water. Sample bottles were collected weekly and stream discharge was measured at each of the four sites on that occasion (Global Water Flow Pro FP111). Continuous discharge data was obtained from the EPA staff gauge at the base of the Drumleek River (Section 3.3). Daily discharge rates at the four sampling sites were estimated from this data using the percentage catchment area draining into the stream at particular sampling points, the drainage area-ratio method (Archfield and Vogel, 2010). These estimated discharge rates showed a good correlation with the discharge rates measured at each site on the sampling dates (r =0.95; p < 0.0001 (fenced upper site); r = 0.93; p < 0.0001 (fenced lower site) r = 0.92; p < 0.0001 (unfenced upper site) and r = 0.87; p < 0.0001 (unfenced lower site)). Conductivity (μ S cm⁻¹), pH, DO in both (mg L⁻¹) and (% SAT) were measured using a YSI[®] 556 Handheld Multi-parameter Instrument or YSI Professional Plus Multi-parameter Instrument. The portable YSI probe was calibrated in the laboratory before carrying out field work and assessed periodically throughout the sampling day.

In addition, four high discharge events were sampled over a 24 hr period at the lower sites of the fenced and unfenced tributaries. These events included one during the main grazing season, in 2010 and 2011, and the first high-discharge event in autumn in each year. High discharge events were sampled, with the use of the above automated samplers, at the lower sites on the fenced and unfenced tributaries. Samplers were programmed to collect a 300 ml sample at hourly intervals, over the 24 hr sampling period. The sample bottles were treated in the same manner as stated above, prior to sample collection. Stream discharge, conductivity (μ S cm⁻¹), pH, DO (mg L⁻¹ and % SAT), were measured at each of the two sites, before and after each event.

Nutrient analyses were carried out as described in Section 3.4. Total P and N concentrations were measured on all samples collected from the weekly sample collection and all samples collected for the high discharge events. Samples for dissolved nutrients were measured on the final two days of the weekly sampling programme and on all samples collected for the high discharge events, based on permitted holding times for analysis.

4.4.3 Data analysis

Statistical analysis were carried out using the R statistical package version 3.2.2 (R 195 Core Team, 2015). Significant differences in nutrient concentrations between the fenced and unfenced tributaries were determined using the non-parametric Mann Whitney U test (Helsel and Hirsch, 1992; Zuur et al., 2007) using the mgcv package (Wood, 2006) in R (version 3.2.5, R 195 Core Team, 2015). Generalized additive modelling (GAM) was used to determine the main climatic and hydrological drivers of change in TP and TN concentration for the lower sites in the fenced and unfenced streams, and for each of the management periods. The GAM analyses were carried out using the mgcv package in R (Wood, 2006). The residuals of the initial model and final model were checked for any breach of the assumption of heterogeneity using gam.check function in R. The final model was also checked for any breach of the assumption of independence by checking the autocorrelation function of the model residuals. The optimum or best fit GAM models were selected using the lowest Akaike Information Criterion (AIC) (Akaike, 1974; Zuur et al., 2009). The independent variables assessed included discharge (log transformed), air temperature, solar radiation (obtained from the Ballyhaise Met Éireann weather station) and rainfall (obtained from the Coose Met Éireann weather station).

Plots of relationships between concentration (TP, SRP, particulate P, TN and TDN) and stream discharge for each high discharge event were inspected to identify hysteresis that reflected consistent differences in the concentration between the rising and falling limbs of the

hydrograph (Bowes et al., 2015; Lloyd et al., 2016). The distinct features looked for were (1) clockwise, a quick flushing of sediment or nutrients to the waterbody which may become exhausted by the end of the storm event, therefore indicating a source close to the stream, (2) anti-clockwise, indicating high particulate or dissolved nutrient concentrations in the falling limb of the storm hydrograph, signifying that the sediment peak is lagging the discharge and (3) figure of eight behavior, considered to be related to a shift in the form of the relationship between discharge and sediment or nutrients, and therefore the source, during the course of a single flood event (Williams 1989; Bowes et al., 2015). Observed hysteresis was subsequently characterised for all four high discharge sampling events at the lower sites on the fenced and unfenced tributary.

4.5 Results

4.5.1 Meteorological and hydrology data

The weather during the sampling period was characterised by generally mild and wet conditions, typical for the region. The mean daily air temperature at the Ballyhaise weather station for the period between 20th of April 2011 to the 16th of April 2012 was 8.8 °C. However, the maximum of 17.5 °C was actually recorded on the 03rd of June 2011, with a minimum of -2.0 °C on the 2nd of February 2012 (Figure 4.3a). Mean daily air temperature at the Ballyhaise weather station for the Grazing study period was relatively low at 11.0 °C (mean of 9.5 °C for the Early Grazing and 11.5 °C for the Late Grazing period). During the Closed and Open periods, the mean daily air temperatures were 8.8 °C and 6.9 °C respectively (Table 4.2). Total annual and mean daily rainfall at the Coose weather station for the study period were 994 mm yr⁻¹ and 2.7 mm day ⁻¹ respectively, with a maximum rainfall of 32.7 mm day ⁻¹ on the 11th of November 2011 (Figure 4.3b). Mean daily rainfall for the Grazing period, while the values for the Early Grazing and 3.2 mm day ⁻¹ for the Late Grazing period), while the values for the sampling days in the Closed and Open periods were 4.2 mm day ⁻¹ and 1.2 mm day ⁻¹, respectively (Table 4.2). Overall, there were 179 wet days (of 256 total sampling days)

recorded (days $\geq 1 \text{ mm of rain}$) during the overall study period. The wettest month was October 2011 (151 mm month⁻¹) while the driest month was March 2012 (19 mm month⁻¹) (Figure 4.3c).

Daily stream discharge at the EPA staff gauge on the Drumleek River ranged from a low of 0.04 m³ sec⁻¹ on the 4th of August 2011 to 8.81 m³ sec⁻¹ on the 25th of October 2011 (Figure 4.3b). Mean stream discharge for the Grazing period was 0.39 m³ sec⁻¹ day⁻¹, that for the Closed period sampled was 0.84 m³ sec⁻¹ day⁻¹, whilst the Open period had the lowest mean 0.24 m³ sec⁻¹ day⁻¹ (Table 4.2). There was a weaker relationship between monthly rainfall and stream discharge between April to September 2011, when air temperature (and thus, evapotranspiration) was higher (r = 0.75) (Figure 4.3c). It is important to note that the Milltown Catchment is located on a poorly productive aquifer and has low permeability soil which include gleys and peat, the potential for loss to groundwater is low. In contrast, the correlation coefficient between rainfall and stream discharge during autumn and winter months was significantly higher at 0.92 (Figure 4.3c). No significant correlation was found between daily rainfall and stream discharge in the Grazing season, indicating that the precipitation which fell was evapotranspired.



Figure 4.3 (a) daily mean air temperature (°C) at Ballyhaise Met Éireann weather station, **(b)** daily rainfall (mm day⁻¹) at Coose Met Éireann weather station and daily stream discharge ($m^3 \sec^{-1}$) for the EPA staff gauge site on the Drumleek River, **(c)** Monthly rainfall (mm month⁻¹) (grey bars) and average monthly stream discharge data ($m^3 \sec^{-1} \mod^{-1}$) (continuous black line) for the study period (April 2011 – April 2012). r = Pearson correlation for April to September 2011 and October 2011 to April 2012.

Period	Mean daily air temperature (Ballyhaise)	Mean daily rainfall (Coose)	Cumulative rainfall (Coose)	CumulativeMax dailyrainfallrainfall(Coose)(Coose)		n
	(°C)	(mm day ⁻¹)	(mm)	(mm day ⁻¹)	$(m^3 \text{ sec}^{-1})$	
Overall	8.8	2.7	994	32.7	0.57	362
Grazing	11.0	2.9	563	31.6	0.39	176
Early Grazing	9.5	2.0	98	13.5	0.08	50
Late Grazing	11.5	3.2	465	31.6	0.49	126
Closed	8.8	4.2	71	32.7	0.84	17
Open	6.9	1.2	78	11.6	0.24	63

Table 4.2 Summary of meteorological data (Ballyhaise and Coose Met Éireann weather station) and hydrological data (EPA staff gauge on the Drumleek River) for the main study periods. n denotes the number of days in each period.

4.5.2 General nutrient concentrations at the lower sites of the fenced and unfenced tributary Total P concentrations measured at the lower sites on both the fenced and unfenced stream reaches were high (Figure 4.4). Peaks in concentration were observed throughout the sampling periods and were generally related to high discharge events. During the Grazing season, the fenced tributary had the highest peaks in concentration, in general, with the exception of one very high TP concentration (1.43 mg TP L⁻¹) on the 12th of June 2011 for the lower site on the unfenced reach (Figure 4.4c). It is important to note that this spike in concentration was apparent for all nutrient parameters measured on that day and may be attributed to the runoff from land due to recent rainfall, from slurry spreading on freshly cut silage fields. The fenced tributary also had the higher TP concentrations throughout the Closed period (Figure 4.4e). Total P concentrations in the Open period were marginally higher for the unfenced tributary, with three well defined high concentration peaks seen towards the early stages. However, toward the end of the Open period there was a shift, with TP concentrations higher in the fenced tributary (Figure 4.4g).

Peaks in TN concentration were observed throughout the period, once again commonly related to high discharge events (Figure 4.4). During the Grazing season, for example, the highest peaks for the two lower sites on the stream reaches occurred at the downstream site on the unfenced tributary, with two very high concentrations (6.57 mg TN L⁻¹ and 4.97 mg TN L⁻¹) on the 12^{th} of June 2011 and 19^{th} of June 2011 (Figure 4.4d). Total N concentrations were higher in the unfenced tributary throughout the Closed period (Figure 4.4f). During the Open period, TN concentrations were generally higher in the unfenced tributary, peaking on the 18^{th} of March 2011 (Figure 4.4h). However, a peak in TN concentration was seen in the fenced reach (2.00 mg TN L⁻¹) on the 19^{th} of February 2011, which was not reflected in the unfenced stream reach (Figure 4.4h). This peak in TN concentration coincided with a concentration peak in TP for both stream reaches (Figure 4.4g).



Figure 4.4 Daily TP concentration (mg TP L^{-1}) during the (**a**) Early Grazing, (**c**) Late Grazing, (**e**) Closed and (**g**) Open periods and daily TN concentration (mg TN L^{-1}) during the (**b**) Early Grazing, (**d**) Late Grazing, (**f**) Closed and (**h**) Open periods, for the lower sites on the fenced (black) and unfenced (grey) tributaries. Stream discharge from the EPA gauge is shown as grey bar. Note difference in scales between TP and TN figures.

4.5.3 Upstream to downstream nutrient concentration change

The changes in TP and TN concentration from upstream to downstream varied for both management period and type of nutrient. It was of note, also, that the median TP concentrations for the total study period were higher for the fenced stream reach at the upstream and downstream sites (both were 0.091 mg TP L⁻¹) than for the unfenced stream at the upstream (0.057 mg TP L⁻¹) and downstream (0.060 mg TP L⁻¹) sites. In contrast, both sites on the unfenced tributary had greater TP concentration ranges, in particular for the upper limit: (upstream: 0.012 - 0.127 mg TP L⁻¹ and downstream: 0.010 - 0.106 mg TP L⁻¹). The equivalent ranges for the fenced tributary sites were 0.011 - 0.128 mg TP L⁻¹ (upstream) and 0.012 - 0.125 mg TP L⁻¹ (downstream). For the dates on which SRP data were available (weekly intervals), SRP accounted for, on average, 46% of the TP fraction for both the upper and lower sites on the fenced tributary and 46% for the upper and 44% for the lower sites on the unfenced tributary, indicating that c. 55% of TP was in particulate form at all four sites.

In the Early Grazing period, the two stream reaches differed in the pattern of change in TP concentrations as water travelled between the upper and lower autosampler sites. Despite the higher overall TP concentrations for the fenced reach, there was a significant decrease in TP concentration on this reach (median for upper site: 0.110 mg TP L⁻¹; median for lower site 0.065 mg TP L⁻¹; p = <0.001) between the upstream and downstream sites (a distance of 3.1 km), indicating a net retention of TP over this stream reach. Conversely, there was a significant increase between the upstream (0.034 mg TP L⁻¹) and downstream (0.060 mg TP L⁻¹) sites on the unfenced reach (a distance of 2.3 km) (p = <0.001), indicating net release of TP along this reach (Table 4.3; Figure 4.5a, e). This pattern, of a significant decrease in concentration as water travelled down the reach, was also apparent for the fenced site in the Late Grazing period, with median concentrations of 0.104 mg TP L⁻¹ at the upstream site and 0.091 mg TP L⁻¹ at the downstream site (p = <0.001). In contrast, the median TP concentration was again significantly

higher at the lower unfenced site than at the upstream site for that reach (0.057 mg TP L⁻¹ upstream; 0.082 mg TP L⁻¹ downstream to) (p = <0.001) indicating increased input of TP as the water travelled along the reach (Table 4.3; Figure 4.5b, f). During the subsequent short (2.5 weeks) sampling period in the Closed period, when slurry spreading was prohibited and cattle were also housed, there was no significant difference between TP concentrations at the upstream and downstream sites for either of the two stream reaches (Table 4.3; Figure 4.5c, g).

The patterns in upstream-downstream median TP concentration between the two streams did not persist through to the Open spreading period. During this sampling period (slurry spreading was again permitted, and cattle were still housed), the median TP concentration on the fenced tributary was lower for the upstream site (0.057 mg TP L⁻¹) compared to the equivalent site on the unfenced tributary (0.110 mg TP L⁻¹). However, the median concentration at the downstream site on the fenced tributary was now statistically higher that that at the upstream site, that is the concentration increased as water travelled down the reach (median for upper site: 0.057 mg TP L⁻¹; median for lower site 0.079 mg TP L⁻¹; p = <0.05), while the upstream site on the unfenced tributary was statistically lower (median for upper site: 0.110 mg TP L⁻¹; median for lower site 0.060 mg TP L⁻¹; <0.001) (Table 4.3; Figure 4.5d, h).

	Period		n	Median (mg L ⁻¹)	Q (25 th)	Q (75 th)	W	Р
ТР	Total study	US	256	0.091	0.011	0.128	9534	< 0.001
	Early Crazing	US	236 50	0.091	0.012	0.125	1725	<0.001
		DS	50	0.065	0.040	0.104	1723	<0.001
	Late Grazing	US	126	0.104	0.086	0.124	0480	<0.001
Fenced	(& extension period)	DS	126	0.091	0.070	0.124	9409	<0.001
	Closed	US	17	0.103	0.083	0.128	140	ns
		DS	17	0.095	0.086	0.125	110	
	Open	US	63	0.057	0.048	0.081	1531	< 0.05
	open	DS	63	0.079	0.055	0.111	1001	-0.05
	Total study	US	256	0.057	0.012	0.127	11062	< 0.001
		DS	256	0.060	0.010	0.106	11002	
	Farly Grazing	US	50	0.034	0.027	0.053	786	<0.001
		DS	50	0.060	0.046	0.081	700	-0.001
ТР	Late Grazing	US	126	0.057	0.041	0.101	5553	< 0.001
Unfenced	(& extension period)	DS	126	0.082	0.062	0.106	5555	
	Closed	US	17	0.055	0.049	0.061	116	ns
		DS	17	0.059	0.053	0.067		
	Open	US	63	0.110	0.061	0.127	2738	<0.001
		DS	63	0.060	0.047	0.099	2130	<u>\0.001</u>
	Total study	US	256	1.000	1.000	1.630	10911	< 0.001
		DS	256	1.030	1.030	1.483		
	Forty Grazing	US	50	1.160	0.950	1.265	1507	ns
		DS	50	0.994	0.762	1.202		
TN	Late Grazing	US	126	1.425	1.183	1.630	10146	< 0.001
Fenced	(& extension period)	DS	126	1.225	0.919	1.483	10140	
	Closed	US	17	0.691	0.644	0.711	84	<0.05
		DS	17	0.718	0.693	0.782	04	~0.05
	Open	US	63	0.943	0.822	1.163	1278	<0.001
	Open	DS	63	1.114	0.988	1.282	12/0	~0.001
	Total study	US	256	0.093	0.930	1.979	12085	<0.001
TN Unfenced		DS	256	1.090	1.090	1.390	12085	~0.001
	Forly Grazing	US	50	0.699	0.597	1.004	583	<0.001
		DS	50	1.020	0.809	1.310		~0.001
	Late Grazing (& extension period)	US	126	0.911	0.701	1.298	559	< 0.001
		DS	126	1.120	0.913	1.390		
	Closed	US	17	0.968	0.923	1.000	64	< 0.05
		DS	17	1.020	0.994	1.090		
	Open	US	63	0.902	0.789	1.979	044	<0.001
		DS	63	1 160	0 956	1 310	744	~0.001

Table 4.3 Median TP and TN concentrations for the fenced and unfenced, upstream (US) and downstream (DS) sites (mg L⁻¹) during the Total, Early and Late Grazing, Closed and Open time periods, together with 25th and 75th percentiles. Significance levels for the Mann Whitney U test are also presented.



Figure 4.5 Median TP concentrations for the fenced (**a**, **b**, **c** and **d**) and unfenced (**e**, **f**, **g** and **h**) upstream (US) and downstream (DS) sites (mg TP L⁻¹) for the Early Grazing (EG, n = 50), Late Grazing including extension period (LG.ex, n = 126), Closed (C, n = 17), and Open (O, n = 63) periods. The whiskers indicate the maximum and minimum concentrations.

Median TN concentration for the total study period for the upper site on the fenced tributary (1.00 mg TN L⁻¹) was slightly higher (by 1.08 times) than the equivalent site on the unfenced tributary (0.93 mg TN L⁻¹). However, this relative difference was lower than that for TP. At the lower sites, the fenced tributary had a slightly lower median TN concentration (1.03 mg TN L⁻¹) compared to the unfenced tributary (1.09 mg TN L⁻¹). This difference was also apparent in the ranges for the upper and lower sites of the two tributaries (fenced: upper site 1.00 mg TN L⁻¹ and fenced lower site 1.03 mg TN L⁻¹) which were again greater for the unfenced tributary (unfenced: upper site 0.93 mg TN L⁻¹ and lower site 1.09 mg TN L⁻¹). On the sampling dates on which TDN data were available, TDN accounted for on average a higher percentage of the total load than was observed for TP: 74% of the TN fraction for the upper and lower sites on the unfenced tributary.

For the Early Grazing period, the relative difference in TP concentration that was observed between the upstream and downstream sites on the fenced reach was not apparent for TN, that is, there was no significant difference between the concentration for upstream site (1.160 mg TN L⁻¹) and that for the downstream site (0.994 mg TN L⁻¹) (Table 4.3). In contrast, the lower site on the unfenced tributary (1.020 mg TN L⁻¹) had a significantly higher median TN concentration than the upstream site (0.699 mg TN L⁻¹) (p = <0.001) (Table 4.3; Figure 4.6a, e).

In the Late Grazing period, there was a slight but significant decrease in concentration between the upper (1.425 mg TN L⁻¹) and lower sites (1.225 mg TN L⁻¹) on the fenced reach (p = <0.001). In contrast, there was a slight but significant increase in concentration for the two sites on the unfenced reach (from 0.911 mg TN L⁻¹ to 1.120 mg TN L⁻¹ (p = <0.001) (Table 4.3; Figure 4.6b, f). During the Closed spreading period, when livestock were also housed, the unfenced reach had a median TN concentrations of 0.968 mg TN L⁻¹ upstream and 1.120 mg TN L⁻¹ downstream while those for the fenced reach were 0.691 mg TN L⁻¹and 0.718 mg TN L⁻¹ upstream and downstream respectively (Table 4.3). This slight increase in concentration was significant for both the fenced and unfenced reach (Table 4.3; Figure 4.6c, g).

During the Open period and again in contrast to TP, there was net increase in TN concentrations from the upper to the lower sites (p = <0.001) (Table 4.3; Figure 4.6d, f). The median TN concentration for the upper site on the fenced tributary was 0.943 mg TN L⁻¹ compared to 1.114 mg TN L⁻¹ at the lower site (Table 4.3). The unfenced stream reach also indicated a significant increase in concentration and therefore, net release during the Open period (upper site 0.902 mg TN L⁻¹; lower site 1.160 mg TN L⁻¹). In summary, the pattern of change in total nutrient concentrations included a significant downstream decrease in both TP and TN concentration for the fenced reach during the Grazing season, with, in contrast, significant increases in both TP and TN concentrations for the unfenced reach during the same period. These patterns reversed for TP only during the Open period, with a significant downstream increase in TP concentration for the fenced reach and a significant decrease in TP concentration for the unfenced reach. Both reaches had a significant downstream increase in TN concentration during the Open period.



Figure 4.6 Median TN concentrations for the fenced (**a**, **b**, **c** and **d**) and unfenced (**e**, **f**, **g** and **h**) upstream (US) and downstream (DS) sites (mg TN L⁻¹) for the Early Grazing (EG, n = 50), Late Grazing including extension period (LG.ex, n = 126), Closed (C, n = 17), and Open (O, n = 63) periods. The whiskers indicate the maximum and minimum concentrations.

Overall, the annual TP and TN loads estimated based on the daily concentration data from this study were lower for the fenced (0.57 kg TP ha⁻¹ yr⁻¹ and 5.99 kg TN ha⁻¹ yr⁻¹) tributary compared to that of the unfenced tributary (0.77 kg TP ha⁻¹ yr⁻¹ and 7.24 kg TN ha⁻¹ yr⁻¹). The
estimated annual loads exported from the fenced catchment represented 74% and 83% of the loads from the unfenced catchment for TP and TN respectively.

4.5.4 Drivers of nutrient concentrations in the fenced and unfenced tributary

The optimum GAM for the TP concentration data at the lower site on the fenced tributary during the overall Grazing period (20th of April to 30th of October 2011) had one significant smoother (log transformed stream discharge – L.Flow) but this explained only 12% of deviance (Table 4.4; Figure 4.7a). The smoother describing the relationship between TP concentration and L.Flow for the fenced tributary indicated higher values at both low and high discharges (Figure 4.7a). Other weather related variables e.g. air temperature and rainfall were not significant. In contrast, the optimum GAM for TP concentration in the unfenced reach during the same period included two smoothers (L.Flow and air temperature) and explained 23% of variance (Table 4.4; Figure 4.7b; Figure 4.8). Here, the smoother describing the relationship between TP concentration and L.Flow had a very different pattern to that for the fenced reach, with low TP concentrations at low stream discharge, and a generally linear increase at higher discharges (Figure 4.7b).

The only significant term in the optimum model for the unfenced reaches in either time period was L.Flow. In contrast to TP, the smoothers between TN concentration and L.Flow for the Grazing period indicated similar relationships for both the fenced and unfenced reaches (Figure 4.7c, d), which were also relatively similar in pattern to that for TP concentration for the fenced reach. The smoothers describing the relationship indicated high TN concentration associated with both low stream discharges and at higher discharges, with a decrease at low to moderate discharges. The deviance explained by L.Flow was 19% for the fenced tributary and 13% for the unfenced tributary (Table 4.4; Figure 4.7c, d). The relationships between TP and L.Flow were not significant for either site during the Open period (14th of February to 16th of April

2012). However, the optimum GAMs for TN again indicated differing concentration-discharge behaviours between the fenced and unfenced tributaries. The smoother describing the relationship between TN concentration and L.Flow for the fenced tributary explained 27% of the deviance (Table 4.4). There was no clear direction for this relationship during lower and medium discharge periods, however, as discharge increased above 0.3 m³ sec⁻¹, TN concentration then decreased (Figure 4.7e). In contrast, and similar to TP in the Grazing season, TN concentrations gradually increased with stream discharge for the unfenced tributary during the Open period (Figure 4.7f). The optimum GAM explained 18% of the deviance (Table 4.4).

Table 4.4 Generalized additive models for the lower sites on the fenced and unfenced tributary for TP and TN concentrations (mg L^{-1}), during the grazing period including extension period and TN concentration (mg L^{-1}) for the Open period.

Period		Site	parameter	Ν	edf	Р	R ² adj	% dev
		Fenced	L.Flow	176	4.11	0.0024	0.10	12.0
	TP	Unfenced	L.Flow	176	5.73	0.0235	0.19	22.8
Grazing			Air_Temp	176	3.10	0.0164		
	TN	Fenced	L.Flow	176	4.81	2.41e-05	0.16	18.6
		Unfenced	L.Flow	176	4.30	0.0012	0.10	12.6
Open	TN	Fenced	L.Flow	63	7.19	0.0566	0.17	26.7
		Unfenced	L.Flow	63	1.92	0.0095	0.15	17.9



Figure 4.7 Estimated smoothing curves in the Grazing periods including extension period for TP concentrations (a) L.Flow fenced, (b) L.Flow unfenced and TN concentrations (c) L.Flow fenced, (d) L.Flow unfenced, lower reaches. Estimated smoothing curves in the Open period for TN concentrations (e) L.Flow fenced and (f) L.Flow unfenced, lower reaches. The solid line is the smoother and dotted lines indicate the 95% confidence intervals.



Figure 4.8 Estimated smoothing curves in the Grazing period including extension period for TP concentrations, for Air_Temp unfenced lower reach. The solid line is the smoother and dotted lines indicate the 95% confidence intervals.

4.5.5 Concentration-discharge relationships at a sub-daily scale

During the Grazing season of 2010 and 2011, nutrient concentrations over the time period of four high discharge events (11th- 12th of July 2010, 7th - 8th of September 2010, 25th - 26th of June 2011 and 2nd - 3rd of October 2011) were sampled at the lower sites on the fenced and unfenced tributary (Table 4.5; Figure 4.9). These included the first large high discharge event in autumn after a drier period in each of the two years, and one mid-summer but lower magnitude discharge event. A summary of the weather and stream discharge data for the high discharge events sampled in 2010 and 2011 is given in Table 4.5.

Table 4.5 Summary of meteorological data (Coose Met Éireann weather station) and hydrological data (EPA staff gauge) for the four high discharge events sampled during the grazing season in 2010 and 2011.

Date	Cumulative rainfall (7 day previous) (mm)	Cumulative rainfall (during event) (mm)	Max stream discharge (during event) (m ³ sec ⁻¹)	Mean stream discharge (during event) (m ³ sec ⁻¹)
11 th – 12 th July 2010	10.8	8.4	1.4	0.3
7 th – 8 th Sep 2010	18.1	90.8	11.9	6.2
25 th – 26 th June 2011	15.6	18.9	0.5	0.2
$2^{nd} - 3^{rd}$ Oct 2011	7.4	36.4	4.6	2.8



Figure 4.9 Stream discharge (m³ sec⁻¹) from the EPA gauge is shown as grey line and daily rainfall (mm) from Coose Met Éireann weather station for (**a**) 2010 and (**b**) 2011. The events circled indicate the high discharge event sampled. The area between the dashed black lines indicates the Grazing period.

No consistent differences between the fenced and unfenced tributaries in the concentrationdischarge patterns emerged across the hysteresis graphs for the four events sampled. However, there were some clear differences in behaviour, especially for the July and September 2010 events (Figure 4.10; Figure 4.11). During the initial mid-summer lower discharge event in July 2010 (maximum discharge = $1.4 \text{ m}^3 \text{ sec}^{-1}$), TP concentrations were generally higher for the fenced tributary site (mean TP concentration: fenced 0.18 mg L⁻¹; unfenced 0.14 mg L⁻¹) (Table 4.6; Figure 4.10). Total P concentration in both tributaries increased rapidly on the rising limb of the event, indicating an initial washout of TP as discharge increased. While concentrations then fell on the falling limb for the unfenced site (clockwise hysteresis), the fenced tributary had a second increase in TP concentrations midway through the event on the falling limb, as indicated by a figure-of-eight pattern (Table 4.6; Figure 4.10). For both tributaries, the particulate P concentration displayed a clockwise pattern, indicating high particulate P washout with rising stream discharge and then a drop in concentration as stream discharge decreased (Table 4.6; Figure 4.10). The concentration range for both sites was similar for SRP, but the pattern again differed, with a figure-of-eight pattern being observed for the fenced site, suggesting a second pulse in SRP being delivered to the stream, while there was a clockwise pattern for the unfenced site with falling concentrations as discharge fell (Figure 4.10).

The September event in 2010 was one of the highest discharge events recorded for the Drumleek River discharge gauge site, reaching 12.8 m³ sec⁻¹ at its peak (Figure 4.9a). It came after a relatively dry period in August and early September. Both tributaries had an initial flush of TP from the catchment as the stream discharge increased, with concentration increasing from 0.10 mg TP L⁻¹ to a maximum of 0.17 mg TP L⁻¹ for the fenced tributary and from 0.11mg TP L⁻¹ to 0.18 mg TP L^{-1} for the unfenced tributary. The pattern for the fenced tributary showed a clockwise hysteresis, with concentrations decreasing on the falling limb. In contrast, TP concentrations for the unfenced tributary showed a figure-of-eight pattern, characterised by a further increase in TP concentration as discharge receded, and with the highest concentration (0.18 mg TP L⁻¹) occurring at moderate discharges (Figure 4.11a, d). For the particulate P fraction, concentration increased rapidly with discharge for both sites, indicating an initial washout of particulate material. Total P concentrations then again decreased at both sites as discharge receded, with a generally clockwise hysteresis, although it is of note that concentrations remained higher on the falling limb for the unfenced tributary than for the fenced tributary. However, the two sites again had different hysteresis patterns for SRP concentrations in this event (Figure 4.9). Interestingly, SRP concentration rose more rapidly with increasing stream discharge at the fenced tributary site, with further increases in concentration occurring on the falling limb (again a figure-of-eight pattern). Soluble reactive phosphorus concentrations at the unfenced tributary were slow to increase from baseflow to peak flow, after which they had a clear anti-clockwise direction with highest concentrations this time coinciding with the falling limb of stream discharge (Table 4.6; Figure 4.11c, f).

No clear differences in the concentration discharge relationships was observed between the fenced and unfenced tributaries for TP, or for its two fractions, during both the June and October events in 2011 (Table 4.6; Figure 4.11g, j, h, k, i, l). The concentration-discharge relationships for TP, particulate P and SRP concentrations all displayed anti-clockwise behaviours with the higher TP, particulate P and SRP concentrations occurring on the falling limb, as the stream discharge decreased.

Table 4.6 Phosphorus concentration-discharge relationship in the fenced and unfenced tributary during four high discharge events in the Grazing period of 2010

 and 2011.

		-	Concentration				Stream discharge		
Date	Parameter	Tributary		(mg	L-1)	-	(m ³ s	sec ⁻¹)	Direction hysteresis plot
			Start	Min	Max	Cum	Min	Max	
	ТР	Fenced	0.09	0.09	0.28	5.05			Figure-of-eight
		Unfenced	0.10	0.11	0.16	3.63	_	1.26	Clockwise
11th 12th July 2010	DD	Fenced	0.06	0.13	0.19	3.81	0.60		Clockwise
11 = 12 July 2010	11	Unfenced	0.09	0.09	0.13	2.54	0.00	1.50	Clockwise
	CDD	Fenced	0.03	0.04	0.07	1.24			Figure-of-eight
	SKP	Unfenced	0.02	0.04	0.07	1.15			Clockwise
	тр	Fenced	0.10	0.10	0.17	3.40			Clockwise
	Ir	Unfenced	0.11	0.11	0.18	4.09	6.20	12.75	Figure-of-eight
7 th – 8 th Sep 2010	РР	Fenced	0.08	0.07	0.11	1.93			Clockwise
		Unfenced	0.09	0.10	0.13	2.65			Clockwise
	SRP	Fenced	0.02	0.05	0.08	1.46			Figure-of-eight
		Unfenced	0.02	0.05	0.08	1.42			Anti-clockwise
	ТР	Fenced	0.03	0.03	0.16	2.01	0.31	0.46	Anti-clockwise
		Unfenced	0.10	0.04	0.11	1.72			Anti-clockwise
25th 26th Land 2011	РР	Fenced	0.02	0.05	0.12	1.28			Anti-clockwise
$25^{-1} - 26^{-1}$ June 2011		Unfenced	0.04	0.04	0.08	0.99			Anti-clockwise
	CDD	Fenced	0.01	0.03	0.06	0.76			Anti-clockwise
	SKP	Unfenced	0.04	0.03	0.05	0.74			Anti-clockwise
	тр	Fenced	0.02	0.02	0.18	4.03		5 6.25	Anti-clockwise
	Ir	Unfenced	0.07	0.07	0.31	5.82	2.75		Anti-clockwise
and 2rd 0 4 2011	DD	Fenced	0.04	0.12	0.16	3.22			Anti-clockwise
$2^{nu} - 3^{nu}$ Oct 2011	PP	Unfenced	0.05	0.15	0.23	3.93			Anti-clockwise
	CDD	Fenced	0.02	0.08	0.11	2.08			Anti-clockwise
	SRP	Unfenced	0.02	0.07	0.09	1.96			Anti-clockwise



Figure 4.10 Phosphorus concentration-discharge relationships at the lower sites on the fenced and unfenced tributaries in July 2010. Red circle indicates the start of the event and the arrows indicates the direction and the end of the event. C indicates clockwise direction and Fig 8 indicates figure-of-eight direction.



Figure 4.11 Phosphorus concentration-discharge relationships at the lower sites on the fenced and unfenced tributary during high discharge events in September 2010 and October 2011. Red circle indicates the start of the event and the arrows indicates the direction. C indicates clockwise direction, Anti-C indicates anti-clockwise direction and Fig 8 indicates figure-of-eight direction.

For TN, the fenced tributary concentration-discharge relationships displayed anti-clockwise behaviour throughout all four sampling events, indicative of slower nutrient delivery to the watercourse (Table 4.7; Figure 4.12). However, the concentration-discharge relationships for the unfenced tributary site were more variable. In mid Grazing season during the event in July 2010, both TN and TDN displayed a clockwise concentration-discharge relationship, showing a rapid washout of N at the start of the event. Total dissolved N represented 65% of TN for this event at this site. This pattern with an initial rapid increase in N concentrations was also observed for the unfenced tributary for the extreme high discharge event in September 2010, however, as the discharge weakened, higher TN and TDN concentrations were observed, giving a figure-of-eight pattern (Figure 4.12).

Similar to the P fractions, TN and TDN concentration-discharge behaviour for both sites in the June 2011 and October 2011 events showed anti-clockwise behaviour, indicating higher N washout as the stream discharge decreased (Table 4.6; Table 4.7; Figure 4.12e, g, f, h).

Fable 4.7 Nitrogen concentration	n-discharge relationship in the	fenced and unfenced tributary	y during four high discharge	e events in the Grazing peri	od of 2010 and
2011.					

Date	Parameter	Tributary	Concentration (mg L ¹)			Stream discharge (m ³ sec ⁻¹)		Direction hysteresis plot	
2		y	Start	Min	Max	Cum	Min	Max	
	TN	Fenced	0.86	0.86	1.72	33.90		1.36	Anti-clockwise
11th 12th Labor 2010	IN	Unfenced	0.95	0.75	1.49	30.89	0.60		Clockwise
$11^{\circ} - 12^{\circ}$ July 2010	TDM	Fenced	0.82	0.80	1.04	22.38	0.00		Anti-clockwise
	IDN	Unfenced	fenced 0.59 0.73 1.56 20.70			Clockwise			
	TN	Fenced	0.10	0.10	3.05	68.38	6.20	12.75	Anti-clockwise
7 th - 8 th Sep 2010	111	Unfenced	1.22	1.22	4.20	78.08			Figure-of-eight
	TDN	Fenced	0.10	1.12	1.33	30.18			Anti-clockwise
		Unfenced	1.02	1.54	1.97	41.64			Clockwise
	TN	Fenced	0.76	0.51	2.99	35.91	0.31 0.46	0.46	Anti-clockwise
25th 26th June 2011		Unfenced	0.94	0.26	2.26	25.33			Anti-clockwise
$25^{-1} = 20^{-1}$ June 2011	TDN	Fenced	0.42	0.56	0.98	15.04			Anti-clockwise
	IDN	Unfenced	0.33	0.51	0.81	13.64			Anti-clockwise
	TN	Fenced	1.01	1.01	1.71	32.84	2.75		Anti-clockwise
and ard Oat 2011	11	Unfenced	1.10	1.05	1.63	34.94		6.25	Anti-clockwise
$2^{-3} = 3^{-5}$ Oct 2011	TDM	Fenced	1.01	1.48	2.41	39.83			Anti-clockwise
	IDN	Unfenced	0.80	1.91	2.97	51.68			Anti-clockwise



Figure 4.12 Nitrogen concentration-discharge relationships at the lower sites on the fenced and unfenced tributary during high discharge events in September 2010 and October 2011. Red circle indicates the start of the event and the arrows indicates the direction. C indicates clockwise direction, Anti-C indicates anti-clockwise direction and Fig-8 indicates figure-of-eight direction.

4.6 Discussion

Catchment riparian areas are considered key zones in which to target mitigation measures aimed at reducing the loss of nutrients from agricultural land to surface waters (Stutter et al., 2012). However, there is limited knowledge available on the evaluation of these mitigation measures, particularly in the UK and Ireland, and especially on their effectiveness at a catchment scale (Kay et al., 2009; Bergfur et al., 2012; Conroy et al., 2016). Most studies on the topic have been undertaken in the USA (Galeone, 2000; Galeone, 2006; Vidon et al., 2008; Vidon et al., 2010), Canada (Miller et al., 2010; Miller et al., 2011) and Australia (McKergow et al., 2003; Bartley et al., 2010), where agricultural practice and management approaches will differ. The main difference between agricultural practices these study catchments and those in Ireland are smaller field sizes due to different topography and Ireland being significantly smaller in size, summer grazing and winter housing due to Ireland's wetter climate, and different slurry spreading practices on agricultural land. The objective of the current study was to investigate the impact of streamside fencing, which had been installed five years earlier, on P and N concentrations in two adjacent streams. Fencing, as a mitigation measure, excludes cattle from direct access to the water and allows riparian vegetation to re-grow (Linnane et al., 2010).

Overall, this study found significant differences in the downstream concentration compared to upstream concentration, between the fenced and unfenced reaches; differences in overall concentration-discharge relationships; and differences in concentration-discharge relationships for an extreme large discharge event (Table 4.8). In particular, the results indicated that net retention of both TP and TN was occurring during the Grazing season in the fenced stream reach, while net release of both nutrients was occurring in the unfenced reach (Table 4.8). However, this relative difference between the two reaches was reversed, for TP only, during the Open season (February to April), with net release of TP from the fenced reach, and net retention of TP in the unfenced reach, suggesting that differing processes were controlling export of the two nutrients and in both reaches. There were also differing concentrationdischarge relationships for TP for both reaches during the Grazing season, and for TN during the Open slurry spreading period. In addition, the sub-daily pattern during an extreme high discharge event also differed, with the fenced site showing a clockwise pattern for TP, and a more complex figure-of-eight pattern for TN. Despite the net TP release in the fenced stream reach, the estimated annual TP and TN loads were lower for the fenced stream, representing 74% and 83% of those from the unfenced tributary respectively.

	Fenced	Unfenced
Estimated TP load	0.57 kg TP ha ⁻¹ yr ⁻¹	0.77 kg TP ha ⁻¹ yr ⁻¹
Estimated TN load	5.99 kg TN ha ⁻¹ yr ⁻¹	7.24 kg TN ha ⁻¹ yr ⁻¹
Downstream change [TP] <i>Grazing season (Apr-Oct)</i>	Net retention	Net release
Downstream change [TP] Open season (Feb-Apr)	Net release Net retention	
Downstream change [TN] Grazing season (Apr-Oct)	Net retention	Net release
Downstream change [TN] Open season (Feb-Apr)	Net release	Net release
Overall concentration- discharge relationship TP <i>Grazing season (Apr-Oct)</i>	Lower at moderate discharges; high at low and higher discharges	General increase with discharge
Overall concentration- discharge relationship TN <i>Open season (Feb-Apr)</i>	Lower at high discharges	General increase with discharge
Sub-daily concentration- discharge relationship TP (Sep 2010)	Clockwise (source close to stream)	Figure-of-eight (lagged transit time)
Sub-daily concentration- discharge relationship TN (Sep 2010)	entration- ionship TNAnti-clockwiseFigure-of-eig (source distant from stream)(lagged transit from stream)	

Table 4.8 Summary of chapter findings

There are several mechanisms that could contribute to the overall lower nutrient concentration and the difference in the upstream-downstream patterns in the fenced stream during the Grazing season. These include a lack of direct defecation by cattle (Miller, 2010; Bond et al., 2012), an absence of disturbance of nutrient enriched sediments by cattle (Terry et al., 2014), and the absence of areas of exposed bankside soil as found at cattle drinking sites. Cattle were excluded from the fenced stream reach, while the unfenced stream had 14 active CAPs (Table 3.1), all of which also had large areas of bare and poached soil. Bond et al. (2012) found in their study that temperature was a key driver of when cattle entered the River Meon at Droxford in the UK. During warm dry periods, for example during the summer grazing months, cattle have a tendency to use waterways for thermoregulation (Legrand et al., 2011; Bond et al., 2012). Cattle intrusion into watercourses also results in soil poaching at the site. This can cause soil compaction, reduced soil infiltration rates and increased soil detachment rates (Conroy et al., 2016) (Figure 4.13). Furthermore, nutrient export rates from areas of bare soil, as seen close to CAPs in the unfenced reach in the Milltown Lake Catchment, tend to be higher particularly after periods of heavy rainfall (Figure 4.11) (Reichenberger et al., 2007). In addition to soil poaching, there is also a greater tendency for cattle to defecate and urinate while in the stream. Cameron et al. (2013) reported that, on average, a cow may urinate 10 to 12 times per day, which may be more frequent within waterways during warm/dry weather. Bond et al. (2012) also found in their study that cattle defecated more frequently while standing in a stream (3.2%) than in any other environment (0.6%). In addition to these cattle related pressures, higher rates of dissolved nutrient uptake by vegetation will occur during summer, while the capacity to trap particulate nutrients will also be higher in more densely vegetated buffer areas, along with a reduced discharge velocity (Dorioz et al., 2006; McConnell et al., 2013).



Figure 4.13 Cattle access points and areas of bare soil exposure along the unfenced tributary in the Milltown Lake Catchment.

All of these effects could explain the significant downstream increase in TP concentration (of which, on average, 55% was particulate P) during the Grazing season for the unfenced stream

reach, and the results of the GAM model for the overall TP concentrations in that tributary. This showed a distinct positive relationship between TP export and discharge, suggesting a washout of eroded material. In contrast, TP concentrations fell from upstream to downstream in the fenced reach and there was no clear direction in the TP concentration-discharge relationship. On this fenced tributary, cattle access and associated erosion on stream banks, was absent. However, a consistent difference between the fenced and unfenced sites was not evident in the sub-daily concentration-discharge relationships for the four individual events, highlighting the complexity of the processes involved in nutrient transfer and export.

In Ireland, grass silage harvesting for livestock winter food storage mainly occurs during the later Grazing period, with the first cut for silage generally taking place in early to mid-June (Teagasc, 2011). Slurry spreading, usually *via* the traditional broadcast (splash plate) technique, normally takes place immediately after a harvest or grazing event to minimise contamination of the grass sward (Carter et al., 2010; Mc Connell et al., 2013). High rainfall occurring after such activities will substantially increase the quantity of total P and N exported in overland discharge (Galeone, 2006; Reichenberger et al., 2007; Macintosh et al., 2011; Buckley, 2012). The vegetative buffer that had developed in the fenced reach of the Milltown Catchment over the five years since fencing, is likely to have acted as a filter to trap particulate P especially during the summer periods when vegetation cover would be higher (Figure 3.3) (Hoffmann et al., 2009). Furthermore, bioavailable P and N uptake by the vegetation within the buffer strip would have been at its highest during the growing season (Stoate et al., 2009; Roberts et al., 2012). In a study that investigated vegetative buffers similar in size to those in this study (1.5 m), Galeone (2000) found that nutrient export in overland runoff was reduced, to some extent, along streams adjacent to pastured land, with fencing consequently having a positive influence on the receiving surface water quality. Washout of slurry from adjacent fields is also likely to have contributed to export in the unfenced reach owing to high rainfall levels, particularly as unusually high levels were recorded in Ireland in late 2011 (Oireachtas, 2011).

Interestingly, although the total rainfall for the Closed period (322 mm) was the highest of all the study periods, there was no difference in upstream to downstream nutrient behavior between the two reaches, suggesting that similar processes were driving export at this time. During winter, rainfall events become more prevalent in northern climates, while vegetation growth dies back and becomes dormant, and consequently, dissolved nutrient uptake by plants is lowest (Roberts et al., 2012). Furthermore, soils become saturated due to high rainfall, decreasing the nutrient infiltration capacity, and thus leading to increased surface runoff. Under such conditions, the vegetation within the buffer zone would be unable to filter the high influx of particulates exported in overland discharge. Such a die-off in buffer vegetation would explain the change in the upstream-downstream TP concentrations that was observed for the fenced tributary. It is likely that particulate material trapped during the summer Grazing period would then have been washed out during higher winter discharges when vegetation cover was reduced. High nutrient export in the winter months from agricultural catchments was reported, for example, in studies by Tappin et al. (2013), Bond et al. (2012) and Rozemeijer et al. (2014). Rozemeijer et al. (2014) noted in their study that nutrients exported in winter played a crucial role in summer eutrophication in downstream receiving waters.

In the study catchment and in northwest Ireland, slurry spreading is prohibited until after the 31st of January (S.I. No. 31 of 2014). Consequently, farmers may enter the early spring with significant quantities of slurry in storage tanks that must be spread on the land (Dale et al., 2012; Cameron et al., 2013). Vegetation on streambanks at this time will have died back and will only just be starting to re-establish after the winter months (Roberts et al., 2012). In the current study, during the Open period, TP (which comprised c. 55% particulate P fraction) was

released from the fenced reach, a time also when slurry spreading was again permitted. Low vegetation cover in the vegetative buffer zone (the Open period) may have resulted in the washout of additional particulate P that had previously been retained within the buffer (Hoffmann et al., 2009). A similar effect was reported by Stoate et al. (2009) who also concluded that grassy buffer strips were ineffective in retaining P during early spring, particularly after heavy rainfall events, when vegetation was underdeveloped. The Open period in the current study, had the largest sum total N concentrations, which were dominated by the dissolved nitrogen fraction (c. 75% was TDN), for both the fenced and unfenced stream reach compared to the grazing period. The high net increase in total N during this period for both stream reaches may be due, in part, to the farmers spreading slurry at this time. While vegetation growth was low, particularly in the vegetative buffer strip on streambanks during the open period, vegetation would still not play a major role in filtering the total N concentration during this period as it was dominated by the dissolved N fraction. It is of note that the relationship to discharge was distinctly different between the two reaches. While the model for the unfenced tributary showed a clear positive relationship between discharge and total N concentration, the relationship between the same parameters for the fenced tributary was less so clear.

The majority of studies to date focusing on fencing as a management option to minimise nutrient export have been carried out in Canada and the US, and they mainly investigated effects on stream total P, rather than total N. These studies, however, would have been based in areas with different agricultural and land management practices to those in Ireland, making comparison difficult. In addition, studies also have adopted a range of differing approaches, which include: upstream/downstream, pre-/post-fencing and the fenced/unfenced sampling approach, for measuring the changes in nutrient concentrations in agricultural catchment surface waterbodies. In a study of the Little Bow River, Southern Alberta, Canada, which took a similar upstream-downstream approach to the current research, Miller et al. (2010) found that

TP concentrations were similar up and downstream in a fenced stream reach over a four-year study period. Line et al. (2003), in a study with pre-and post-fencing time periods in the North Carolina Piedmont, found an overall reduction of 76% in TP concentration and a 78% reduction in TN concentration from a 14.9 ha pasture post cattle exclusion fencing. Galeone (2006) found, for an eight year study in the Mill Creek watershed, Pennsylvania, which had a similar buffer size zone (1.5 - 3.7 m) to that in the current study, a 20% reduction in total N concentration post fencing of the riverbanks. Furthermore, similar to the finding of the current study, Galeone (2006) reported that the greatest reduction in TN concentration (47%) was during the Grazing season.

Previous work carried out in the Milltown Lake Catchment has shown that the catchment has been impacted by nutrient loading from agriculture since the 1970s, with resultant eutrophication clearly evident in the lake (Carson et al., 2015). High nutrient concentrations from the unsewered rural populations in this catchment are likely to have also had water quality impacts (McCarthy et al., 2012; Rafferty, 2015), but these were found to be much lower in magnitude compared to the agricultural sources (Carson et al., 2015). The annual TP and TN loads estimated in the current study can be considered moderate when compared to studies carried out in similar catchments elsewhere in Ireland. Jordan et al. (2012) reported nutrient loads of between $0.175 - 0.785 \text{ kg TP ha}^{-1} \text{ yr}^{-1}$ and $8.9 - 28.8 \text{ kg TN ha}^{-1} \text{ yr}^{-1}$ from two intensive grassland and two arable catchments in the ROI in a study based on high frequency automated sample collection. A further study by Jordan et al. (2005) reported annual TP loads for three different Irish grassland catchments (0.2 to 88.5 km^2) of between 0.2 to 3.1 kg P ha⁻¹ yr⁻¹. Furthermore, a study undertaken in a similar heavy clay drumlin catchment by Cassidy and Jordan (2011) reported annual TP loads of 1.8 kg TP ha⁻¹ yr⁻¹ up to 3.8 kg TP ha⁻¹ yr⁻¹.

Based on the paleolimnological evidence, such moderate levels of nutrient loading can maintain eutrophic conditions and cyanobacterial blooms in Milltown Lake (Carson et al., 2015), indicating that, as with many catchments in areas with intensive cattle based agriculture in Ireland, programme of measures (POMs) that include mitigation to reduce nutrient loads are required. The reduction in nutrient loading found in the current study, particularly during the Grazing period, indicated that fencing could contribute to such measures, particularly for lakes such as Milltown Lake that have a short retention time and where autumn and winter loading will likely be washed out of the lake the following summer. Phosphorus and N loading would, therefore, be reduced during the critical time for algal blooms, i.e. in summer. However, the increase in release of P from the fenced stream reach in the period February to April found in the current study could, potentially, increase nutrient availability for early season algae in the downstream lake.

4.7 Conclusion

This study highlighted both the effectiveness and limitations of streamside fencing, and of associated narrow buffer zones, as a management tool for nutrient transfer in an intensive agricultural landscape. It indicated that such measures can be effective in TP and TN management during the Grazing period. This is the time of year when nutrient uptake by plants is at its highest and the growth of vegetation in buffer areas will be greatest. However, the results also indicated that streamside fencing is less effective in nutrient management, especially for TP, during times when vegetation in the buffer zone is reduced. The study also confirmed that the processes driving nutrient concentrations differed between the fenced and unfenced stream reaches, especially for TP. This study provides valuable information to stakeholders, policy makers, and the local community and shows how streamside fencing can be considered an effective nutrient management measure. Particularly during critical periods in the calendar year where the effects of low flow conditions and excess nutrients resulting in eutrophication are amplified.

Chapter 5: Multiple drivers of phytoplankton biomass in Milltown Lake

5.1 Abstract

Understanding the drivers of change in phytoplankton standing stock in lakes is critical for the management of water resources. While nutrient availability from either the surrounding catchment, discussed in the previous chapter, or internal lake sources play an important role in controlling the magnitude of algal blooms in lakes, local climatic factors contribute either directly or indirectly through their influence on lake biota, particularly in small lakes. A study was undertaken of the relative importance of (1). climatic factors and (2). nutrient availability in controlling phytoplankton biomass in Milltown Lake, County Monaghan, at five depths between the surface and lake bottom. This chapter used the Lake Analyzer programme to calculate a set of metrics describing the physical structure of Milltown Lake using temperature sensor data, to get a better understanding of the lakes seasonal thermal mixing regime. The principal drivers of change in Chl-a were investigated using a General Additive Model approach. The sampling period (April to September 2012) was characterised by high rainfall and nutrient loading throughout the summer, with lake TP concentrations remaining above 0.042 mg TP L⁻¹. The analysis found that, under these high nutrient conditions, climate-related factors were the main drivers of change in Chl-a at all depths. However, different climatic forcing factors operated at the surface (wind speed and inflow discharge rate), compared to the lower sampling depths (thermocline depth and PAR). The only significant relationship with nutrient concentration was with in-lake NH₄-N at the lower depths. These results highlighted the sensitivity of phytoplankton biomass in small eutrophic lakes to current and future climatic change.

5.2 Introduction

Globally, the environment is changing at an unprecedented rate, with temperature and precipitation having altered dramatically already and predicted to change even more (Christensen et al., 2007; Meehl et al., 2007). These changes may bring about changes in other environmental conditions, such as nutrient loading and water column mixing, making it a challenge to assess the drivers of Chl-*a* in a lake's ecosystem (de Senerpont Domis et al., 2013). The climate change effects on lakes that have received most attention have been those related to warming, with the effects predicted to be substantial (Havens et al., 2016). An example of this is seen in the comparison made by Jackson et al. (2007) between lakes in Denmark, where winter air temperatures tend to stay above 0 °C, and lakes in Canada, where winter air temperatures fall below -10 °C and there is ice cover on the lakes. Jackson et al. (2007) found that the Danish lakes had a fourfold higher ratio of Chl-*a* to TP, and a stronger relationship between the biomass of zooplankton to phytoplankton than the Canadian lakes. Further studies by Kosten et al. (2012) showed that the comparison of phytoplankton community structure data from lakes along a latitude gradient from northern Europe to the tip of South America indicated that percentage dominance of cyanobacteria increased steeply with increasing temperature.

In addition to temperature changes, a change in the intensity and annual distribution of rainfall is also projected for some regions (Wentz et al., 2007; Nolan et al., 2010; Jennings et al., 2009). Such changes will likely result in increased nutrient loading, and changes in the pattern of loading to lakes from the surrounding catchment, especially if high rainfall events occur more frequently during summer. Nutrient enrichment of freshwater lakes in Ireland is predominantly attributed to incidental losses from agriculture in the surrounding watershed during periods of wetter weather (Jordan et al., 2005). The pattern in loading has also been projected to change in the future in response to changes in rainfall patterns (Jennings et al., 2009), with consequent changes in Chl-a in lakes (Dalton et al., 2016). Overall, global climate warming will have major implications for freshwater ecosystems and may cause a seasonal shift (Dalton et al., 2016), alter the incidence of harmful algal blooms and lead to

changes in ecosystem biodiversity (Sommer et al., 2012). Furthermore, Blenckner et al. (2007) predicted that, under future climate change scenarios, a change in phenology or increase in invasive species may occur.

The composition of phytoplankton communities in lakes, and therefore the total standing stock of phytoplankton, is controlled by site specific factors, including local meteorological conditions and nutrient availability (Cloern and Jessby, 2010). The link between the level of nutrient enrichment and phytoplankton and benthic biomass is well understood for lakes (Canfield and Hodgeson, 1983; Vollenweider, 1989). The reduction of nutrients, and of P concentrations in particular, is seen as a key requirement to achieve good ecological status within waterbodies (Bowes et al., 2012). Reynolds (2002), however, stated that external forcing or disturbances are also key drivers of changes in phytoplankton assemblages. Disturbance effects on changes in phytoplankton species succession and biomass may be directly correlated with environmental stressors, for instance phytoplankton bloom formation may be influenced by thermal structure and mixing regime within a lake (Paerl and Paul, 2012). Furthermore, Mortensen et al. (1994) identified shallow lakes, in particular, as being more easily influenced by fluctuations in the physical environment caused by climatic factors such as temperature, wind, and light. As suggested by Blenckner et al. (2007), it is these local weather conditions which govern many of the physical aspects of lakes ecosystems and therefore, either directly or indirectly influence lake biota. This chapter first reviews the known effects of climate and nutrients on phytoplankton standing stock and then presents results for a case study over one growing season in 2012.

5.3 Aims

Previous work carried out in the Milltown Lake Catchment used paleolimnological techniques to examine the historical drivers of nutrient enrichment and in-lake biological responses (Carson et al., 2015). Carson et al. (2015) found that the indicators of overall algal abundance were significantly related to both sedimentary P levels and P losses to water from the catchment, with agricultural intensification being the main contributor to increases in nutrient loading between 1970 and 2008 in Milltown Lake. Building on these findings, the work presented in this chapter aimed to investigate the relative importance of (1). local weather changes and their effects on lake physical characteristics, including stratification, and (2). nutrient (P and N) levels on the standing stock of phytoplankton biomass as measured by Chl-*a*. To put the data from the study year (2012) into context, climate and lake thermal regime data are also presented for the two preceding years: 2010 and 2011.

5.4 Material and methods

Meteorological data were obtained from the Ballyhaise and Coose Met Éireann weather stations, while high frequency stream discharge data were obtained from the EPA staff gauge on the Drumleek River (Figure 3.1; Figure 3.4). High resolution lake temperature and DO data used in this chapter were obtained from *in-situ* lake Tidbit sensors and D-Opto sensors, respectively (Section 3.5). Analytical methods for nutrients and Chl-*a* analysis used in this chapter for monitoring at the lake inflow, and at the deepest point in Milltown Lake, are described in Sections 3.4 and 3.6 (Figure 3.1).

5.4.1 Examining the inter-annual thermal structure using Lake Analyzer

The changes in lake physical structure were examined from the 16th of April 2010 until the 31st of December 2012 using high resolution lake temperature data (Section 3.5). Schmidt stability, buoyancy frequency, Wn and thermocline depth were determined from vertical temperature profiles and wind speed data using Lake Analyzer (Read et al., 2011) (Section 3.7). Thermocline depth calculations excluded the uppermost 0.5 m of the water column due to

diurnal surface water heating which at times created a strong but short-lived (sub-daily) density gradients (Simmonds et al., 2015).

5.4.2 Sampling programme assessing the importance of both climatic stressors and nutrient availability in controlling the standing stock of phytoplankton biomass

Lake water samples were collected at a range of depths on 16 occasions between the 1st of May 2012 and the 17th of September 2012 at the deepest point of the lake (~10 m) (Figure 3.1b). These were collected on a fortnightly basis during the spring (1st of May 2012 to 31st of May 2012) and autumn (4th of September 2012 to 17th of September 2012) and on a weekly basis during the remaining sampling period. It was decided to focus on weekly sampling during the summer months, as stratification for Milltown Lake had occurred during this period in the previous two years. Samples for nutrients analysis were collected at the sampling point using an Eijkelkamp® peristaltic pump with pre-measured 1 m interval tubing, at approximately 1 m depth intervals (0.1, 1, 2, 3, 4, 5, 6, 7, 8 and 9 m). In addition, samples for Chl-a analysis were collected at the surface and at approximately 2 m depth intervals (i.e. at 0.1, 2, 4, 6 and 8 m) at the same location. All water samples (nutrient and Chl-a) were collected using 2 L pre-acid washed polyethylene bottles, rinsed with ultrapure water. Sample bottles were rinsed with lake water from the depth to be sampled prior to sample collection. Methods used for the analysis of total nutrients, dissolved nutrients and Chl-a are described in Chapter 3. Additional lake measurements on each occasion included transparency using a Secchi disk, DO concentration and temperature, using a YSI 556 MPS multi probe (with a 30 m extension) to verify the in-situ sensor data. The portable YSI probe (YSI 556 MPS multi probe with a 30 m extension) was calibrated in the laboratory prior to carrying out field work and assessed periodically throughout the sampling trip. Depth at the sampling point on the day of sample collection was measured using a handheld Hondex (PS-7) Portable Depth Sounder.

An autosampler (HACH[®] Sigma 900 Max) was placed at the same location as the EPA staff gauge on the Drumleek River, 500 m upstream of the only lake inflow. This autosampler was set to collect composite daily water samples for TP and TN analysis for the same time period (1st of May 2012 to 17th of September 2012) (Figure 3.1a). The sampler was programmed to collect six 50 ml samples per day at 4 hour intervals, giving a final daily sample volume of 300 ml. The polyethylene sample bottles were pre-acid washed and rinsed with Milli-Q water. Sample bottles were collected on the same days as the lake samples. Stream discharge data from an EPA managed staff gauge provided continuous stream discharge data.

5.4.3 Data analysis

Discharge and meteorological data were obtained from the Ballyhaise and Coose Met Éireann weather stations and were available for all 137 days of the 2012 sampling period (1st May to the 17th of September 2012) (Figure 3.1; Figure 3.4). Photosynthetically Active Radiation (µE m⁻¹ sec⁻²) measurements were estimated from hourly shortwave radiation (SW) (Wm⁻²) from the Ballyhaise Met Éireann weather station by multiplying the SW radiation by 0.45 (Pinker et al., 2010). Daily nutrient loads were calculated for the catchment at the inflow site as the product of concentration and 137 estimated daily flows and expressed in g ha⁻¹ day⁻¹. Statistical analyses were carried out using the R statistical package version 3.2.2 (R 195 Core Team, 2015). General additive modelling was used to identify the main drivers of estimated Chl-*a* concentrations. Analysis was carried out using the mgcv package (Wood, 2006) in R (version 3.2.5, R 195 Core Team, 2015). The spreads of the residuals were relatively homogenous, indicating that a Gaussian distribution was appropriate for the model (Zuur et al., 2009). All data sets were assessed for any breaches of the assumptions of independence and heteroscedasticity using the methods in Zuur et al. (2009). All data sets conformed to these assumptions.

5.5 Results

5.5.1 Meteorological and hydrology data

The study catchment had relatively mild and wet conditions for the years 2010-2012, typical of weather conditions in the east of Ireland (Figure 5.1). Overall, the mean annual air temperature was 8.0 °C in 2010, 9.6 °C in 2011 and 9.2 °C in 2012. Total rainfall in these three years were 1017 mm year⁻¹ (2010), 1045 mm year⁻¹ (2011) and 1059 year⁻¹ (2012), respectively. The highest number of wet days (184 days) were recorded in 2012, with 2010 having 145 wet days and 2011, 177 wet days. Average daily rainfall was also highest (2.90 mm day⁻¹) in 2012 compared to the previous two years (2.78 mm day⁻¹ in 2010 and 2.86 mm day⁻¹ in 2011), coinciding with the highest average stream discharge of 2.90 m³ sec⁻¹ (2012). Average discharge recorded in 2010 and 2011 were 2.78 m³ sec⁻¹ and 2.86 m³ sec⁻¹, respectively.



Figure 5.1 2010 to 2012 data for (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station, plotted against daily stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (b) daily average wind speed and 30 day moving average wind speed (black line) for Milltown Lake, (c), daily average photosynthetically active radiation (PAR) and (d) daily mean air temperature (°C) at Ballyhaise Met Éireann weather station, between 1st of January 2010 and 31st of December 2012. The dotted black lines define each calendar year, while the area between the solid red lines define the sampling period in 2012.

The mean daily air temperatures recorded were similar in the period of the lake Chl-a study (1st of May to the 17th of September 2012) to the same time frame for the two preceding years. Overall the highest mean daily air temperature between the 1st of May to the 17th of September, at the Ballyhaise weather station was recorded in 2010 (13.5 °C), while mean daily air temperatures for the same period in 2011 and 2012 were both lower with 12.4 °C and 12.3 °C respectively (Table 5.1). Maximum air temperatures recorded for the equivalent study months in 2010 (on 21st of June 2010) and 2012 (on 26th of May 2012) were both 18.3 °C, while in 2011 the maximum air temperature reached 17.0 °C on the 03rd of June 2011 (Figure 5.1d; Figure 5.2d). The mean daily wind speed from the 1st of May to the 17th of September in 2010 was 5.2 m sec⁻¹, with the highest daily wind speed (9.7 m sec⁻¹) occurring on the 7th of July 2010 (Table 5.1; Figure 5.1b). Mean daily wind speed during the same period in 2011 was 6.6 m sec⁻¹, with the highest daily wind speed (13.1 m sec⁻¹) occurring on the 23rd of May 2011 (Table 5.1; Figure 5.1b). In 2012, the mean daily wind speed was lower at 5.5 m sec⁻¹, while the highest daily wind speed of 10.8 m sec⁻¹ was recorded on the 14th of May (Table 5.1; Figure 5.1b; Figure 5.2b). Surface PAR, as might be expected, showed a similar trend as air temperature, increasing in the spring and summer months and decreasing in the autumn and winter months (Figure 5.1c, d). During the study period equivalent months in 2010, the maximum daily mean PAR was 717 µE $m^{-1} \sec^{-2}$ on the 21st of June 2010, while the minimum of 76 µE $m^{-1} \sec^{-2}$ was recorded on the 9th of June 2010 (Table 5.1; Figure 5.1c). Maximum daily mean PAR for 2011 and 2012 were 704 µE m⁻¹ sec⁻² (3rd of June 2011) and 653 µE m⁻¹ sec⁻² (26th of May 2012), while the minimum PAR recorded for the same years were 125 µE m⁻¹ sec⁻² (16th of May 2011) and 95 µE m⁻¹ sec⁻ ² (8th of June 2012) respectively (Table 5.1; Figure 5.1c; Figure 5.2c).

Notably, total rainfall at Coose weather station for the study period was almost double that for 2010 and 2011 (Table 5.1). In those years, it was 264 and 226 mm respectively, while in 2012 it was 479 mm. The mean daily rainfall during the same period ranged was 2.9 and 2.5 mm day⁻

¹ in 2010 and 2011, and 3.5 mm day⁻¹ in 2012 (Table 5.1). Maximum rainfall of 27.5 mm day⁻¹ was recorded on the 8th of June 2012 (Table 5.1; Figure 5.1a; Figure 5.2a). The wettest year (days \geq 1 mm of rain) was again 2012 with 174 wet days, while 2010 was the driest (137 wet days) (Figure 5.1a; Figure 5.2a). Daily stream discharge at the EPA staff gauge on the Drumleek River over the three years (2010, 2011 and 2012) reflected these differences in rainfall. The maximum daily value was 12.1 m³ sec⁻¹ on the 6th of September 2010 (Figure 5.1a). The mean daily stream discharge was 0.45 m³ sec⁻¹ in 2012 (Table 5.1; Figure 5.2a).

Table 5.1 Comparative summary table of meteorological data (Ballyhaise and Coose Met Éireann weather station) and hydrological data (EPA staff gauge) during the study period months (1st May to the 17th of September 2012) for the same period in the previous years 2010 and 2011.

1 st May to the 17 th of September	2010	2011	2012
Mean daily air temp (Ballyhaise) (°C)	13.5	12.4	12.3
Mean daily wind speed (Ballyhaise) (m sec ⁻¹)	5.2	6.6	5.5
Max PAR (µE m ⁻¹ sec ⁻²)	717	704	653
Min PAR (µE m ⁻¹ sec ⁻²)	76	125	95
Mean daily rainfall (Coose) (mm day ⁻¹)	2.9	2.5	3.5
Cumulative rainfall (Coose) (mm y-1)	264	226	479
Max daily rainfall (Coose) (mm day ⁻¹)	24.6	15.9	27.5
Mean daily stream discharge (m ³ sec ⁻¹)	0.08	0.13	0.45



Figure 5.2 2012 data for (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station, plotted against stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, (b) daily average wind speed and 30 day moving average wind speed (black line) for Milltown Lake, (c) daily average photosynthetically active radiation (PAR) from the Ballyhaise Met Éireann weather station and (d) daily average surface water temperature for Milltown Lake and air temperature from the Ballyhaise Met Éireann weather station, January to December 2012. The white area between the dashed lines represents the study period (1st May to the 17th of September 2012).

5.5.2 Annual lake temperature and stratification profile (2010, 2011 and 2012)

While Milltown Lake generally displayed monomictic behaviour, there were also periods of mixing and re-stratification during summer in each of the years 2010, 2011 and 2012, and differing pattern in other metrics of lake physics in each year (Figure 5.3a). Higher wind speeds often resulted in the lake becoming mixed. In 2010, Milltown Lake stratified on the 28th of April 2010 and generally remained stratified, with high stability, until particularly high wind speeds (maximum wind speed reaching 11.5 m³ sec⁻¹) on the 30th of October 2010 when the lake became fully mixed (Figure 5.1b; Figure 5.3a). In 2011, in contrast, the lake initially stratified on the 17th of March. Throughout the spring and summer, the lake went through numerous short stratification and mixing processes. There was a notable period of intense stratification and high stability in 2012, starting on the 21st of May which lasted until the 08th of June (Figure 5.3a; Figure 5.4c). It then continued to stratify for short periods until it became fully mixed on the 15th of September 2012.

In 2010, S and buoyancy frequency increased from 07th of April 2010 and remained high over the stratification period (Figure 5.3a, b, c). Values of S ranged from 8.6 J m⁻² to a maximum of 83.8 J m⁻² and N ranged from 0.0007 N s⁻¹ to 0.0068 N s⁻¹ (07th April to the 02nd of July 2010). Milltown Lake remained stable and fully stratified until a breakdown in stratification occurred after the 30th of October 2010 and the lake became fully mixed once more. Overall, the strongest vertical stratification was seen in 2010 when the Wn reached 52.1 Wn, on the 21st of December (wind speed reached 4.6 m³ sec⁻¹) when the lake experienced a brief period of isothermal stratification (Figure 5.3d). Average temperatures above the thermocline were 1.3 °C and below the thermocline 3.1 °C during this period. In 2011, however, S and N did not have the same consistent pattern as in 2010. Both initially increased from the 22nd of February until the 30th of April 2011 to 33.6 J m⁻² and 0.003 N s⁻¹ respectively, when there was sudden decrease from the 30th of April until the 26th of May 2011. There was a build-up in lake stability from the 30th of May until the 05th of August 2011 with a maximum stability of 40.2 J m⁻², Wn on the same day was 29.3 Wn. Breakdown in stability occurred on the 16th of October 2011. Both S and N increased again on the 21st of May 2012 and peaked at 50.3 J m⁻² and 0.006 N s⁻¹ respectively on the 30th of May 2012 (Figure 5.3b, c).



Figure 5.3 16th of April 2010 and the 31st of December 2012 (a) contour plot of lake water temperature (°C), (b) Schmidt stability (J m⁻²), (c) buoyancy frequency (N s⁻¹) and (d) Wedderburn number (Wn). The dotted grey lines define each calendar year.

5.5.3 Lake abiotic data (01st of May to the 17th of September 2012)

Mean daily TP loads from the Milltown Lake Catchment at the inflow to Milltown Lake in the study period (1st of May to the 17th of September 2012) were relatively high for summer, at 1.6 g P ha⁻¹ d⁻¹, while the mean total N export was 22.0 g N ha⁻¹ d⁻¹. The maximum total P (28.3 g P ha⁻¹ d⁻¹) and total N (277.9 g N ha⁻¹ d⁻¹) loads were recorded on the 8th of June 2012, coinciding with the maximum rainfall and stream discharge (Figure 5.4a). This peak was followed by consistent pulses in P and N loading for the rest of the summer, reflecting the peaks in river discharge.

During the more intense stratification period from 21^{st} of May until 8th of June (19 days), the surface mixed layer depth ranged between 1 m and then progressively increased to a maximum depth of 3.6 m on the 8th of June 2012. The mean Secchi depth reading during this period was 1.5 m. The surface mixed layer temperatures ranged from 12.1 °C (21^{st} of May) to 19.7 °C (30^{th} of May). Temperatures below the thermocline had a lower range, from 9.7 °C to 10.3 °C. The hypolimnetic water started to become anoxic during this period of stratification (Figure 5.4b). The average DO above the thermocline was 8.91 mg L⁻¹ while that below the thermocline was 5.75 mg L⁻¹ (Figure 5.4b). The mean S was 36.1 J m⁻² and included the maximum S level for the overall sampling period (50.4 J m⁻²) on the 30th of May 2012 (Figure 5.4d). However, the lake then mixed to 7.7 m immediately after the high rainfall on 8th of June and a concurrent decrease in air temperature from 13.9 °C to 11.0 °C, and remained fully mixed until 16th of July 2012. Secchi depth on the 8th of June measured 0.97 m and maintained an average depth of 0.95 m until the lake began to stratify again on the 16th of July 2012.

Between, the 16th of July and the 25th of August 2012, stratification was again established but was weaker than in late May/early June and followed a deepening trend as the summer progressed. This time period included several shorter periods where the mixed surface layer depth was shallow (ranging between 1.0 m to 2.3 m) which only lasted between 3 to 7 days. A
period with slightly more intense stratification occurred between the 11th of August and the 25th of August 2012 (15 days), with the thermocline depth ranging between 2.0 m (11th of August) and 6.1 m (18th of August). The mean Secchi depth between this stratified period measured 0.9 m on the 12th August 2012. During this stratified period, the mean water temperature for the surface mixed layer was 15.6 °C, while the mean water temperature beneath the thermocline depth was 13.0 °C. The mean DO during this period (11th of August to the 18th of August 2012), above and below the thermocline were 6.81 mg L⁻¹ and 0.99 mg L⁻¹ respectively (Figure 5.4b). The lake became isothermal again on the 15th of September 2012 (Figure 5.4c).



Figure 5.4 1st of May to the 1st of October 2012, (a) stream discharge $(m^3 \sec^{-1})$ and nutrient loadings $(g ha^{-1} d^{-1})$ from the lake inlet site, (b) dissolved oxygen $(mg L^{-1})$ at 1 m and 7 m depth (c) contour plot of lake temperature (°C) with thermocline depth (m) (superimposed black line), Secchi depth measurement on day of sampling (m) (superimposed white dots) and (d) Schmidt stability (J m⁻²). Note difference in scale for total N and total P loads.

The maximum Chl-*a* levels (44.5 μ g L⁻¹ at the surface) were actually recorded at the start of the study period on the 17th of May 2012, with high concentrations (mean 31.2 μ g L⁻¹, n = 5) occurring throughout the water column on this date (Figure 5.5a). Immediately following the onset of the period of intense stratification in late May, concentrations at the surface actually reduced to 9.0 μ g L⁻¹ (on the 24th of May). However, it was also notable that during the initial days of intense stratification, Chl-*a* levels were higher below the shallow 1.0 m thermocline depth, with concentrations of 32.8 μ g L⁻¹ and 9.5 μ g L⁻¹ measured at the 2 m sampling depth on 24th of May and 29th of May, respectively. In contrast, surface Chl-*a* concentrations on these days were 9.0 μ g L⁻¹ and 4.7 μ g L⁻¹. This pattern changed toward the end of that stratification period as the thermocline depth increased to a maximum of 3.6 m on the 8th of June 2012, when a higher Chl-*a* concentration was again recorded in the surface (0.1 m) layer (mean Chl-*a* 17.3 μ g L⁻¹) than in the deeper layers (Chl-*a* range: 2.7 μ g L⁻¹ at 8 m to 7.5 μ g L⁻¹ at 4 m). A further peak in Chl-*a* concentrations occurred around the 31st of July 2012 (Figure 5.5a). However, again, the maximum levels of 27.2 μ g L⁻¹ were detected at the 4 m depth, which was below the thermocline depth of 2.4 m.

Phosphorus concentrations in the lake itself ranged between 0.042 and 0.149 mg P L⁻¹ for TP and 0.017 and 0.084 mg P L⁻¹ for SRP. High concentrations of TP (mean 0.088 mg P L⁻¹) were recorded in the upper layers (thermocline depth ranging between 1.0 m to 5.8 m) in the two weeks prior to stratification on the 21st of May 2012 (Figure 5.5b). However, concentrations gradually decreased throughout the water column once stratification had started, with a minimum of 0.043 mg P L⁻¹ on the 4th of June (Figure 5.5b). Total P concentrations increased after this stratification period at all depths (8th of June) and remained consistently high (> 0.556 mg P L⁻¹) until the 4th of September (Figure 5.5b). Throughout late July and early August, TP concentrations were generally found to be higher above the thermocline depth (2.8 m) than below it. However, high TP concentrations were observed below the thermocline on both the 16^{th} of July (0.142 mg P L⁻¹) and 31^{st} of July (0.148 mg P L⁻¹). During the period with most intense stratification (21^{st} of May to the 8^{th} of June 2012), high concentrations of SRP were observed below 6 m, ranging between 0.028 mg P L⁻¹ and 0.043 mg P L⁻¹ (Figure 5.5c). During July, increased SRP concentrations were found throughout the water column on two separate occasions, a mean of 0.053 mg P L⁻¹ on the 4^{th} of July and a mean of 0.059 mg P L⁻¹ on the 24^{th} of July (Figure 5.5c). Once the lake became isothermal on the 15^{th} of September 2012, the mean TP and SRP concentrations recorded were 0.042 mg TP L⁻¹ and 0.024 mg SRP L⁻¹, respectively.



Figure 5.5 Contour plots of (a) chlorophyll a (μ g L⁻¹), (b) TP (mg L⁻¹) and (c) soluble reactive phosphorus (SRP) (mg L⁻¹), with thermocline depth superimposed (m), at the deepest point in Milltown Lake, between the 1st of May 2012 and the 17th of September 2012.

During the months of May and June, total N concentrations were always less than 2 mg N L⁻¹, but increased throughout the water column from July onwards, with maximum concentrations of 4.6 mg N L⁻¹ (≤ 1 m) on the 24th of July 2012 (Figure 5.6a). For the two weeks prior to stratification on the 21st of May, NO₃-N concentrations were high throughout the water column (between 0.65 mg N L⁻¹ and 0.88 mg N L⁻¹). During the stratified period (21st of May to the 08th of June 2012), NO₃-N concentrations decreased to a water column mean of 0.44 mg NO₃-N L⁻¹. Concentrations of NO₃-N increased on the 11th of June (mean 0.79 mg N L⁻¹) but decreased by the 19th of June (mean 0.44 mg -N L⁻¹) (Figure 5.6c). On the 4th of July 2012 at the 3 m depth interval, concentrations of NO₃-N spiked to 0.92 mg N L⁻¹. While concentrations remained high until the 7th of August, NO₃-N levels reached further highs with an average of 0.78 mg N L⁻¹, throughout the water column until the end of the sampling period (17th of September 2012) (Figure 5.6c).

Ammonium concentrations where highest between the 4th of July and 24th of July 2012, with levels below the thermocline (mean 3 m), ranging between 0.12 mg N L⁻¹ to 0.52 mg N L⁻¹ (Figure 5.6b). As concentrations of NH₄-N in the hypolimnion increased during this stratified period (4th of July to the 24th of July 2012), DO concentrations decreased, for instance the highest level of NH₄-N concentration (0.52 mg N L⁻¹) occurred at 9 m on the 24th of July which coincided with the lowest DO concentration (2.52 mg L⁻¹) recorded at the 7 m DO sensor. From the 1st of August to the 17th of September 2012, NH₄-N levels were homogenous throughout the water column, with a mean value of 0.14 mg N L⁻¹ (Figure 5.6b).



Figure 5.6 Contour plots of (a) TN (mg L^{-1}), (b) ammonium (NH₄-N) (mg L^{-1}) and (c) nitrate (NO₃-N) (mg L^{-1}), with thermocline depth at the deepest point in Milltown Lake, superimposed (m), between the 1st of May 2012 and the 17th of September 2012.

5.5.4 Drivers of chlorophyll a concentration (1st of May to the 17th of September 2012)

For all depths, physical and meteorological variables were the main drivers of change in Chl-*a* concentrations, rather than in-lake nutrient concentrations or inflow nutrient loads (Table 5.2). Two disturbance factors, wind speed (p = 0.0067) and stream discharge (p = 0.0100), explained 53% of the deviance in Chl-*a* concentrations at the lake surface (0.1 m) (Table 5.2). The relationships were generally positive and linear (estimated degrees of freedom (edf)) of 1.5 and

1.1 for wind speed and stream discharge, respectively (Figure 5.7a, b). At the 2 m depth interval, S was the dominant driver (p = 0.0008), with wind speed still significant but having a weaker relationship (p = 0.0358). This model explained a higher portion of the deviance, however, with a R^2 adjusted of 0.74 (Table 5.2). The relationship to stability was curvilinear with Chl-a biomass peaking at a stability of 20 J m⁻² to 30 J m⁻² and decreasing at higher levels (Figure 5.7c). At all of the remaining three depth intervals, and in contrast to the two shallower sampling levels, mean thermocline depth and PAR were significant terms in the optimum model. In addition, it is of note that the only nutrient term that was significant in any of the models, NH₄-N, explained a portion of the deviance at two of the lower sampling depths (4 m and 6 m). The optimum models explained 90%, 84%, and 73% of the deviance at 4 m, 6 m, and 8 m, respectively (Table 5.2). The relationship between biomass and mean thermocline depth was generally negative, that is Chl-a concentrations were higher as the thermocline depth became shallower. This term was most significant at 4 m but less significant as depth increased (Table 5.2). In contrast the relationship between Chl-*a* concentrations and estimated PAR levels was generally positive (Figure 5.7f, i, l), with the exception of the 4 m depth where higher Chl-a levels coincided with both low and high PAR levels. At both the 6 m and 8 m depths, PAR was a more significant term in the optimum model than mean thermocline depth (p = <0.0001, p =0.0003 respectively) (Table 5.2). Lake NH₄-N concentrations were significant at both 4 m and 6 m only (Table 5.2) with higher Chl-a levels coinciding with both high and low NH₄-N concentrations at the 4 m depth (Figure 5.7g), and a weak but negative relationship indicated at 6 m (Figure 5.7j).

Level	n	Parameter	Edf	р	R ² adjusted
Surface	14	Wind speed	1.50	0.0067	0.53
		Stream discharge	1.10	0.0100	
2 m	14	S	1.97	0.0008	0.74
		Wind speed	1.91	0.0358	
4 m	14	Mean thermocline depth	1.96	0.0023	0.90
		PAR	1.99	0.0005	
		Lake NH ₄	1.98	0.0008	
6 m	14	Mean thermocline depth	1.00	0.0146	0.84
		PAR	1.96	< 0.0001	
		Lake NH ₄	1.00	0.0351	
8 m	14	Mean thermocline depth	1.00	0.0465	0.73
		PAR	1.93	0.0003	

Table 5.2 GAM model for the drivers of Chl-*a* concentrations (μ g L⁻¹) at the surface, 2 m, 4 m, 6 m and 8 m depth intervals at the deepest point in Milltown Lake.



Figure 5.7 Estimated smoothing curve for chlorophyll *a* concentrations (μ g L⁻¹) at the surface (**a**) stream discharge, (**b**) WS (wind speed); 2 m (**c**) S (Schmidt stability), (**d**) WS (wind speed); 4 m (**e**) mean thermocline depth, (**f**) PAR (photosynthetically active radiation), (**g**) Lake NH₄; 6 m (**h**) mean thermocline depth, (**i**), PAR (photosynthetically active radiation) (**j**) Lake NH₄ and at the 8 m (**k**) mean thermocline depth, (**l**) PAR (photosynthetically active radiation) intervals at the deepest point in Milltown Lake. Grey shaded area indicates the 95% confidence bands.

5.6 Discussion

Local weather conditions govern many of the physical aspects of lake ecosystems, which are important drivers of change in phytoplankton assemblage (Reynolds, 2002; Blenchner et al., 2007). The high resolution temperature measurements from Milltown Lake indicated that the lake, at times, fluctuated between mixing and stratifying during the summer seasons. This thermal regime is indicative of the thermal behaviour of a small polymictic lake (Hutchinson, 1957; Woolway et al., 2017). The polymictic nature of Milltown Lake was previously noted by Carson et al. (2015). However, that study was based on weekly temperature profiles taken over the summer months in 2007-2008. Ireland has an oceanic climate that is relatively mild, with high levels of precipitation and highly variable wind speeds (Keane, 1986). The comparison of climatic and in-lake thermal conditions for Milltown Lake indicated that, in 2010, the air temperature range was greater during the study period months (1st May to the 17th of Sept 2010), than in the following two years. The wind speed range and the stream discharge within the catchment was also lower in 2010. It is important to note that both 2010 and 2011 had a greater number of drier low flow periods than the study year, 2012. In response to these conditions, the lake stratified for longer periods in 2010 and there was a stronger S, buoyancy frequency and Wn and a more consistent and stable thermocline depth. In contrast the opposite weather conditions were seen in 2011 and 2012, resulting in weaker periods of stratification and thermocline stability. The main factors that contributed to this in 2011 and 2012 were wind speed and air temperature, while in 2012 additional pressures, including periods of higher rainfall and, as a result, greater daily stream discharge also played a significant role. The transient stratifications that were seen in Milltown Lake are commonly seen in polymictic lakes, for example, Lough Neagh (Northern Ireland), Bassenthwaite Lake (UK) or Rotorua (New Zealand), where wind and convective mixing are the most common driving factors (Woolway et al., 2014).

In addition to climatic effects on the lake thermal structure caused by changes in wind speed and rainfall, high rainfall also results in day to day changes in nutrient loading to lakes, particularly in agricultural catchments (Jordan et al., 2005; Macintosh et al., 2011). Analysis of nutrient losses from 17 catchments covering different climatic conditions across Europe, carried out by Bouraoui et al. (2009), found that climatic variables, in particular total rainfall, explained most of the variation in the nutrient load at the catchment's outlet (Nõges et al., 2011). This effect of high rainfall on loading was particularly evident for Milltown Lake in 2012. Nutrient availability and grazing pressures are generally considered to be the main controls on phytoplankton biomass in lakes (Teissier et al., 2011). Generally, temperate lakes experience nutrient limitation due to reduced inflow and higher uptake rates at some point during the growing season, especially when the lake is stratified (Wetzel, 2001), although, seasonal anoxic conditions can result in nutrient release from the sediments to the hypolimnion (O'Dwyer et al., 2013).

It was particularly notable that nutrient levels in the study lake remained high throughout the wet summer of 2012. Such conditions are not uncommon in the oceanic temperate climate of western Europe and, indeed, may become more common in more regions in response to directional climate change (Paerl et al., 2016). The availability of daily nutrient loading data, together with high frequency temperature and nutrient data along a depth profile provided a unique opportunity to assess the relative importance of both physical conditions and nutrients as drivers of phytoplankton biomass in the surface water and at depth. Studies of the drivers of phytoplankton biomass over a lake's entire depth profile are rare (Brentrup et al., 2016).

The GAM analysis indicated that both nutrient and climatic drivers affected Milltown Lake's phytoplankton differently along the lake depth profile. The percentage deviance explained at all depths was high, ranging from 53% to 90%, giving confidence in the model output.

However, it is of note that the main driving variables that were identified differed for the surface and deeper layers, and that climate-related factors dominated at all depths. In the surface waters of Milltown Lake, mean daily wind speed and stream discharge showed significant positive relationships with Chl-a levels suggesting that disturbance and surface mixing were the factors controlling phytoplankton standing stock, as measured by Chl-a. Moreno-Ostos et al. (2009) reported that chlorophyll fluorescence in the upper and surface layers of a small lake responded rapidly to wind stresses. They suggested that wind-induced mixing enabled internal nutrient mixing, facilitated phytoplankton species to remain in suspension, allowed species replacement and transported species from deeper waters to the upper water column. Wu et al. (2010) measuring Chl-a as a proxy for phytoplankton standing stock, reported that wind conditions affected phytoplankton distribution through advective water movements over a short time frame in Lake Taihu, China. The consistent high levels of discharge to Milltown Lake after the beginning of June 2012 would also have contributed to mixing in the surface waters. It was notable that neither TP nor TN loading to the lake were significant terms in the optimum model for the surface layer, but that inflow discharge was. This confirmed that the effect was related to the physical effects of inflowing water rather than to any nutrient effect. The strong influence of stream discharge as a physical driver in Milltown Lake was particularly evident on the 8th of June 2012, when the initial early summer period of strong stratification was broken down by a high stream discharge event (4.25 m³ sec⁻¹). Wind speed at this time did not increase and was relatively low at 3.0 m³ sec⁻¹. Along with this de-stratification, a high pulse of both TP and TN was carried into the lake from the catchment during this event.

The 2 m sampling depth was beneath the shallow thermocline in May, but at or above the mean thermocline depth in July and August. Schmidt stability is the stability of a waterbody and its resistance to mixing, therefore the amount of work required to transform a waterbody to a uniform water density state (Wetzel, 2001). Schmidt stability was a significant term in the GAM

model at the 2m depth in Milltown Lake, with highest Chl-*a* concentrations occurring at intermediate stability levels of around 30 J m⁻². This indicated that, at this depth, as the lake became more stable and less prone to wind mixing stresses, phytoplankton biomass increased. This may have resulted the formation of a cyanobacteria bloom in the epilimnion of Milltown Lake, as lake stability due to decreased wind stresses is important factor in the formation of cyanobacteria blooms in the upper epilimnion (Ibelings et al., 1994; Huisman et al., 2004; Monero-Osten et al., 2009).

Both physical factors (mean thermocline depth and PAR) and one nutrient factor (NH₄ concentration within the lake) were drivers of phytoplankton biomass in the deeper waters, below 4 m. Solar radiation levels, on which PAR was based, were obtained from a meteorological weather station 52 km south-west of Milltown Lake. However, the existence of this relationship suggests that they adequately captured conditions at the lake. The mean photic zone depth (Luhtala and Tolvanen, 2013) (2.5 times mean Secchi depth) of 3.6 m in this eutrophic lake was greater than the mean thermocline depth (2.8 m) on 7 of the 15 sampling occasions between 1st of May 2012 and the 1st of October 2012. Simmonds et al. (2015) found, for a highly eutrophic lake in New Zealand (Lake Ōkaro), that the photic zone was greater than the thermocline depth and the Chl-*a* maximum for the duration of their study. Simmonds et al. (2015) described this pattern as being typical for the Chl-*a* maximum formation in clear water lakes where the photic zone always exceeds the depth of the lower metalimnion. However, this pattern is not commonly seen in many eutrophic lakes systems.

The depth of the photic zone in Milltown Lake, below the thermocline depth, may explain why PAR was a significant driver of phytoplankton biomass in the hypolimnion. Chlorophyll *a* can become concentrated at discrete depths where there may be contrasting gradients in light and nutrients that limit light availability (Simmonds et al., 2015). It is likely therefore, that

phytoplankton in the hypolimnion of Milltown Lake could survive below the thermocline depth due to receiving adequate PAR levels. Some species can also migrate to ensure optimum PAR levels. Diatoms, for example, can have reduced settling rates at the thermocline (Coon et al., 1987), and cyanobacteria can regulate buoyancy (Padisa'k et al., 2004; Carraro et al., 2012), while ciliates and dinoflagellates are motile (Modenutti et al., 2004; Simmonds et al., 2015). Even though no underwater PAR measurements were available, the significance of PAR in the GAM models, and indeed the fact that the maximum Chl-*a* was recorded at lower depths in the lake on 7 of the 15 sampling occasions, indicate that PAR was reaching the lower waters below the thermocline. The relationship of Chl-*a* to thermocline depth was negative, that is higher phytoplankton biomass occurred when the overlying epilimnion was shallow. Since phytoplankton only have a short generation time, they show a quick response to changes in their environmental conditions (Lebret et al., 2012). Therefore, a shallow thermocline would allow phytoplankton in the lower hypolimnion of Milltown Lake to be circulated toward upper layers and increased light exposure.

The export of nutrients from catchments to lakes is strongly influenced by land use practices (Vanni et al., 2010). Milltown Lake is situated in an agricultural catchment with high rainfall and a flashy hydrology (Carson et al., 2015). The estimated annual loads from the Milltown Catchment, measured in the present study at the lake inlet were 0.58 kg TP ha⁻¹ yr⁻¹ and 8.03 kg TN ha⁻¹ yr⁻¹, respectively. In an Irish context, loads to Milltown Lake from the catchment were at the mid to high range of the scale compared to losses presented by Jordan et al. (2012) from two intensive grassland catchments (0.17 - 0.78 kg TP ha⁻¹ yr⁻¹ and 8.9 - 28.8 kg TN ha⁻¹ yr⁻¹), and Jordan et al. (2005) from three grassland catchments (0.17 to 3.13 kg TP ha⁻¹ yr⁻¹). These loads would have included those from the fenced western tributary, where mitigation measures have already been implemented and have been shown to potentially decrease export during summer (Chapter 4). Catchment farming activities, such as organic and inorganic

fertiliser spreading during spring and summer months, would have contributed to the observed high nutrient loading to the lake, maintaining the lake in a high nutrient state.

As reported by Carson et al. (2015) for Milltown Lake and similar to the last two decades, the lake was eutrophic in summer 2012 based on the O.E.C.D classification scheme (Taylor et al., 2002). However, although lake TP and TN concentrations were consistently high throughout the water column during the current study, NH4-N levels were found to be a significant driver of Chl-*a* levels in the lower waters. Ammonium is preferred by phytoplankton due to it being readily assimilated into cells compared to NO₃-N which must be transformed by nitrate reductase before it can be metabolised (Reynolds, 1997; Burger et al., 2008; Simmonds et al., 2015). The observed relationship to NH₄-N suggests that the phytoplankton species within the hypolimnion were N limited. Additionally, this study has shown that internal loading can be involved in the control of phytoplankton biomass in a shallow eutrophic lake which is already under stress from external nutrient loading.

Management of excessive phytoplankton production is usually achieved through a reduction in P inputs from the catchment, as P is often regarded as the limiting nutrient (Carpenter et al., 1998; May et al., 2009). Moss et al. (2013) suggested that this is often the case due to a long held presumption that restricting N load in freshwaters is pointless, due to N control being more complex and expensive than P control particularly from diffuse sources (Directive 91/676/EEC). Nonetheless, our results clearly showed that N can be a limiting factor for phytoplankton biomass, as NH₄ concentrations accumulated from decomposing organic matter and sediment release can stimulate phytoplankton growth, exhibit toxicity to aquatic biota, and exert an oxygen demand on the surface waters (Directive 91/676/EEC; Wetzel, 2001; Beutel, 2006). Therefore, management strategies for this catchment need to consider for both P and N control. On-going and projected climate change can have a profound effect on nutrient loading

in catchments that have a flashy hydrology by delivering high nutrient loading, particularly before the onset of stratification (Crockford et al., 2015; Jeppesen et al., 2010, 2011; Moss et al., 2011). As Milltown Lake is an agricultural catchment that is prone to high rainfall, nutrient inputs from this source are a primary concern particularly when considering lake restoration approaches. Results show that, not only is Milltown Lake impacted by surface nutrient loading, but sediment nutrient release, particularly NH4, may also be a contributing factor. Management strategies are mainly devised to target nutrient loads from the surrounding catchment. However, more attention is needed in understanding the internal lake sediment release, as it is considered to play a significant role in the years after catchment loading have been reduced (Jeppesen et al., 2010). Management strategies in more recent years are looking at using geo-engineering materials (lanthanum modified bentonite and aluminium) for controlling P release from bed sediments, particularly in shallow eutrophic lakes (Copetti et al., 2016). The main premises behind this is that, by controlling internal P loading the ecological effects of eutrophication can be rapidly reversed (Mackay et al., 2014; Copetti et al., 2016). However, the chemical behaviour and effectiveness of these materials varies and, therefore, more comprehensive research needs to be carried out before being rolled out as a viable remedial management measure for these eutrophic lake systems (Spears et al., 2013).

5.7 Conclusion

This study has provided new insights into why depth and time related differences occur in phytoplankton biomass in small lakes. In particular, it highlighted the role that climatic factors play in influencing phytoplankton biomass throughout the whole lake profile, directly at the surface, and where the photic zone reaches below the thermocline depth. These insights are important on a global scale, not only because small water bodies such as Milltown Lake (surface area 0.14 km², average depth of 5.5 m) are far more common world-wide than the iconic, charismatic lakes often studied (e.g. Lake Constance; surface area 563 km², average depth of 90 m) (Downing et al., 2006; Verpoorter et al., 2014), but because of on-going changes in local

weather patterns. In addition, this study highlighted how discrete surface sampling, which is a common protocol for many lake nutrient monitoring programmes, may not provide an overall picture of the complex processes occurring throughout the lake's depth profile.

Chapter 6: Influences of multiple drivers on phytoplankton diversity in Milltown Lake during spring stratification

6.1 Abstract

Determining how phytoplankton species or group abundance is influenced by the balance between changes in water column stability, and other factors such as nutrient availability and zooplankton grazing, during the onset of spring stratification is an important element in understanding lake ecosystems. This is particularly so for Milltown Lake given the significant impacts of physical and meteorological drivers of Chl-a identified in Chapter 5. This chapter builds on those findings, focusing especially on the spring and early summer period in Milltown Lake, when the lake structure generally changes rapidly from fully mixed to a stratified state. Data were obtained on the physical characteristics of the lake temperature profile, nutrient concentrations at 1 m intervals, and on the plankton communities, over an intense 10 week sampling period carried out in 2013. Results from the GAM analysis for Milltown Lake, indicated that light and P availability were more influential at the upper epilimnion, while N availability was more influential in controlling the standing stock of phytoplankton at the lower depths. Diatoms (Asterionella sp. and Cyclotella sp.) were identified as the dominate taxa, during the spring bloom formation at both Depth 1 (sub-surface) and Depth 2 (the extent of the photic zone) in Milltown Lake, which conforms with seasonal patterns outlined in the Plankton Ecology Group (PEG) model. Data from the principal component analysis (PCA) showed little influence of zooplankton grazing at either depth. Overall, these findings have outlined how complex a lake's community structure is and that many different factors, both climatic and nutrient related, play a role in its overall status.

6.2 Introduction

6.2.1 Phytoplankton

Phytoplankton are considered to be the dominant producers of energy in lakes and represent the largest source of biomass in a freshwater ecosystem. The growth of phytoplankton and presence

varies seasonally and results from the supply of nutrients and light availability within a lake (Xu et al., 2010; Cunha Pereira et al., 2011; Edwards et al., 2016; de Eyto et al., 2016). In addition, certain phytoplankton groups have the ability to adjust their migration behaviour, shape, size or through adaptation to physical factors such as water movements within a lake. This helps them to remain in the upper photic zone, therefore, improving the interception of light (Coon et al., 1987; Padisa'k et al., 2004; Modenutti et al., 2004; Reynolds, 2006; Carraro et al., 2012; Simmonds et al., 2015). Examples include the Cyanobacteria, a phylum of bacteria that, due to their photosynthetic capabilities, are included among the algal collections. Cyanobacteria remain buoyant in the upper water column by using their gas vacuoles as floatation devices while *Cryptophyta*, *Pyrrhophyta* and *Chrysophyta* stay mobile using flagella. Phytoplankton can also adapt to maximise light absorbance, through adjusting their pigment concentration level by altering their feeding strategies (Proctor and Roesler, 2010). Once these demands are satisfied, phytoplankton can experience rapid growth, leading to large biomass accumulation in the upper layers referred to as phytoplankton bloom (Martinez et al., 2011).

Phytoplankton spring blooms are a global phenomenon in temperate aquatic ecosystems, and the timing and magnitude of the spring bloom prompts much of the subsequent dynamics in a lake ecosystem throughout the year (Platt et al., 2003; Edwards and Richardson, 2004). As a result, a major focus for plankton ecologists has been to explain the mechanisms driving the onset of the phytoplankton spring bloom and its dynamics (Riley, 1942; Sverdrup, 1953; Atkins, 2009; Lewandowska et al., 2015). The role of light availability and water column mixing processes in bloom formation have been emphasised by researchers since the 1920s (Gran and Braarud, 1935; Lewandowska et al., 2015). In the early 1940s, the compensation depth concept was introduced by Riley (1942). This is the depth at which photosynthesis exactly matches respiration rate. In 1950s, Sverdrup (1953), incorporated all of these concepts and proposed the first quantitative explanation of the occurrence of the spring phytoplankton bloom. Since then,

there have been several advances in explaining spring bloom formation. These highlight the role of turbulence, grazing by zooplankton and nutrient concentrations in controlling phytoplankton blooms (Fischer *et al.*, 2014). However, the critical depth hypothesis (Sverdrup, 1953) remains the most widely cited (Lewandowska et al., 2015). In general terms, the spring phytoplankton bloom occurs as a result of the upper water column having sufficient availability of nutrients brought to the surface by deep winter mixing, or by inflows from the catchment, coupled with higher levels of irradiance, all of which combine to give ideal conditions to support rapid phytoplankton growth (Martinez et al., 2011). The magnitude and duration of the spring phytoplankton bloom are mainly controlled by a variety of factors including water temperature, stratification of the water column, availability of light and nutrients, and losses due to zooplankton grazing and respiration (Lucas et al., 1999; Peeters et al., 2007; Behrenfeld 2010; Schabhüttl et al., 2013).

Phytoplankton community structure has been the subject of intense study for many decades (Riley, 1942; Sverdrup, 1953; Atkins, 2009; Deng et al., 2014). The spring bloom is now recognised to consist of a series of sequential blooms involving different phytoplankton species (Lewandowska et al., 2015). Shifts in the dominant phytoplankton species are likely caused by biological and physical factors. For instance, increased nutrient loads are considered to be one of the main drivers of cyanobacterial dominance and blooms in lakes. However, Deng et al. (2014), also suggests that cyanobacteria dominance can shift to chlorophytes under excessive nutrient loading, as chlorophytes in temperate lakes are characterised by high growth and loss rates, and high nutrient demands, while cyanobacteria often have lower growth and loss rates due to their ability to convert inert atmospheric nitrogen into an organic form such as NO₃ and NH₃, thus lowering their demand for nutrients. Furthermore, climate change maybe an additional pressure on the qualitative change in phytoplankton community dynamics. Climate can impact phytoplankton both directly through physiology and indirectly by changing water

column stratification and resource availability, mainly nutrients and light, or intense grazing by heterotrophs (Jeppesen et al., 2009; Jeppesen et al., 2011; Winder and Sommer, 2012). Such modifications affect various phytoplankton processes, and a widespread advance in phytoplankton spring bloom timing and changing bloom magnitudes have both been observed (Brookes and Carey, 2011; Jeppesen et al., 2011). Such shifts in phytoplankton composition can have a far reaching consequence for ecosystem structure and functioning, and thus an improved understanding of the mechanistic links between climate and phytoplankton dynamics is important for predicting climate change impacts on aquatic ecosystems (Winder and Sommer, 2012).

Dominant phytoplankton groups within relatively shallow lake ecosystems include fastgrowing unicellular nanoplanktic forms (e.g. *Chlorella*, *Monoraphidium* spp.), nanoplanktic flagellates (*Chlamydomonas, Plagioselmis, Chrysochromulina*), chrysophycean (especially *Synura* spp.) and small volvocalean colonies (especially *Gonium*) (Reynolds, 2006). Within Ireland, Free et al. (2006) in a study of 201 lakes located across the isle of Ireland, identified Desmids, Merismopedia dinoflagellates (Pyrrophyta), Gymnodinium, Ceratium, diatoms (Bacillariophyta), *Peridinium* spp. and *Dinobryon* spp. to be the most common phytoplankton groups.

6.2.2 Zooplankton

Patterns in zooplankton seasonal succession are mainly influenced by the availability of food and temperature, with an increase in zooplankton assimilation of algae and bacteria occurring at higher temperatures (Wetzel, 2001). There are two dominant groups of zooplankton found in freshwater lakes, cladocerans (e.g. *Diaphanosoma brachyurum*, *Bosmina* sp., *Ceriodaphinia* sp. and *Daphnia* sp.) and copepods (e.g. *Calanoida* spp., *Cyclopoida* spp. and juvenile calanoids or cyclopoids known as nauplii). Zooplankton exhibit diel vertical migration, which refers to the migration of zooplankton to deeper waters during the day and to the surface at night (Williamson et al., 2011; Armengol et al., 2012). Williamson et al. (2011) proposed that this behaviour was a response to biotic factors such as food and predation, and abiotic factors such as light and temperature. Wetzel (2001) described migration behaviour at night as a means of going unnoticed by predators such as fish, and also of allowing zooplankton to take full advantage of the high phytoplankton biomass in the upper waters.

Zooplankton have the ability to influence the seasonal succession of phytoplankton and significantly impact the overall productivity of a lake's ecosystem (Jeppesen et al., 2000; Sommer et al., 2012). Different zooplankton species exhibit diverse behaviours for acquiring their food. Some species of zooplankton as selective feeders, grazing on detrital material and autochthonous phytoplankton. Cyclopoids are predatory and are regarded as selective feeders, feeding mainly on rotifers, flagellates, ciliates and nauplii (Faithfull et al., 2011; May and Wallace, 2019). Calanoids are omnivorous filter feeders and are selective as they asses the nutritional quality of their prey or plankton population (McCarthy et al., 2006). *Daphnia*, in contrast, are known as non-selective feeders and utilise a wide range of particles, grazing on phytoplankton and organic detritus including bacteria (Carney and Elser, 1990; Perga and Gerdeaux, 2006).

6.2.3 Deep Chlorophyll Maximum (DCM)

The vertical distribution of chlorophyll in stratified lakes frequently displays a peak deep in the water column, which is referred to as the deep chlorophyll maximum (DCM) (Simmonds et al., 2015; Leach et al., 2018). The zones where this occurs are ecologically important hotspots for nutrient cycling and primary production, and their position can determine the vertical habitat gradient for phytoplankton (Leach et al., 2018). Deep chlorophyll maxima are commonly found in optically deep, low productivity lakes, and less so in eutrophic systems (Serizawa et al., 2010;

Simmonds et al., 2015). In a study by Hamilton et al. (2010), they suggest that a DCM in a lake formed only when the euphotic zone exceeded the maximum depth of the epilimnion. Several studies have also reported this close relationship, including those by Heaney and Talling (1980), Whittington et al. (2000) and Modenutti et al. (2004). Further drivers for the formation of a DCM in a lake include the vertical distribution of nutrients and the location of the thermal gradients in the water column (e.g. Fee, 1976; Abbott et al., 1984; Cullen, 2015; Leach et al., 2018). It is the thermal stratification layer that determines the depth of the DCM, as it can regulate nutrient mixing from deep waters into the euphotic zone (Abbott et al., 1984; Varela et al., 1994; Mellard et al., 2011; White and Matsumoto, 2012). Additionally, it is believed that the metalimnion offers protection from surface driven mixing that could entrain phytoplankton into the upper mixed layer and exposed populations to potentially harmful levels of irradiance (Abbott et al., 1984; Kamykowski and Yamazaki, 1997; Whittington et al., 2000; Simmonds et al., 2015). Taxa commonly found in the DCM can vary in their behaviour and physical adaptations and may include groups such as the diatoms which have reduced settling rates at the thermocline, buoyancy regulating cyanobacteria, and motile ciliates and dinoflagellates (Coon et al., 1987; Modenutti et al., 2004; Padisa'k et al., 2004; Carraro et al., 2012; Simmonds et al., 2015).

6.2.4 PEG model concept

The Plankton Ecology Group is a conceptual model to explain the seasonal patterns of change in phytoplankton biomass (Sommer et al., 1986; Sommer et al., 2012; De Senerpont Domis et al., 2013). The PEG model uses 24 sequential steps to discuss the abiotic and physical mechanisms and biotic interactions with zooplankton that occur during the spring growing season (Sommer et al., 1986), and remains one of the most cited models in plankton ecology (Carey et al., 2016). According to the PEG model, in eutrophic system, it is expected that the community would exhibit two or three distinct peaks within a year: a spring bloom of diatoms, a summer bloom of chlorophytes and/or cyanobacteria, and potentially an autumn diatom bloom (Sommer et al., 1986, Sommer et al., 2012). Possible triggers for phytoplankton spring blooms in freshwater lakes, as outlined by Reynolds (2006), are the temperature and light regime within the lake ecosystem. However, in an updated PEG model, Sommer et al. (2012), assessed these possible triggers of the phytoplankton spring bloom and concluded that, although temperature played a vital role, light was the dominant factor that triggered blooms in spring. Various deviations from the PEG models have been documented over the past number of years in several lakes (Jeppesen et al., 1997; Alvarez-Cobelas et al., 2005; Moustaka-Gouni et al., 2014), mainly relating to advances in the understanding of the microbial loop, phytoplankton food quality for example. However, the PEG template provided an important starting point for examining lake phytoplankton community dynamics (Free et al., 2006; Pociecha et al., 2010; Padisa'k et al., 2010; Carey et al., 2016).

Milltown Lake was one of eleven lakes in the GLEON Spring Blitz, a project established in 2012 to determine how phytoplankton richness is influenced by changes in water column stability during the onset of spring stratification. The work described in this chapter contributed to that project, but also was expanded to fulfil specific aims for the studies undertaken at Milltown Lake and to build on the work described in Chapter 5. The aims of Spring Blitz were based on the concepts of the PEG model. Sampling was undertaken at the deepest point in the lake to examine the role that water column stability (over space and time) played in determining plankton diversity. Sampling of physical parameters, nutrient concentrations and plankton communities were conducted (~ every 3-4 days) during the spring period of 2013. In addition to that protocol, further nutrient and Chl-*a* sampling was undertaken at 1 m depth intervals in a way that was similar to the sampling protocol seen in Chapter 5 in Milltown Lake, to allow comparison with data from the previous study year (2012).

6.3 Aims of this study

The primary aim of the research presented in this chapter was to study the role that the balance between changes in water column stability, nutrient availability, and zooplankton grazing played in determining plankton diversity in the study lake at the surface and at the photic depth. The latter depth was chosen to represent the extent of the euphotic zone. The study was focused on the spring and early summer period when the lake structure went from fully mixed to a stratified state over a very short time period.

6.4 Methods

Monitoring of Milltown Lake took place in the weeks prior to and during the on-set of stratification, in spring 2013. High resolution data for DO concentrations and % saturation, and temperature were obtained from *in-situ* lake sensors (described in Section 3.5). Analytical methods for nutrients and Chl-*a* analyse for monitoring at the lake inflow and at the deepest point within the lake are described in Sections 3.4 and 3.6 (Figure 3.1b).

6.4.1 Sampling depth calculation and abiotic sampling protocol

Milltown Lake was sampled bi-weekly (each Tuesday and Thursday), at the deepest point in the lake (~10 m), for a period of 10 weeks (18th of April 2013 to the 20th of June 2013). This period included the weeks leading up to and the 5 weeks after the initial onset of thermal stratification (Figure 3.1b). The depth at the deepest point on the day of sample collection was measured using a handheld Hondex (PS-7) Portable Depth Sounder. The approximate timeframe of the onset of thermal stratification was based on the stratification patterns of Milltown Lake in previous years (Figure 5.3a). The onset of the lake thermocline (defined as a one degree centigrade change in temperature per metre) was monitored using a chain of nine TidbiT temperature loggers, recording temperature at 15 minute intervals (Section 3.5).

Two sampling depths were measured: Depth 1 (0.1 m below the lake surface) and Depth 2 (Secchi depth measurement, Figure 6.1) (Luhtala and Tolvanen, 2013). Depth 2 was sampled as being the extent of the euphotic zone and to investigate the occurrence of any potential DCM (Simmonds et al., 2015). The Secchi depth was measured using a 20 cm diameter Secchi disk. The disk was lowered on the shady side of the boat and the depth at which the disk disappeared was recorded to the nearest 0.1 m. The disk was then slowly hauled up and the depth at which the disk transparency was calculated according to the following equation:

Secchi depth =
$$(d_1+d_2)/2$$
 (Equation 6.1)

Where d_1 is the depth (m) at which the disk disappeared and d_2 is the depth (m) at which the disk reappeared; 2 is the number of measurements recorded.

Based on the Secchi depth sampling, the depths determined were calculated as follows:

Depth 1: Subsurface sample (0.1 m below lakes surface) (Figure 6.1).

Depth 2: Secchi depth measurement recorded on sampling day (potential for DCM) (Figure 6.1).



Figure 6.1 Sampling depths in Milltown Lake in spring 2013.

Additional measurements on each sampling occasion included water column DO (mg L⁻¹ and % SAT) using a YSI 556 MPS multi probe with a 30 m extension, to verify the *in-situ* sensor data. The portable YSI multi probe was calibrated in the laboratory prior to carrying out field work and assessed periodically throughout the sampling trip.

Water samples were collected at the deepest point within Milltown Lake, at sampling Depths 1 and 2, and addition at 1 m intervals (1, 2, 3, 4, 5, 6, 7, 8 and 9 m) throughout the lake profile for comparison purposes with the sampling protocol used in 2012 (Chapter 5). Samples were collected using a Fieldmaster[®] Basic Van Dorn sampler and analysed for nutrients, total P, total N, SRP, NH₄-N and NO₃-N (Figure 3.1b). Samples for Chl-*a* were also collected using the Fieldmaster[®] Basic Van Dorn sampler at 0.1, 2, 4, 6 and 8 m, in the same sampling location to allow comparison with the previous sampling year (Chapter 5). Water samples (nutrient and Chl-*a*) were collected using pre-acid washed 2 L polyethylene bottles, rinsed with ultrapure water. Sample bottles were rinsed with lake water from the depth to be sampled prior to sample collection. Methods used for the analysis of total nutrients, dissolved nutrients and Chl-*a* are described in Section 3.4 and 3.6.

Daily composite samples for determining TP and TN load were collected using a HACH[®] Sigma 900 Max autosampler during the study period. The autosampler was located at the EPA staff gauge on the Drumleek River, 500 m upstream from the only lake inflow (Figure 3.1a). The sampler was programmed to collect six 50 ml samples per day at 4 hour intervals, giving a final daily sample volume of 300 ml. The polyethylene sample bottles were pre-acid washed and rinsed with Milli-Q water. Sample bottles were collected on the same days as the lake samples. Stream discharge data from an EPA managed staff gauge provided stream discharge data at 15 minute intervals.

6.4.2 Collection of biotic parameters based on Secchi depth measurements

Phytoplankton and zooplankton samples were collected from Depth 1 and Depth 2 only using a Fieldmaster[®] Basic Van Dorn sampler. Phytoplankton samples were preserved by adding 2-3 ml of Lugol's iodine to a 200 ml glass amber bottle, the bottles were capped and shaken until well mixed. Zooplankton samples were narcotised by adding 2 Alka-seltzer[®] tablets, which narcotises the organisms, to a 250 ml wide necked plastic bottle, to which 99% industrial methylated spirits were added for preservation. All sample bottles were rinsed with lake water from the depth being sampled prior to sample collection.

6.4.3 Phytoplankton counting

Counting of preserved phytoplankton was based on the sedimentation technique developed by Utermohl in 1958, the standard method used in WFD (EN 15204 2006) (European Standards, 2006). The phytoplankton samples were mixed gently by rotating the sample bottle in a figure of eight motion for 20 revolutions to thoroughly mix the contents. A 25 ml Utermohl sedimentation chamber was filled with a pipette to the top with sufficient excess to create a bead of water. A glass cover was then placed on top of the sedimentation chamber to remove the excess water and to enclose the exact column of sample without entrapping water bubbles. Algae were allowed to settle to the bottom of the sedimentation chamber. To ensure complete sedimentation of the sample, the sedimentation time should be twice the volume of the sedimentation chamber (25 ml) (Margalef, 1969), therefore, each sample was allowed to settle for approximately 48 hours prior to counting. Identification and enumeration of phytoplankton samples was conducted under an inverted microscope (Leica[™] S 80/0.30) at a defined magnification. The microscope was linked to a camera that enabled accurate phytoplankton measurements to be calculated from the images. Measurements were taken from images calibrated using a 1 mm stage micrometre. At low magnification (x 100), an individual overview of the content of the chamber was carried out, with records made of any large individual organisms and unusual specimens in the sample which may not appear at subsequent finer examination. Empty cells were not counted. Three randomly chosen transects of the chamber were counted (x 400) to identify and enumerate the organisms present.

Samples were identified to genus and species level, where possible, by using taxonomic references (Huber-Pestalozzi, 1983; Whitton and Brook, 2003; Berllinger and Sigee, 2010). As far as possible, filamentous or colony formations were counted as individual colonies. Phytoplankton abundance (numbers L⁻¹) and biomass (pg C ml⁻¹) were calculated for Depth 1 and Depth 2.

6.4.4 Zooplankton counting

Zooplankton sample containers were inverted a number of times to ensure a mixed sample was being analysed. Samples were initially filtered through a 64 μ m filter and washed with deionised water to remove the alcohol preserve. The sample was then transferred into a graduated cylinder and diluted to 50 ml with deionised water. From this two 5 ml sub-samples were subsequently transferred using a wide bored 5 ml pipette to a Ward rotary chamber. The samples were measured and counted under an Olympus SZX12 stereoscope microscope at a magnification of x 25. Two sub-samples were enumerated from the two sampling depths (Depth 1 and Depth 2) on all sampling dates. Zooplankton individuals were identified to groups (Cladocerans and Copepods) and to genus level (*Daphnia* sp., *Ceriodaphnia* sp., *Diphanosoma* sp. and *Bosmina* sp.), calanoids and cyclopoids, using the Harding and Smith (1974), Pontin (1978) and Scourfield and Harding (1994) identification keys. The first 50 individuals encountered were measured in length and width. Zooplankton abundance (numbers L⁻¹) was calculated based on estimating the number of individuals per litre, while biomass (pg C ml⁻¹) was calculated based on the length-weight relationship (Bottrell et al., 1975; McCauley, 1984; Pauli, 1989).

6.4.5 Other data

Flow and meteorological data (Ballyhaise and Coose Met Éireann weather stations) were available for the 10 week sampling period (Figure 3.1; Figure 3.4). Photosynthetically Active Radiation measurements were estimated from hourly SW radiation from the Ballyhaise Met Éireann weather station. The SW (Wm⁻²) was subsequently converted to PAR (μ E m⁻¹ sec⁻²) by multiplying the SW radiation by 0.45 (Pinker et al., 2010). Daily nutrient loads of the catchment area were calculated for the inflow site as the product of concentration and expressed in g ha⁻¹ day⁻¹. Schmidt stability and thermocline depth were determined from vertical temperature profiles using the LakeAnalyzer programme (Read et al., 2011) (Section 3.7). Thermocline depth calculations excluded the uppermost 0.5 m of the water column due to diurnal surface water heating sometimes creating a strong but temporary (diurnal) density gradient (Simmonds et al., 2015).

6.4.6 Data analysis

Statistical analyses were carried out using the R statistical package version 3.2.2 (R 195 Core Team, 2015) and Microsoft Excel (2013). General additive modelling was used to identify the main drivers of estimated Chl-*a* concentrations. Analysis was carried out using the mgcv package (Wood, 2006) in R (version 3.2.5, R 195 Core Team, 2015). The spreads of the residuals were relatively homogenous, indicating that a Gaussian distribution was appropriate for the model (Zuur et al., 2009). All data sets were assessed for any breaches of the assumptions of independence and heteroscedasticity using the methods in Zuur et al. (2009). All data sets conformed to these assumptions. Principal Component Analysis was used to assess the change in phytoplankton species abundance prior to and after the onset of spring thermal stratification. Analysis was carried out using the vegan package (Wood, 2006) in R (version 3.2.5, R 195 Core Team, 2015). Various data sets were used during the analysis (catchment nutrient loading, internal lake nutrient loading, meteorological data and zooplankton taxa) to separate the effects of individual drivers on phytoplankton species response.

6.5 Results

6.5.1 Meteorological and in-lake vertical temperature profile data

The mean daily air temperature during the study period (18th of April to the 20th June 2013) was 10.6 °C, with a maximum air temperature of 15.3 °C recorded on the final day of the study. The average precipitation was 2.2 mm day⁻¹, with the maximum daily precipitation (13.2 mm day⁻¹) coinciding with a peak in stream discharge on the Drumleek River, occurring the day prior to the lake becoming thermally stratified on the 26th of May 2013 (Figure 6.2a). The average daily wind speed for the study period was 3.4 m sec⁻¹.

A period of higher PAR levels was recorded in late May, with a maximum of 663 μ E m⁻¹ sec⁻² on the 30th of May. The photic depth of Milltown Lake on this date was 5.5 m (Figure 6.2c). Stratification of Milltown Lake occurred during this period of high PAR, and the lake then

remained stratified until the end of the 10 week sampling period. The photic depth of Milltown Lake during this stratified period ranged between 4.4 m and 7.4 m (2.5 x Secchi depth measurement), which was relatively similar to the depth range (3.5 m to 7.5 m) of the photic zone prior stratification. The thermocline depth during stratification ranged from 1 m and then progressively increased to a maximum depth of 3.7 m on the 17th of June 2013. The surface mixed layer temperatures during stratification were relatively low and ranged from 13.1 °C (26th of May 2013) to 19.7 °C (10th of June 2013), while the air temperature during the stratified period ranged between 10.0 °C (31st of May 2013) and 15.3 °C (20th of June 2013) (Figure 6.2d).



Figure 6.2 (a) daily rainfall (mm day⁻¹) at Coose Met Éireann weather station and stream discharge (m³ sec⁻¹) for the EPA staff gauge site on the Drumleek River, **(b)** daily average wind speed for Milltown Lake, **(c)** daily average photosynthetically active radiation (PAR) from the Ballyhaise Met Éireann weather station and **(d)** daily average surface water temperature for Milltown Lake and air temperature from the Ballyhaise Met Éireann weather station, for six month period (January to June 2013). The area between the dashed lines indicates the study period. The red line indicates the day on which the lake stratified (26th May 2013).

Average daily TP and TN loads entering Milltown Lake over the study period from the catchment were 3.2 g P ha⁻¹ d⁻¹ and 44.3 g N ha⁻¹ d⁻¹ respectively. The two weeks prior to stratification (26th May) had the highest nutrient loads from the catchment, with peak loads coinciding with increases in stream discharge (20.6 g TP ha⁻¹ d⁻¹ and 163.0 g N ha⁻¹ d⁻¹) (Figure 6.3a). The mean DO concentration recorded during the 10 week study was 11.0 mg L^{-1} at 1 m, and 7.9 mg L⁻¹ at 7 m. Prior to the onset of stratification (26th of May 2013), DO concentration ranged between 10.0 mg L⁻¹ and 13.8 mg L⁻¹ at the 1 m depth and between 9.1 mg L⁻¹ and 12.0 mg L⁻¹ at 7 m. However, DO concentration declined at the 7 m depth following stratification to a low of 0.4 mg L⁻¹ on the 20th of June 2013 (Figure 6.3b). Milltown Lake became fully stratified on the 26th of May 2013, when the depth of the thermocline was 1.0 m, the stratified layer deepened to a depth of 2.8 m and remained stratified until the end of this study (Figure 6.3c). During this stratified period, the mean water temperature for the surface mixed layer was 14.3 °C, while the mean water temperature beneath the thermocline depth was 11.0 °C. Schmidt stability increased gradually during this stratified period (mean of 28.6 J m⁻²) with a three day period of higher values occurring in early June (57.8 J m⁻² on the 8th of June to 46.6 J m⁻² on the 10th of June 2013) (Figure 6.3d). This coincided with the maximum surface water temperature of 21.4 °C on the 9th of June 2013 (Figure 6.3c).



Figure 6.3 (a) stream discharge (m³ sec⁻¹) and nutrient loadings (g ha⁻¹ d⁻¹) from the lake inlet site, **(b)** dissolved oxygen (mg L⁻¹) at 1 m and 7 m depth. The grey area between the grey dashed line indicates the range for DO at 1 m and the area between the black dashed line indicates the range for DO at 7 m. **(c)** contour plot of lake temperature (°C) with thermocline depth (m) superimposed and **(d)** Schmidt stability (J m⁻²), between the 18 April 2013 and the 20 June 2013. Note difference in scale for TN and TP loads, where TP concentrations are multiplied by a factor of 5. The red line indicates the start of thermal stratification in Milltown Lake.

6.5.2 Lake abiotic data

Phosphorus concentrations in the lake ranged between 0.009 - 0.179 mg TP L⁻¹ for TP and between 0.009 - 0.065 mg SRP L⁻¹ for SRP. High concentrations of TP (mean 0.082 mg P L⁻¹

throughout the water column) were recorded three days prior to the lake stratifying (26^{th} of May 2013), between the 16^{th} of May 2013 and the 23^{rd} of May 2013, coinciding with the pulse in nutrient loading into the lake, but also with high Chl-*a* levels (Figure 6.4a, b). Total P concentrations then decreased following stratification, however, there was a peak in TP levels on the 13^{th} of June 2013, with maximum concentrations ($0.179 \text{ mg P L}^{-1}$) above the photic zone (bottom of photic zone depth measuring 6.7 m) (Figure 6.4b). Increased SRP concentrations (averaged at 0.030 mg P L⁻¹) were seen throughout the water column at the start of the study period (18^{th} of April 2013). Higher concentrations of SRP were evident in the photic zone for the remainder of the study period ($> 0.030 \text{ mg L}^{-1}$), indicating nutrient input from the Milltown Catchment, while below the photic zone SRP concentrations were below 0.010 mg L⁻¹, indicating SRP was not being released from the lake sediment. (Figure 6.4c). The mean SRP concentration in the photic zone was 0.031 mg SRP L⁻¹, while mean concentration below the photic zone was 0.008 mg SRP L⁻¹.

During the first three weeks of the sampling campaign, the mean TN concentrations were less than 1.2 mg N L⁻¹ for all depths. However, similar to TP, concentrations increased throughout the water column prior to stratification (26th of May 2013), with maximum levels reaching 2.1 mg TN L⁻¹ (\leq 1m) on the 16th of May 2013 (Figure 6.4d). During stratification, the mean TN concentration was 1.4 mg TN L⁻¹, with higher concentrations in the lower water depths after the 6th June 2013 (Figure 6.4d). High NO₃-N concentrations were found throughout the water column during the first two weeks of the study period, with concentrations ranging between 0.5 mg NO₃-N L⁻¹ to 1.2 mg NO₃-N L⁻¹ (Figure 6.4e). However, during the stratified period, higher NO₃-N concentrations were generally observed below the photic zone ranging between 0.18 mg NO₃-N L⁻¹ to 0.71 mg NO₃-N L⁻¹. Following stratification, when the DO concentration at the 7 m depth declined (becoming fully anoxic on the 20th of June 2013), NH₄-N concentrations were higher at the lower depths (range: 0.08 NH₄-N L⁻¹ – 0.61 NH₄-N L⁻¹), possibly indicating
release of NH₄-N from the sediment. Ammonium concentrations peaked in the deeper waters on the 30th of May ranging between 0.24 mg NH₄-N L⁻¹ to 0.51 mg NH₄-N L⁻¹ and remained high until the end of the sampling period (Figure 6.4f).

Between the 16th of May 2013 and the 23rd of May 2013, just before stratification but following the high loadings of nutrients to the lake, high Chl-*a* concentrations were recorded throughout the water column, with the maximum Chl-*a* of 62 μ g L⁻¹ recorded on the 23rd of May 2013 at the surface (Figure 6.4a). However, at the onset of stratification 26th May 2013, Chl-*a* concentrations at the surface sampling depth (0.1 m) reduced to 21.7 μ g L⁻¹. During the stratified period, there was a further peak in Chl-*a* concentration (65 μ g L⁻¹) at the surface sampling depth (0.10 m) on the 10th of June 2013. This peak in Chl-*a* concentration at the surface depth reduced to 25 μ g L⁻¹ by the end of the sampling campaign (20th of June 2013) (Figure 6.4a). Chlorophyll *a* at the lower depths were generally lower than those recorded at the surface depths (range: 4 μ g L⁻¹ – 29 μ g L⁻¹). However, on the 13th of June there was a peak in Chl-*a* recorded at 2.7 m (42 μ g L⁻¹), the Chl-*a* at the surface depth on this date measured 23 μ g L⁻¹.



Figure 6.4 Contour plots of (a) chlorophyll *a*, (b) total P, (c) soluble reactive phosphorus, (d) total N, (e) nitrate (NO₃-N) and (f) ammonium (NH₄-N) concentrations with Secchi depth indicated by the solid white line and the bottom of the photic zone indicated by the dashed white line superimposed (m), at the deepest point in Milltown Lake, between the 18 April 2013 and the 20 June 2013. All units are in mg L⁻¹, with the exception of Chl-*a* which is in μ g L⁻¹. The white section at the lower righthand corner of the figure denotes a lowering in lake water level.

6.5.3 Drivers of chlorophyll a concentration

Interestingly, for all depths, nutrient variables were the dominant drivers of change in Chl-a concentrations (Table 6.1). At the surface sampling depth, PAR levels (PAR 3D (p = 0.0002), lake TP concentrations (Lake TP) (p = 0.0106) and Lake SRP (p = 0.0465)) explained 74% of the deviance in Chl-a concentrations (Table 6.1). This GAM for the surface water depth explained the highest % deviance of the models for all depths. The relationships were all positive (edfs of 1.96, 1.90 and 1.00 for PAR 3D, Lake TP and Lake SRP respectively) (Table 6.1; Figure 6.5a, b, c). At the 2 m depth interval, a weak relationship with the TN loading from the catchment (TN catch load) was identified (p = 0.0319), however, this explained only 32% of the deviance (Table 6.1). This weaker relationship for TN was also evident at the next lowest 4 m depth interval (p = 0.0269, R^2 adj = 0.27) (Table 6.1). At the two lower depths (6 m and 8 m) the optimum models indicated that in-lake nitrogen again played an important role in plankton biomass (Figure 6.5d, e). Both Lake TN (p = 0.0009) and Lake NH₄ (p = 0.0147) showed a positive relationship with Chl-a concentrations (Table 6.1; Figure 6.5f, g). A strong positive relationship (edf of 1.9) between Chl-a concentration and Lake TN only was also observed at 8 m (p = <0.0001), explaining 67% of the deviance in Chl-a at this depth (Figure 6.5h).

Level	n	parameter	edf	Р	R ² adj
Surface	18	PAR_3D	1.96	0.0002	0.74
		Lake_TP	1.90	0.0106	
		Lake_SRP	1.00	0.0465	
2 m	18	TN_catch_load	1.68	0.0319	0.32
4 m	18	TN_catch_load	1.32	0.0269	0.27
6 m	18	Lake TN	1.00	0.0009	0.56
		Lake_NH4	1.89	0.0147	
8 m	18	Lake_TN	1.94	< 0.0001	0.67

Table 6.1 GAM model for the drivers of chlorophyll *a* concentration (μ g L⁻¹) at the surface, 2 m, 4 m, 6 m and 8 m depth intervals at the deepest point in Milltown Lake.



Figure 6.5 Estimated smoothing curve for chlorophyll *a* concentration (μ g L⁻¹) at the surface (a) PAR_3D, (b) Lake_TP, (c) Lake_SRP, 2 m (d) TN_catch_load, 4 m (e) TN_catch_load, 6 m (f) Lake_TN, (g), Lake_NH4 and 8 m (h) Lake_TN, depth intervals at the deepest point in Milltown Lake. Grey shaded area indicates the 95% confidence intervals. PAR = photosynthetically active radiation.

6.5.4 Lake biotic data: phytoplankton species composition and abundance

During the initial weeks of the sampling campaign, prior to the onset of thermal stratification in Milltown Lake, peaks in both the Chlorophytes (dominated by the species *Stichococcus sp* and *Mougeotia sp*) and Euglenophytes (*Euglena* spp. and *Trachelomonas volvocina* groups were observed at Depth 1 (0.1 m below the surface) (Figure 6.6a). Chlorophytes contributed 40% to 80% of the total numbers of cells counted during this time (Figure 6.7a). This dominance shifted to the Cyanophytes just prior to the onset of thermal stratification (which then represented between 50% to 70%). These were dominated by *Zygenema* species (Figure 6.6a, Figure 6.7a). The Cryptophyte plankton group (*Cryptomonas sp* and *Rhodomonas sp*) then began to increase at Depth 1, peaking on the 27th of May, the day after stratification occurred in the lake. This group remained dominant until the 10th of June 2013, contributing between 70% to 90% (Figure 6.6a, Figure 6.7a).

At Depth 2 (Secchi depth) in the weeks prior to stratification in Milltown Lake, the Chlorophytes were the main plankton group, contributing between 40% to 55%. These were dominated by *Mougeotia* species (Figure 6.6b, Figure 6.7b). There was a shift in group dominance once the lake stratified on the 26th of May, with the Crytophytes (*Cryptomonas sp* and *Rhodomonas sp*) peaking on the 30th of May 2013 and the Bacillophytes (*Astrionella sp*) peaking shortly after on the 04th of June 2013 (Figure 6.6b). However, toward the end of the sampling period the Chlorophytes once again dominated Milltown Lake at Depth 2, contributing between 50% to 60% of the total phytoplankton population, with the dominant species changing from *Mougeotia sp* (evident prior to stratification) to *Stichococcus sp* (Figure 6.6b, Figure 6.7b).



Figure 6.6 (a) Phytoplankton abundance (Number L⁻¹) counted at Depth 1 (0.1 m below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.



Figure 6.7 (a) Percentage contribution (based on abundance) of Phytoplankton enumerated at Depth 1 (0.1 m below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.

6.5.5 Phytoplankton biomass

In general. in the weeks prior to stratification, the phytoplankton biomass for the different groups at both Depth 1 and Depth 2 in Milltown Lake were considerably lower than on the days after stratification occurred (Figure 6.8a, b). At Depth 1 Cyanophytes (*Zygenema* spp.) biomass peaked on the 13th of May at 27 pg C ml⁻¹ and remained high until the 16th of May (22.9 pg C ml⁻¹) (Figure 6.8a). Even though the Bacillophytes (*Astrionella sp* and *Cyclotella sp*) were not the most numerous species in the days prior to and during the onset of stratification, they had the largest biomass during this period, and steadily increased and peaked at 21.8 pg C ml⁻¹ on the 30th of May 2013 (Figure 6.8a). However, once stratification was established in Milltown Lake, Cryptophyte (*Cryptomonas sp* and *Rhodomonas sp*) biomass dominated, peaking at 59.6 pg C ml⁻¹ on the 06th of June 2013 (Figure 6.8a.).

The Raphidophytes (*Gonyostomum sp*) had the largest biomass in the days immediately prior to stratification at Depth 2 (9th May to the 20th of May 2013) peaking at 30.5 pg C ml⁻¹ on the 16th of May (Figure 6.8b). This then shifted to the Bacillophytes (*Astrionella sp*) which increased in biomass prior to and during the onset of thermal stratification in Milltown Lake peaking at 60.9 pg C ml⁻¹ on the 04th of June 2013 (Figure 6.8b).



Figure 6.8 (a) Phytoplankton biomass (pg C ml⁻¹) counted at Depth 1 (0.1 m below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake

6.5.6 Zooplankton species composition and abundance in terms of population densities.

The three main zooplankton taxa that dominated Milltown Lake in the days prior to stratification at Depth 1 were *Keratella* (20% to 30% contribution by population density), *Ascomorpha* (40% contribution) and *Asplancha* (70% to 85% contribution) (Figure 6.9a, Figure 6.10a). However, once Milltown Lake stratified on the 26th of May 2013, the taxal dominance shifted to *Brachionus* (65% to 75% contribution by population density), which peaked on the 30th of May (Figure 6.10a). This dominance was short lived as the main zooplankton taxa dominance after the onset of stratification was *Keratella* (95% contribution by population density) (Figure 6.9a, Figure 6.10a). Overall, at Depth 2, there was a higher number of zooplankton individuals across all sampling dates (Figure 6.9b). *Keratella* was generally the dominant zooplankton taxa throughout the sampling campaign. *Ascomorpha* (contributing 45%) were but for a short period in the days prior to stratification, peaking on the 13th of May (Figure 6.9b, Figure 6.10b). As seen in Depth 1, the *Brachionus* assemblage increased post-stratification peaking on the 4th of June (Figure 6.9a, b). This was shortly followed by *Conochilus* which peaked on the 10th of June (Figure 6.9b).



Figure 6.9 (a) Zooplankton abundance (Number L⁻¹) counted at Depth 1 (10 cm below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.



Figure 6.10 (a) Percentage Contribution of Zooplankton counted at Depth 1 (10 cm below the surface) and (b) Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.

In terms of biomass crustaceans, in general, were the dominant group for both Depth 1 and Depth 2 throughout the sampling period (Figure 6.11a, b). Although, total biomass in pg C ml⁻¹ peaked in the time period before stratification at 584 pg C ml⁻¹ there was a decline at both depths just prior to stratification. At Depth 1, crustaceans peaked in the days after stratification, while at Depth 2 this peak was not observed until a week after the initial peak seen at Depth 1 (Figure 6.11b). Rotifer biomass was also more abundant at Depth 1 than at Depth 2.



Figure 6.11 (a) Zooplankton biomass (pg C ml⁻¹) counted at Depth 1 (0.1 m below the surface) and **(b)** Depth 2 (Secchi depth), prior to and during the onset of the spring thermal stratification in Milltown Lake.

6.5.7 Drivers of change in phytoplankton group assemblage

All environmental factors considered in the PCA analysis for both Depth 1 and Depth 2 contributed significantly (p < 0.05) to explaining variability in the distribution of phytoplankton groups in the weeks leading up to and after thermal stratification in Milltown Lake. The analysis of the phytoplankton group assemblage at Depth 1 (0.1 m below the surface), showed, however, differences in the environmental drivers for each of the six groups (Figure 6.12a). Cyanophytes and Chrysophytes groups were previously reported as being predominantly evident at Depth 1 prior to the lake stratifying on the 26th of May 2013. These had a strong relationship with mean lake thermocline depth (Figure 6.12a) in the PCA analysis. In addition, the Cyanophytes showed a positive relationship with both in-lake NO₃-N concentration and to mean daily stream discharge to the lake. However, the strongest relationship for this group was with in-lake TN, while the Chrysophytes showed a strong relationship with in-lake NH₄-N (Figure 6.12a). After the lake became stratified, the dominant phytoplankton groups at Depth 1 shifted to the Cryptophytes and Euglenophytes (Figure 6.12a). In the PCA analysis, both of these groups showed a dependency on light (i.e. estimated PAR). Additionally, the Cryptophytes had a strong relationship with the lake's stability (S). The Euglenophytes displayed a positive relationship with in-lake dissolved P concentrations (SRP) (Figure 6.12a).

There was a distinctive difference for some phytoplankton groups in the main relationships identified at Depth 2 (Secchi depth) compared to those identified at the near surface depth (Depth 1). For example, the Bacillophytes, which were present dominant during the onset of stratification at Depth 1, were dominant at Depth 2 after stratification (Figure 6.12b). The Cyanophytes which were present prior to stratification at Depth 1 were also present during this time at Depth 2 (Figure 6.12b). However, the Cyanophytes at Depth 2 had a strong relationship with in-lake NH₄-N concentration, as opposed to in-lake TN concentration as found at Depth 1 (Figure 6.12). In addition, the Chlorophytes were also more dominant in the days prior to stratification at Depth 2. They additionally displayed a strong relationship with in-lake NH₄-N

concentration (Figure 6.12b). However, after the lake became stratified at Depth 2, Cryptophytes again showed a strong correlation with lake stability, similar to that identified for Depth 1 (Figure 6.12b). The Bacillophytes at Depth 2 also had a positive relationship to catchment nutrient loading, particularly TN loads (Figure 6.12b). It is important to note that the Bacillophytes at Depth 1 (during stratification) and Depth 2 (post stratification) had an additional pressure from crustacean grazing.



Figure 6.12 Principal Component Analysis (PCA) of individual phytoplankton species (red) and potential drivers (black), before and after thermal stratification (dates in orange) in Milltown Lake at (a) Depth 1 and (b) Depth 2. WS = windspeed, PAR = photosynthetically active radiation, preD_cath_load = previous day catchment nutrient load, catch_load = catchment nutrient load on the day of sampling, Mean_thermo._depth.= mean lake thermocline depth, SS = Schmidt stability, zooplankton biomass (pg C ml⁻¹).

6.6 Discussion

This chapter focused on the effects of the balance between water column stability, nutrients and grazing in determining plankton abundance and diversity in the lake photic zone. It was informed by the work described in Chapter 5. However, in that year (2012), nutrient levels were consistently high, and the main drivers of phytoplankton biomass (as Chl-*a*) were related to weather and lake physical structure. While in 2013, both external catchment loadings, and internal lake nutrient levels were identified as the main drivers of the phytoplankton community structure. In addition, this study highlighted the differences in phytoplankton abundance at different depths and, therefore, drivers of both phytoplankton community composition and abundance were explored, in particular the possible drivers of any DCM.

Overall, climatic variables such as windspeed and light availability for the sampling period (18^{th} Apr – 20^{th} Jun) in 2013 were similar to those for the same time period in 2012. However, as previously mentioned, 2012 had much higher than average rainfall, which lead to a persistent nutrient loading throughout the summer, leading to high P and N levels in the water column. Nutrient loading from the catchment after rainfall events were also evident in 2013. However, these were much lower than those recorded in 2012. Jennings et al. (2012) describes higher rainfall events leading to increased nutrient loading during spring and early summer as being commonly seen for many western and north western Europe lakes. For Milltown Lake, the rainfall events experienced during spring 2012 also contributed to the periodic breakdown of the lakes thermal structure, with stratification of the lake during this time not exceeding 19 days. It is not known how long the lake stratified for in 2013. However, it was apparent that the lake was more stable when compared to the same sampling period in 2012, as stratification exceeded the 5 week sampling period (~ >35 days). Calmer weather conditions that result in higher lake stability and more frequent algal growth, are generally not expected until the mid to late summer period for western European lakes (Christensen et al., 2007; Sammuelsson,

2010; Jennings et al., 2012). Regardless of not knowing when or what caused the mixing of Milltown Lake in 2013, it is clear that the expected calmer weather conditions, as suggested by Christensen et al. (2007), Sammuelsson (2010) and Jennings et al. (2012), were experienced much earlier in the spring of 2013.

Results presented show clear differences in the drivers (e.g. light versus nutrients) of phytoplankton biomass in sub-surface and potential DCM depths of Milltown Lake during the 2013 sampling campaign. At the sub-surface depth (Depth 1) light and phosphorus availability were the significant drivers of phytoplankton biomass. This was particularly evident for dissolved P, with high concentrations only becoming available in the sub-surface layer in the days prior to and once the lake stratified. While at Depth 2 (potential DCM), nitrogen loading from the Milltown Catchment was identified as the significant driver. In contrast to 2012 where climatic variables and lake physical structure were the main drivers of plankton biomass, the results for the 2013 spring bloom show the importance of nutrient availability in phytoplankton growth.

Martinez et al. (2011) described the occurrence of the phytoplankton spring bloom as a product of sufficient nutrient availability in the upper surface water column coupled with higher irradiance levels, as seen in Milltown Lake in 2013, being the fundamental factors in supporting rapid population growth. The utilisation of light by phytoplankton migrating to the upper surface waters is essential for species survival (Reynolds, 2006). Simmonds et al. (2015) described in their study of lake Õkaro, New Zealand, how sunlight penetrating the surface waters of a lake, inadvertently result in rising water temperatures, a factor which is also essential for phytoplankton growth. However, as described in the updated PEG model, despite the role of temperature being vital in plankton growth, it is light which is considered the dominating factor to trigger the phytoplankton spring bloom (Sommer et al., 2012). The requirement for adequate levels of PAR was identified as a significant driver of the phytoplankton biomass at the sub-surface depth in Milltown Lake. In addition, in-lake P availability was also found to be significant at this depth for the accumulation of phytoplankton once the lake became stratified. Overall, the relationship with the drivers PAR and P explained 74% of the deviance in Chl-*a*, and therefore, phytoplankton abundance. Phosphorus concentrations in the lake at the start of the sampling campaign were again high throughout the water column, similar to the preceding year. This did not coincide with a peak in P loading from the catchment. Phosphorus that has accumulated within the lakes sediment under anoxic conditions can be re-introduced to the surface waters through deep winter mixing (Wang et al., 2008; Hupfer and Lewandowski, 2008; Martinez et al., 2011). However, in the days prior to the lake stratifying, high concentrations of dissolved P became evident only in the surface mixed layer of Milltown Lake, thus suggesting that the sediment was not a source P at this time. This peak in P coincided with high P loadings from the catchment.

Catchment nutrients were, again, found to influence plankton growth at Depth 2. Dominance shifted from catchment P concentrations driving plankton growth in the sub-surface depth to catchment N concentrations stimulating plankton growth at Depth 2 (32% deviance explained). Continuous high N concentrations from the lake's inflow in the two weeks prior to the lake stratifying on the 26th of May, became available throughout the lakes profile. This highlights how nutrient loadings from the catchment during critical time windows can have significant implications for phytoplankton biomass, particularly in the lakes photic zone.

As mentioned previously, the PEG model conceptualises the seasonal dynamics of plankton growth (Bentrup et al., 2016). The model hypothesises that physical controls are mainly attributed to plankton growth at the start and end of the plankton growing season and is not a contributing factor throughout the season (Bentrup et al., 2016). The first step of the model highlights how the availability of nutrients and light towards the end of the winter period contribute to a small crop of fast growing algae such as Cryptophytes (Sommer et al., 1986; 2012). This only became evident in the sub-surface depth of Milltown Lake where the tail end of a Cryptophyte peak was observed. This peak coincided with high levels of SRP within the sub-surface depth. Sommer et al. (1986) notes in their paper that Cryptophytes are capable of depleting this form of P to near undetectable levels. A depletion in SRP concentrations was noted in the sub-surface waters of Milltown Lake after the peak in Cryptophytes, however, levels were still detectable.

At the same time as the Cryptophytes peaked in the sub-surface of Milltown Lake a peak in Cyanophytes (Zygenema spp.) was recorded at Depth 2. It appears that as the days progressed, the Cyanophytes (Zygenema spp.) migrated up to the sub-surface depth, with biomass at Depth 2 decreasing while at the sub-surface an increase in biomass was recorded. Cyanobacteria have the ability to form motile filaments, called hormogonia, which can travel away from the main biomass to form new colonies elsewhere (Reynolds, 2006). In addition, the PCA analysis carried out in this study highlighted the strong dependency this species has with in-lake N. Cyanobacteria are said to favour elevated NO₃ and NH₄ concentrations within a lake's ecosystem (Beversdorf et al., 2013; Paerl and Otten, 2013). Jackson et al. (2016) reported similar dominance of Cyanophytes prior to stratification in their study of several lakes in the USA and China. They demonstrated how Cyanophyte dominance correlated significantly with elevated NO₃ and NH₄ concentrations in the lake. Jackson et al. (2016) attributed this association to the ideal ratio of NO₃ and NH₄ concentrations in the water column. There was short period after the Cyanophytes at Depth 2 declined, where a Raphidophytes taxa (Gonvostomum sp.) was recorded. The numbers of this taxa were low (only two individuals were observed). This species is known as a nuisance alga mainly due to their size $36 - 92 \mu m$ and issues with drinking water taste and odour episodes. (Johansson et al. 2013; Watson et al., 2015).

The PEG model surmises that once the fast growing algae have developed, the next stage involves grazing by herbivorous zooplanktonic species, whose population increase is due to the availability of food (Sommer et al 1986; 2012). Within Milltown Lake, both crustaceans and rotifers increased in response to the availability of food, with lag peaks for both the sub-surface and Depth 2 being observed following the Cryptophytes and Cyanophytes blooms. Following, this initial grazing by the zooplankton, new planktonic herbivores with short generation times begin to increase their population (Sommer 1986; 2012; Davis et al., 2012; Ger et al., 2016).

Diatoms have been recognised as the most opportunistic of the phytoplankton species as far as taking advantage of nutrient availability is concerned (Olli et al., 1996; Çelik and Ongun, 2007; Frasquet et al., 2012) and are said to prefer well mixed water columns where turbulence keeps them in the photic layer (Reynolds, 2006; Watson et al., 2015). In the days prior to stratification in Milltown Lake, turbulence due to increased wind speed was evident in this study, which contributed to a dominance shift from the Cyanophytes to the Bacillophytes (*Astrionella sp, Cyclotella sp*) in the sub-surface depth. The Bacillophytes bloom at Depth 1 was short lived, most likely due to the decline in windspeed, increased water column stability, and the high settling rates of diatoms (Reynolds, 2006). With a decline in windspeed noted during this time, it is probable that the diatoms (due to them being non-motile) settled down to Depth 2 just as the lake began to stratify (Monero-Osten et al., 2009). Furthermore, a peak in crustaceans was also observed during this period at the sub-surface depth, which may have provided additional pressure on the decline in population. The PCA analysis of the phytoplankton assemblage indicated catchment P to be the limiting factor at Depth 1, while catchment N loading was the limiting factors for the Bacillophytes at Depth 2. It is generally reported that diatom blooms in

freshwater lakes are commonly associated with high in-lake P levels (King et al., 2000; Bradshaw et al., 2002; Ramstack et al., 2003; Wang et al., 2012; Watson et al., 2015), which was evident in the sub-surface depth. However, a study by Scotese (2008) indicated that both P any N can be important component in limiting the growth of diatom species. This became apparent in Milltown Lake, as windspeed decreased the adaptability of the non-grazed, nonmotile diatom species allowed them to adjust from the P rich environment found in the subsurface to the N rich environment found at Depth 2.

As a result of diatoms migrating to the lower depth, the Bacillophyte bloom at Depth 2 was not evident until approximately a week after the initial bloom was seen in the sub-surface depth. It is apparent that the diatoms at Depth 2 were also grazed by crustaceans as a lag peak in the zooplankton productivity was also observed for this depth. Sommer et al. (1986; 2012) outlines in step 5 of the PEG model that, as a consequence of zooplankton grazing, phytoplankton species decrease rapidly to very low levels, creating a clear water phase. This becomes apparent in Depth 2 after the diatom bloom, because an increase in alternate diatom population is not seen before the end of the sampling campaign. In addition, wind speed during this time decreased, which could also contribute to the decline in the diatom community at Depth 2. Monero-Osten (2009) studied the impact of short-term wind induced mixing and found a decline in positively buoyant diatoms in the epilimnion became apparent during periods of low wind speeds. However, this is not the case for Depth 1, because a further population increase in Cryptophytes is seen at the same time as the Bacillophytes bloomed at Depth 2.

After the depletion of the Bacillophytes in the sub-surface depth and stratification has been established in the lake, both the Cryptophytes (*Cryptomonas sp, Rhodomonas sp*) and Euglenophytes (*Euglena* spp. and *Trachelomonas volvocina*) became the dominant groups at the sub-surface depth. Similarly, Harris et al. (2016) reported a spring bloom of Cryptophytes

during the stratification of four reservoirs in the Midwestern United States. Their study attributed this to increased lake temperature and stability all of which are favoured by the Cryptophyte taxa. Principal component analysis for this study also identified this dependency with lake stability. Regarding the dominance of Euglenophytes, various studies which observed changes in group assemblage during lake stratification have identified high nutrient levels and decreasing turbulence as their main drivers (Olli et al., 1996; Çelik and Ongun, 2007; Frasquet et al., 2012). Euglenophytes have been identified as suitably adjusted to small lakes (Zakrys et al., 2004), like Milltown Lake. The PCA analysis at the sub-surface depth of Milltown Lake identified both in-lake N and lake stability as the main drivers of the Euglenophytes assemblage, similar to studies by Çelik and Ongun (2007) and Frasquet et al. (2012). However, in addition to stability and nutrient availability, their analysis also indicated that the breakdown of the Cryptophyte taxa was due to the equal grazing pressures of both the crustaceans and rotifers.

The findings in this chapter demonstrated that different factors are playing an important role in determining the assemblage of diverse phytoplankton groups at the different depths within Milltown Lake. Reynolds (2006) describes how temperature and light are the main variables triggering a spring bloom formation. Further work by Sommer et al. (2015) argues that, although temperature plays a vital role in the formation of the spring bloom, it is in fact light that triggers blooms. This can be seen in the GAM analysis for Milltown Lake, which demonstrates how light and nutrients were the main driver of the spring bloom in 2013. Regarding dominant species composition within a spring bloom, Sommer et al. (2015) describes how diatoms are commonly found in eutrophic systems during the spring formation. This theory can also be applicable to Milltown Lake, as the dominant taxa during the formation of the spring bloom were the diatoms (*Asterionella* sp. and *Cyclotella* sp.) at both the sub-surface and the lower depth.

6.7 Conclusion

This chapter has improved understanding of how phytoplankton taxa, and therefore phytoplankton biomass, respond to the combination of water column stability, nutrients and grazing at different depths within the photic zone. The chapter built on the findings of Chapter 5, which highlighted the role of climatic factors in influencing phytoplankton biomass. That year (2012), however, was usually wet and therefore, there was a high nutrient loading to Milltown Lake throughout that summer. This additional study, in 2013, has demonstrated that the drivers of phytoplankton biomass can change on a year-to-year basis as, in 2013, both external catchment loading and internal lake nutrient levels were the main drivers of the phytoplankton community, with distinct changes also occurring at the time of stratification. These insights are important in highlighting the complexity of the controls on a lake's community structure and illustrating that both weather-related changes in stability, and nutrient loading and availability, affect the Chl-*a* concentration in a lake. In particular, this chapter has highlighted how long-term monitoring programmes need to be adapted to gain a true understanding of the different processes occurring within a given lake so that these processes can be taken into account during lake management activities.

Chapter 7: Final Discussion

7.1 General overview

It has long been recognised that nutrient enrichment of freshwater ecosystems plays a key role in the amplification of algal blooms (Paerl and Fulton, 2006; James et al., 2009; Cai et al., 2012). In Ireland, nutrient enrichment of freshwaters, particularly from agricultural sources, has been highlighted as an important environmental issue (Kelly, 2009; Bradley et al., 2015; Deakin et al., 2016). Agriculture in the ROI accounts for 67% of the total land area (CSO, 2015), which is linked through an intricate network of lakes and rivers that flow through the landscape. It is currently estimated that 64% of the rivers and lakes within the ROI are currently *At Risk* of not meeting their environmental objective of good ecological status (Department of Housing, Planning, Community and Local Government, 2018). Of the estimated 64% of river and lakes failing to meet the WFD objectives, 53% are suspected to be as a consequence of agricultural sources of pollution (Bradley et al., 2015; Deakin et al., 2016).

In recent years, it is becoming more apparent that climate change, either directly or indirectly, may cause qualitative changes in the dynamics of the phytoplankton communities within lakes (Stella et al., 2007; Jackson et al., 2007; Adrian et al., 2010; Sommer et al., 2012; de Senerpont Domis et al., 2013; May and Spears, 2015; Havens et al., 2016; Paerl et al., 2016; Dalton et al., 2016). Factors such as rising air temperatures have been reported to encourage undesirable plankton species such as cyanobacterial species (O'Neil et al., 2012; Deng et al., 2014; Rigosi et al., 2014; Woolway et al., 2019). However, other environmental changes including rising water temperatures and increased rainfall leading to higher nutrient runoff from the surrounding catchments are also increasing eutrophic conditions in freshwater ecosystems (Jeppesen et al, 2009; Moss et al., 2010; Deng et al., 2014). The current study aimed to explore the relative effects of nutrient export from a rural catchment and of local meteorological conditions as

drivers of change in phytoplankton biomass in a small eutrophic lake and, in turn, provide valuable information that can be used to inform the future management of these systems .

7.2 Overview of the trophic status of Milltown Lake and links to catchment nutrient export

Analysis of the water quality in Milltown Lake indicated that the lake was classified as strongly to highly eutrophic during the course of this study. The maximum Chl-*a* concentration recorded in 2012 was 44.5 μ g L⁻¹ and 2013 was 65.0 μ g L⁻¹. Milltown Lake's classification was based on the modified O.E.C.D scheme, which determines status from annual maximum Chl-*a* concentration (O.E.C.D, 1982; EPA, 2015). Previous work carried out by Carson et al. (2015), which included water sampling over two years (2007-2008) and a paleolimnological investigation, concluded that the lake had fluctuated between mesotrophic and eutrophic conditions in the recent past. Furthermore, from the analysis of four years of high frequency temperature profile data as part of the current study, the lake was confirmed to be slightly polymictic in nature. The data presented for 2012-2013 also indicated that it experienced periods of hypolimnetic oxygen depletion during the summer months. Similar findings were reported by Carson et al. (2015) for Milltown Lake during 2007-2008 based on temperature and DO profiles taken at weekly intervals.

It is of note that, five years on from the Carson et al. (2015) study, Milltown Lake has not shown any improvement in classification. In fact, the results presented in this thesis may indicate further deterioration, as the lake was classified as strongly to highly eutrophic in 2012 and 2013. At the very least, the status had not improved. This lack of any improvement was despite the implementation of fencing on one third of the catchment streams as described and investigated in Chapter 4. The paleolimnological study carried out by Carson et al. (2015) concluded that agricultural practices in the Milltown catchment had been impacting the lake since the 1970s and thus were the leading contributors to nutrient loading. Those results led to the installation of cattle exclusion fencing in 2008, with a 1.5 m vegetative buffer along the western most tributary to support a reduction in nutrient pollution from cattle-based agriculture in the Milltown Catchment.

The current study investigated the impact of this exclusion fencing by comparing the nutrient concentrations from a fenced stream reach with those of an unfenced stream reach that had similar land management practices. It was found that the annual estimated TP and TN export rates from the catchment area were reduced in the fenced tributary (0.57 kg TP ha⁻¹ yr⁻¹ and 5.99 kg TN ha⁻¹ yr⁻¹) compared to the unfenced tributary (0.77 kg TP ha⁻¹ yr⁻¹ and 7.24 kg TN ha⁻¹ yr⁻¹). However, the results also indicated that the exclusion fencing was only effective during the grazing season, when cattle were in the fields, nutrient uptake by plants would be at its highest, and buffer vegetation biomass would be at its greatest. Similar findings in relation to seasonality in the mitigation potential of fencing and associated vegetation have been reported in a number of other studies. Dorioz et al. (2006), Stoate et al. (2009), Roberts et al. (2012) and McConnell et al. (2013) all found that bioavailable P and N uptake by the vegetation within the buffer were at their highest during the growing season. In addition, the capacity of the vegetative buffer to trap particulate nutrients were higher at the time of year when the vegetation was denser (Dorioz et al. 2006; McConnell et al. 2013).

In contrast, the findings from the current study also indicated that streamside exclusion fencing was not effective in the management of TP in the late winter and early spring when the vegetation cover in the buffer zone had died back. A similar result was reported by Stoate et al. (2009) and Hoffmann et al. (2009), who also concluded from their work that grassy buffer strips were ineffective in retaining P during early spring, particularly after heavy rainfall events, when vegetation was underdeveloped. This time period will also coincide with the time when slurry

spreading resumes in Ireland following winter, which would be after the 31st of January in this catchment under the 2017 Good Agricultural Practice for the Protection of Waters Regulations (S.I. No 605 of 2017). In the current study, TP loads from the fenced catchment exceeded those from the unfenced catchment at this time of year. It could be argued that the overall effect of this was that fencing delayed the export of P to the lake until a time of the year when its impact may actually be greater, i.e. in the spring prior to the main period of algal growth in the lake.

Despite this possible spring effect, the study of the in-lake responses indicated that nutrient control using exclusion fencing is important during the most critical time window for algal blooms, i.e. early to late summer. However, it should be noted that the retention time of Milltown lake is short (25 days), and therefore, this lake will have a rapid response time to increased or decreased catchment nutrient export. The effect of increased TP loading during the early spring in lakes with a longer response time would likely be prolonged into the main algal growing season. Similar cattle exclusion measures to those implemented in Milltown Lake have recently been included in the 2017 GAP Regulations. These stated that livestock are not permitted to drink directly from waterbodies, and that river and streams should be fenced off with a 1.5 m buffer from the 1st of January 2021 (S.I. No. 605/2017). Fencing of the remaining tributaries as a catchment nutrient reduction measure in the Milltown Lake Catchment could further decrease loading to Milltown Lake during the summer but could also increase TP loading during the early spring, and therefore, may not improve the trophic status of the lake.

7.3 Drivers of phytoplankton biomass within small lakes

The results presented in this thesis also showed that Milltown Lake was highly responsive to meteorological factors, which affected the lake thermal regime during the spring and early summer stratification and the export of nutrients to the lake. Ireland has a milder, more moderate oceanic climate, with high precipitation rates and variable wind speeds (Keane, 1986), than

many regions of a similar latitude. The future climate projections for Ireland include rising air temperatures and increased rainfall events, resulting in wetter winters and drier summers respectively (Sweeney, 2008; Gleeson et al., 2013; Matthews et al., 2016). It is of note that the largest range in air temperature and lowest recorded rainfall was seen for the Milltown Catchment in the summer of 2010 compared to the subsequent years studied in this research (2010 – 2013). In response to this, Milltown Lake stratified for longer periods in 2010 than in the two years when the more detailed lake study was conducted (2012 and 2013), with much stronger buoyancy frequency, and a more consistent stable thermocline. Such warmer summers are projected to occur more frequently in Ireland in the future (Gleeson et al., 2013; Matthews et al. 2016) and will likely result in longer and more intense stratification in the lake. However, future projections also include wetter periods in the spring and early summers (Sweeney, 2008; Jennings et al., 2012; Gleeson et al., 2013; Matthews et al., 2016; Coffey et al., 2016). In 2012, higher rainfall over the summer also contributed to higher TP and TN export from the surrounding catchment, and separately contributed to the initial breakdown of thermal stratification. Jennings et al. (2012) noted that similar intense precipitation events may result in physical disturbances of the water column and, in addition, can be accompanied by pulses of nutrient exported from the surrounding catchment to the lake. The results of this study in the Milltown Catchment illustrated the effect of particularly heavy rainfall events on the nutrient levels in 2012 to a lake that was already affected by high nutrient loading (Jordan et al., 2005; Macintosh et al., 2011). However, interestingly, the lake was not nutrient limited in 2012 and, therefore, lake phytoplankton was even more sensitive to climatic factors than in 2013.

When both study years (2012 and 2013) are taken together, it was clear that both climatic factors and nutrient availability played a significant role in driving in-lake Chl-*a* levels, but that this is very much dependent on weather conditions in a given year. In spring/summer 2012, the main drivers of Chl-*a* biomass were climatic factors, although these affected the lake differently over

the depth profile. The upper 2 m were influenced by mixing forces from wind speed and stream discharge, whereas the lower depths were influenced by light availability and NH₄ release under anoxic conditions from the lake sediment. However, in spring 2013, PAR was the only significant climatic factor of the variables assessed that influenced Chl-*a* concentrations, with nutrients, especially N, affecting the phytoplankton biomass.

It is worth noting that, even though different climatic variables were important drivers of phytoplankton biomass, in 2012 and 2013 the spring bloom appeared to be triggered by light (i.e. PAR) from the upper surface to 2 m depth. This is in line with the updated PEG model of Sommer et al. (2012), which assumes that, although temperature plays a vital role in spring bloom formation, it is light that triggers spring bloom formation. Regardless, studying the responses in the two contrasting years has highlighted the complexity of the Milltown Lake ecosystem and how rapidly it changes and responds to constantly changing weather patterns and nutrient inputs.

7.4 Implications of management

Limnological investigations increase our understanding of lake dynamics and help to explain changes in water quality over the annual cycle. However, as this study has shown, it is the larger catchment-scale that needs to be studied, not only the internal functioning of the lake, or indeed the catchment, as single entities. This study has highlighted the complexity of lake ecosystems and how readily they respond to their changing environment. One of the suggestions made by Carson et al. (2015) for the Milltown catchment was to reduce soil and nutrient transfers at their source, which the authors concluded, would be the only sustainable way to improve water quality. However, Deakin et al. (2016), showed that, to reduce nutrient transfer, the pathways of these nutrients need to be targeted from source to receptor, rather than simply the source alone, and that mitigation measures need to be implemented on a case by case basis rather than on a national scale. This study has shown that measures such as streamside fencing, which is a remedial measure commonly used for the reduction of soil and nutrient transfer, is not always effective. To employ streamside fencing effectively within a catchment, additional management strategies such as farm to field nutrient management programmes also need to be operated concurrently. In addition, the surrounding landscape of a given catchment would need to be assessed to see whether fencing would be a viable option. Milltown Lake Catchment is situated within a rural agricultural setting in a drumlin landscape of Ireland and the surrounding topography often does not lend itself well to nutrient reduction though the use of vegetative buffers, especially during hydrologically active periods and during times of high rainfall when the floodplain may exceed the buffer width.

Carson et al. (2015) suggested further that nutrient loadings from the surrounding catchment, either from agricultural sources or domestic wastewater treatment systems, were the leading factors contributing to Milltown Lake's overall load. Studies by McCarthy et al. (2012) and Rafferty (2015) also showed that high nutrient concentrations from the unsewered rural population in the catchment were likely to have impacts on the water quality, but these were found to be much lower than those from agricultural sources. Within Ireland, regulations have been put in place that impose time constraints on the spreading of inorganic and organic fertiliser on land and a limit on the amount of livestock manure that can be spread on the land (S.I. No. 605 of 2017; Buckley, 2012). These regulations will be of benefit to catchment load nutrient management. However, when external loads to rivers and lakes are reduced, it is likely that legacy issues will mask the immediate impacts of these contemporary conservation measures (Meals et al., 2010; Spears et al., 2012). These legacy issues, associated with the remobilisations and recycling of P in the catchment, may continue to contribute to the overall nutrient load for years, decades, or even centuries to come (McDowell et al., 2002; Kleinman et al., 2011; Sharpley et al., 2013).

Furthermore, the current study showed that, not only are external nutrient sources from the catchment playing a role in maintaining Milltown Lake's eutrophic status, but internal nutrients from sediment release may also have an influence. The factors that are mainly influencing this re-suspension are periods of anoxia within the hypolimnic layer. While the negative impacts associated with this phenomenon can potentially be addressed with engineering solutions through the oxygenation of hypolimnetic waters during stratification (Beutel, 2006) or chemically increasing the P binding capacity of the sediment (Wauer et al., 2005), such an approach does not offer long term sustainable solutions to deterioration in water quality (Lürling et al., 2016; Huser et al., 2016). In addition, this study found that in Milltown Lake P concentration was not the only nutrient leaching from the bed sediment but also NH₄, which played a key role in enhancing phytoplankton biomass within the lake.

Finally, while it is important to have a good understanding of how a lake responds to external and internal nutrient loads, this study has also highlighted the potential sensitivity of a lake to meteorological influences. Forecasting a lake's response to future climate change scenarios can be challenging, especially on a local to regional scale (Paerl et al., 2016). However, as this study has demonstrated, future lake management strategies need to not only consider the role of nutrient loads but also the potential influence of climate change when developing future measures, a factor that is normally not considered.

7.5 Overall conclusion

Previous work in the Milltown catchment highlighted the impact of external nutrient loading on Milltown Lake's ecosystem, with losses from agriculture being one of the main drivers. The evidence provided by this study has confirmed that Milltown Lake is still heavily impacted by external nutrient loads. However, the impact of internal lake nutrient re-suspension and climatic factors also play a role in the lake's ecology. It is apparent that agriculture is a significant pressure in the overall contribution of nutrients to Milltown Lake. However, for water quality in Milltown Lake to improve, management strategies need to be considered for the ecosystem science of the entire catchment including the lake itself. It is important to continuously strive for improvements, particularly as the lake is used as a drinking water source, fishing amenity and as a habitat for a variety of different plant and animal species.

In addition, this study has demonstrated that streamside fencing as an exclusion measure can be beneficial within the catchment, particularly during the summer months which is a critical time period for the lake. However, this measure can be costly. Therefore, before installing fencing it is important to consider both the positive and negative implications and weighing up the cost benefit, before considered it a viable solution. Furthermore, this thesis has demonstrated the complexity involved for both stakeholders and policy makers in implementing a universal strategy that will reduce nutrient pollution on a national scale. Each catchment and lake respond differently to different management strategies. Also, each year, different climatic variables influence different freshwater ecosystems in different ways.

Finally, this thesis has demonstrated how developing a good database which provides an overall picture of the catchments behaviour with respect to nutrient loss and its response to local climatic stresses is important. However, it will be important that the information can be of benefit to the stakeholders within the community (i.e. the local community, GWS, farmers, angling groups etc.), in order for them to build on this knowledge. It may, therefore, be feasible based on these finding to encourage a Citizen Science based approach in this catchment, at a much reduced sampling scale. This will still allow for crucial information to be gathered, and allow for further management strategies to be developed, which will further aid water quality improvement. The findings of this study and through community engagement, the information obtained will not only be beneficial to catchments with similar river lake continuums, but also

to neighbouring communities through the encouragement of local stakeholder engagement. This approach can give the local community a sense of ownership and connection with their local waterbodies.

References

- Aalen, F., Whelan, K., Stout, M. (1997). Atlas of the Irish Rural Landscape, Cork: Cork University Press.
- Abell, J.M., Hamilton, D.P., Rutherford, J.C. (2013). Quantifying temporal and spatial variations in sediment, nitrogen and phosphorus transport in stream inflows to a large eutrophic lake. *Environmental Science: Processes & Impacts*, 15(6), 1137-1152.
- Abbott, M. R., Denman, K. L., Powell, T. M., Richerson, P. J., Richards, R. C., Goldman, C. R. (1984). Mixing and the dynamics of the deep chlorophyll maximum in Lake Tahoe. *Limnology and Oceanography.* 29, 862–878.
- Adrian, R., Gerten, D., Huber, V., Wagner, C., Schmidt, S.R. (2012). Windows of change: temporal scale of analysis is decisive to detect ecosystem responses to climate change. *Marine biology*, **159**(11), 2533-2542.
- Adrian, R., O'Reilly, C.M., Zagarese, H., Baines, S.B., Hessen, D.O., Keller, W., Livingstone, D.M., Sommaruga, R., Straile, D., Van Donk, E. and Weyhenmeyer, G.A. (2009).
 Lakes as sentinels of climate change. *Limnology and Oceanography*, 54, 2283-2297.
- Agawin, N.S., Duarte, C.M., Agustí, S. (2000). Nutrient and temperature control of the contribution of picoplankton to phytoplankton biomass and production. *Limnology and Oceanography*, **45**(3), 591-600.
- Akaike, H. (1974). A New Look at the Statistical Model Identification, IEEE Transactions on Automatic Control, **19**(6), 716-723
- Álvarez Cobelas, M., Rojo, C., Angeler, D.G. (2005). Mediterranean limnology: current status, gaps and the future.
- American Public Health Association, Water Pollution Control Federation & Water Environment Federation. (2005). *Standard methods for the examination of water and wastewater* (21st Ed). American Public Health Association.
- Aneja, V.P., Schlesinger, W.H., Erisman, J.W. (2008). Farming pollution. *Nature Geoscience* 1, 409-411.
- Archfield, S.A., Vogel, R.M. (2010). Map correlation method: selection of a reference stream gage to estimate streamflow at ungagged catchments. *Water Resource Research*, 46, 10513.
- Armengol, X., Wurtsbaugh, W.A., Camacho, A., Miracle, M.R. (2012). Pseudo-diel vertical migration in zooplankton: a whole-lake 15N tracer experiment. *Journal of plankton research*, 34(11), 976-986.
- Art, H.W. (1993). Eutrophication, *in* Art, H.W., ed., A dictionary of ecology and environmental science (1st ed.): New York, Henry Holt and Company, 196.
- Atkins, W. R. G. (2009). The phosphate content of Sea Water in relation to the growth of the Algal Plankton. Part III. *Journal of the Marine Biological Association of the United Kingdom*, 14, 447
- Barker, T., Hatton, K., O'Connor, M., Connor, L., Moss, B. (2008). Effects of nitrate load on submerged plant bio-mass and species richness: results of a mesocosm experiment. *Fundamental Applied Limnology*, **173**, 89-100.
- Bartley, R., Corfield, J.P., Abbott, B.N., Hawdon, A.A., Wilkinson, S.N., Nelson, B. (2010). Impacts of improved grazing land management on sediment yields, Part 1: Hillslope processes. *Journal of Hydrology*, **389**(3), 237-248.
- Batáry, P., Dicks, L.V., Kleijn, D., Sutherland, W.J. (2015). The role of agri-environment schemes in conservation and environmental management. *Conservation Biology*, 29(4), 1006-1016.
- Beardall, J., Ihnken, S., Quigg, A. (2009). Gross and net primary production: closing the gap between concepts and measurements. *Aquatic Microbial Ecology*, **56**(2-3), 113-122.
- Behrenfeld, M.J. (2010). Abandoning Sverdrup's critical depth hypothesis on phytoplankton blooms. *Ecology*, **91**(4), 977-989.

- Bellinger, E.G., Sigee, D.C. (2010). Freshwater algae: identification and use as bioindicators. Springer Netherlands.
- Belsky, A.J., Matzke, A., Uselman, S. (1999). Survey of livestock influences on stream and riparian ecosystems in the Western United States. *Journal of Soil and Water Conservation* 54(1), 419-431.
- Bergfur, J., Demars, B.O.L., Stutter, M.I., Langan, S.J., Friberg, N. (2012), The Tarland catchment initiative and its effect on stream water quality and macroinvertebrate indices. *Journal of Environmental Quality*, **41**(2), 314-321
- Beutel, M.W. (2006). Inhibition of ammonia release from anoxic profundal sediments in lakes using hypolimnetic oxygenation. *Ecological Engineering*, **28**(3), 271-279.
- Billington, N. (1991). A comparison of three methods of measuring phytoplankton biomass on a daily and seasonal basis. *Hydrobiologia*, **226**(1), 1-15.
- BioNinja (2016) The Nitrogen Cycle [image online]. Available from: https://ib.bioninja.com.au/options/option-c-ecology-and-conser/c6-nitrogen-andphosphorus/nitrogen-cycle.html [accessed 25/02/2020]
- Blenckner, T., Adrian, R., Livingston, D.M., Jennings, E., Weyhenmeyer, G.A., George, G., Jankoeski, T., Jarvinen, M., Nic Aonghusa, Noges, T., Straile, D., Teubner, K. (2007).
 Large-scale climatic signatures in lakes across Europe: a meta-analysis. *Global Change Biology*, 13(7), 1314-1326
- Boehrer, B., Schultze, M. (2008). Stratification of lakes. *Reviews of Geophysics*, **46**(2)
- Bond, T. A. (2012). Understanding the effects of cattle grazing in English chalk streams (Doctoral dissertation, University of Southampton).
- Bosch, N.S., Evans, M.A., Scavia, D., Allan, J.D. (2014). Interacting effects of climate change and agricultural BMPs on nutrient runoff entering Lake Erie. *Journal of Great Lakes Research*, **40**(3), 581-589.
- Bottrell, H.H. (1975). The relationship between temperature and duration of egg development in some epiphytic Cladocera and Copepoda from the River Thames, Reading, with a discussion of temperature functions. *Oceologia*, **18**(1), 63-84.
- Bouraoui, F., Grizzetti, B., Adelsköld, G., Behrendt, H., De Miguel, I., Silgram, M., Gómez, S., Granlund, K., Hoffmann, L., Kronvang, B., Kværnø, S. (2009). Basin characteristics and nutrient losses: the EUROHARP catchment network perspective. *Journal of Environmental Monitoring*, **11**(3), 515-525.
- Bowes, M.J., Jarvie, H.P., Halliday, S.J., Skeffington, R.A., Wade, A.J., Loewenthal, M., Gozzard, E., Newman, J.R., Palmer-Felgate, E.J. (2015). Characterising phosphorus and nitrate inputs to a rural river using high-frequency concentration–flow relationships. *Science of the Total Environment*, **511**, 608-620.
- Bowes, M.J., Smith, J.T., Jarvie, H.P., Neal, C., Barden, R. (2009). Changes in point and diffuse source phosphorus inputs to the River Frome (Dorset, UK) from 1966 to 2006. *Science of the Total Environment*, **407**(6), 1954-1966.
- Bowes, M.J., Gozzard, E., Johnson, A.C., Scarlett, P.M., Roberts, C., Read, D.S., Armstrong, L.K., Harman, S.A., Wickham, H.D. (2012). Spatial and temporal changes in chlorophyll-a concentrations in the River Thames basin, UK: Are phosphorus concentrations beginning to limit phytoplankton biomass? *Science of the Total Environment*, **426**, 45-55.
- Bradley, A., Larson, R.A., Runge, T. (2015). Effect of wood biochar in manure-applied sand columns on leachate quality. *Journal of environmental quality*, **44**(6), 1720-1728.
- Bragina, L., Sherlock, O., van Rossum, A.J., Jennings, E. (2017). Cattle exclusion using fencing reduces Escherichia coli (E. coli) level in stream sediment reservoirs in northeast Ireland. *Agriculture, Ecosystems & Environment*, 239, 349-358.
- Bradshaw, E.G., Anderson, N.J., Jensen, J.P., Jeppesen, E. (2002). Phosphorus dynamics in Danish lakes and the implications for diatom ecology and paleoecology. *Freshwater Biology*, 47(10), 1963-1975.

- Brentrup, J.A., Williamson, C.E., Colom-Montero, W., Eckert, W., de Eyto, E., Grossart, H.P., Huot, Y., Isles, P.D., Knoll, L.B., Leach, T.H. and McBride, C.G. (2016). The potential of high-frequency profiling to assess vertical and seasonal patterns of phytoplankton dynamics in lakes: an extension of the Plankton Ecology Group (PEG) model. *Inland Waters*, 6(4), 565-580.
- Brookes, J.D., Carey, C.C. (2011). Resilience to blooms. *Science*, **334**(6052), 46-47.
- Brown, J.R., Marshall, C.S., Smith, S.G. (1993). *Nitrate in soils and plants*. University of Missouri Extension. HU.
- Buckley, C. (2012). Implementation of the EU Nitrates Directive in the Republic of Ireland. A view from the farm. *Ecological Economics*, **79**, 29-36
- Burger, D.F., Hamilton, D.P., Pilditch, C.A. (2008). Modelling the relative importance of internal and external nutrient loads on water column nutrient concentrations and phytoplankton biomass in a shallow polymictic lake. *Ecological Modelling*, 211(3), 411-423.
- Burns, L.C., Stevens, R.J., Laughlin, R.J. (1996). Production of nitrite in soil by simultaneous nitrification and denitrification. *Soil Biology and Biochemistry*, **28**(4), 609-616.
- Cai, L.L., Zhu, G.W., Zhu, M.Y., Xu, H., Qin, B.Q. (2012). Effects of temperature and nutrients on phytoplankton biomass during bloom seasons in Taihu Lake. *Water Science* and Engineering, 5(4), 361-374.
- Cameron, K. C. and Haynes, R. J. (1986). Retention and movement of nitrogen in soils. In: *Mineral Nutrition in the Plant-Soil System* (R. J. Haynes, ed.), pp. 166-241. Academic Press, London.
- Cameron, K.C., Di, H.J., Moir, J.L. (2013), Nitrogen losses from the soil/plant system: a review, *Annuals of Applied Biology*, **162**(2), 145-173
- Canfield, D.E., Hodgson, L.M. (1983). Prediction of Secchi disc depths in Florida lakes: impact of algal biomass and organic colour. *Hydrobiologia*, **99**(1), 51-60.
- Capella, J.V., Bonastre, A., Ors, R., Peris, M. (2014). A step forward in the in-line river monitoring of nitrate by means of a wireless sensor network. *Sensors and Actuators B: Chemical*, **195**, 396-403.
- Carey, C.C., Brown, B.L., Cottingham, K.L. (2017). The cyanobacterium Gloeotrichia echinulata increases the stability and network complexity of phytoplankton communities. *Ecosphere*, **8**(7).
- Carey, C.C., Hanson, P.C., Lathrop, R.C., St. Amand, A.L. (2016). Using wavelet analyses to examine variability in phytoplankton seasonal succession and annual periodicity. *Journal of Plankton Research*, **38**(1), 27-40.
- Carney, H.J., Elser, J.J. (1990). Strength of zooplankton-phytoplankton coupling in relation to lake trophic state. In: *Large lakes* (M. M. Tilzer and C. Serruya ed.)(615-631). Springer Berlin Heidelberg.
- Carpenter, S.R., Cole, J.J., Kitchell, J.F. and Pace, M.L. (1998). Impact of dissolved organic carbon, phosphorus, and grazing on phytoplankton biomass and production in experimental lakes. *Limnology and Oceanography*, 43(1), 73-80.
- Carraro, E., Guyennon, N., Hamilton, D., Valsecchi, L., Manfredi, E.C., Viviano, G., Salerno, F., Tartari, G., Copetti, D. (2012). Coupling high-resolution measurements to a threedimensional lake model to assess the spatial and temporal dynamics of the cyanobacterium Planktothrix rubescens in a medium-sized lake. *Hydrobiologia*, 698(1), 77-95.
- Carson, A. (2010). A Temporal Spatial Analysis of Eutrophication and its Drivers in Milltown Catchment Co. Monaghan. PhD Thesis. Dundalk Institute of Technology.
- Carson, A., Jennings, E., Linnane, S., Jordan, S.N. (2015), Clearing the muddy waters: using lake sediment records to inform agricultural management. *Journal of Paleolimnology*, 53(1), 1-15

- Carter, J.E., Jokela, W.E., Bosworth, S.C. (2010), Grass forage response to broadcastor surface-banded liquid dairy manure and nitrogen fertilizer. *Agronomy Journal*, **102**(4), 1123-1131
- Cassidy, R., Jordan, P. (2011). Limitations of instantaneous water quality sampling in surfacewater catchments: Comparison with near-continuous phosphorus time-series data. *Journal of Hydrology*, **405**(1), 182-193.
- Centner, T.J., Houston, J.E., Keeler, A.G., Fuchs, C. (1999). The adoption of Best Management Practices to Reduce Agricultural Water Contaminants. *Limnologica* **29**, 366-373.
- Chapman, A.S., Foster, I.D.L., Lees, J.A., Hodgkinson, R.A., Jackson, R. H. (2001).
 Particulate phosphorus transport by sub-surface drainage from agricultural land in the UK. Environmental significance at the catchment and national scale. *Science of the Total Environment*, 266(1), 95-102.

Chen, W., Holden, N.M. (2017). Social life cycle assessment of average Irish dairy farm. *The International Journal of Life Cycle Assessment*, **22**(9), 1459-1472.

- Cherry, K.A., Shepherd, M., Withers, P.J.A., Mooney, S.J. (2008). Assessing the effectiveness of actions to mitigate nutrient loss from agriculture: A review of methods. *Science of the Total Environment*, **406**(1-2), 1-23.
- Chislock, M.F., Doster, E., Zitomer, R.A. and Wilson, A.E. (2013). Eutrophication: causes, consequences, and controls in aquatic ecosystems. *Nature Education Knowledge*, 4(4), 10.
- Christensen, J.H., Carter, T.R., Rummukainen, M., Amanatidis, G. (2007). Evaluating the performance and utility of regional climate models: the PRUDENCE project. *Climatic Change*, **81**, 1-6.
- Clews, E., Vaughan, I.P., Ormerod, S.J. (2010). Evaluating the effects of riparian restoration on a temperate river - system using standardized habitat survey. *Aquatic Conservation: Marine and Freshwater Ecosystems*, **20**(1), 96-104.
- Cloern, J.E., Jassby, A.D. (2010). Patterns and scales of phytoplankton variability in estuarine–coastal ecosystems. *Estuaries and Coasts*, **33**(2), 230-241.
- Copetti, D., Finsterle, K., Marziali, L., Stefani, F., Tartari, G., Douglas, G., Reitzel, K., Spears, B.M., Winfield, I.J., Crosa, G., d'Haese, P. (2016). Eutrophication management in surface waters using lanthanum modified bentonite: a review. *Water research*, **97**, 162-174.
- Conley, D.J., Pearl, H.W., Howarth, R.W., Boesch, D., Seitzinger, S. P., Havens, K. E., Lancelot C., Likens, G. E. (2009). Controlling eutrophication: nitrogen and phosphorus. *Science*, **323**, 1014–1015.
- Conroy, E., Turner, J.N., Rymszewicz, A., O'Sullivan, J.J., Bruen, M., Lawler, D., Lally, H., Kelly-Quinn, M. (2016). The impact of cattle access on ecological water quality in streams: Examples from agricultural catchments within Ireland. *Science of the Total Environment*, 547, 17-29.
- Cook, P.L.M., Aldridge, K.T., Lamontagne, S., Brookes, J.D. (2010). Retention of nitrogen, phosphorus and silicon in a large semi-arid riverine lake system. *Biogeochemistry*, **99**, 49–62.
- Coon, T.G., Lopez, M.M., Richerson, P.J., Powell, T.M., Goldman, C.R. (1987). Summer dynamics of the deep chlorophyll maximum in Lake Tahoe. *Journal of Plankton Research*, 9(2), 327-344.
- Cordell, D., Drangert, J.O., White, S. (2009). The story of phosphorus: Global food security and food for thought. *Global environmental change*, **19**(2), 292-305.
- Crockford, L., Jordan, P., Melland, A.R., Taylor, D. (2015). Storm-triggered, increased supply of sediment-derived phosphorus to the epilimnion in a small freshwater lake. *Inland Waters*, **5**(1), 15-26.
- CSO, (2010). Crops and Livestock Survey June 2008-Final Results. Central Statistics Office, Skehard Road, Cork, Ireland, 9.
- CSO, (2015). *Farm Structures Survey 2013*. Available online: www.cso.ie/en/ releasesandpublications/ep/p-fss/farmstructuresurvey2013/ (accessed 24 August 2016).
- CSO, (2018). *Livestock Survey December 2017*. Available online: <u>https://www.cso.ie/en/releasesandpublications/er/lsd/livestocksurveydecember2017/</u> (accessed 12 April 2018).
- Cunha Pereira, H., Allott, N., Coxon, C., Naughton, O., Johnston, P., Gill, L. (2011).
 Phytoplankton of turloughs (seasonal karstic Irish lakes). *Journal of Plankton Research*, 33(3), 385-403.
- Cullen, J. J. (2015). Subsurface chlorophyll maximum layers: Enduring enigma or mystery solved? *Annual Review of Marine Science*, **7**, 207–239.
- Dale, A.J., Ferris, C.P., Frost, J.P., Mayne, C.S., Kilpatrick, D.J. (2012), The effect of applying cattle slurry using the trailing-shoe technique on dairy cow and sward performance in a rotational grazing system, *Grass and Forage Science*, **68**(1), 138-150
- Dalgaard, T., Hutchings, N., Dragosits, U., Olesen, J. E., Kjeldsen, C., Drouet, J. L., Cellier, P. (2011). Effects of farm heterogeneity and methods for upscaling on modelled nitrogen losses in agricultural landscapes. *Environmental Pollution*, **159**(11), 3183-3192.
- Dalton, C., Jennings, E., O'Dwyer, B., Taylor, D. (2016). Integrating observed, inferred and simulated data to illuminate environmental change: a limnological case study.
 In *Biology and Environment: Proceedings of the Royal Irish Academy* (Vol. 116, No. 3, pp. 279-294). Royal Irish Academy.
- Daly, K., Mills, P., Coulter, B., McGarrigle, M. (2002). Modeling phosphorus concentrations in Irish rivers using land use, soil type, and soil phosphorus data. *Journal of Environmental Quality*, **31**(2), 590-599.
- Davis, T.W., Bullerjahn, G.S., Tuttle, T., McKay, R.M., Watson, S.B. (2015). Effects of increasing nitrogen and phosphorus concentrations on phytoplankton community growth and toxicity during Planktothrix blooms in Sandusky Bay, Lake Erie. *Environmental Science & Technology*, **49**, 7197–7207
- Davis, T.W., Koch, F., Marcoval, M.A., Wilhelm, S.W., Gobler, C.J. (2012). Mesozooplankton and microzooplankton grazing during cyanobacterial blooms in the western basin of Lake Erie. *Harmful Algae*, 15, 26-35.
- Deakin, J., Flynn, R., Archbold, M., Daly, D., O'Brien, R., Orr, A., Misstear, B. (2016).
 January. Understanding pathways transferring nutrients to streams: review of a major Irish study and its implications for determining water quality management strategies.
 In *Biology and Environment: Proceedings of the Royal Irish Academy*, **116**(3), 233-243.
 Royal Irish Academy.
- Deasy, C., Quinton, J.N., Silgram, M.S., Bailey, A.P., Jackson, B., Stevens, C.J. (2010). Contributing understanding of mitigation options for phosphorus and sediment to a review of the efficacy of contemporary agricultural stewardship measures. *Agricultural Systems* 103,105–109.
- Demal, L. (1983). An intensive water quality survey of stream cattle access sites. Stratford/Avon River Environmental Management Project Technical Report.
- Deng, J., Qin, B., Paerl, H.W., Zhang, Y., Wu, P., Ma, J., Chen, Y. (2014). Effects of nutrients, temperature and their interactions on spring phytoplankton community succession in Lake Taihu, China. *PloS One*, 9(12), 113960.
- Department of Agriculture, Food and Rural Development, (2000). *Rural Development Plan* for the Republic of Ireland. Government Publications, Dublin.
- Department of Agriculture, Food and the Marine, (2015). GLAS (Green Low Carbon Agri-Environment Scheme), Ireland

- Department of Agriculture, Food and the Marine, (2016). *Fact Sheet on Irish Agriculture February 2016*, Department of Agriculture, Food and the Marine, Dublin
- Department of Agriculture, Food and the Marine, (2017). Protecting and Enhancing Water for Sustainable Agriculture Water-related measures for the agri-food, forestry and marine sectors – March 2017, Department of Agriculture, Food and the Marine, Dublin
- Department of Agriculture, Environment and Rural Affairs, (2018). *Agricultural Census in Northern Ireland 2017 – January 2018*. Department of Agriculture, Environment and Rural Affairs.
- Department of Housing, Planning and Local Government, (2018). *River Basin Management Plan 2018 – 2021 for Ireland – April 2018*. Department of Housing Planning and Local Government, Dublin.
- De Senerpont Domis, L.N., Elser, J.J., Gsell, A.S., Huszar, V.L., Ibelings, B.W., Jeppesen, E., Kosten, S., Mooij, W.M., Roland, F., Sommer, U., Van Donk, E. (2013). Plankton dynamics under different climatic conditions in space and time. *Freshwater Biology*, 58(3), 463-482.
- De Eyto, E., Jennings, E., Ryder, E., Sparber, K., Dillane, M., Dalton, C., Poole, R. (2016). Response of a humic lake ecosystem to an extreme precipitation event: physical, chemical, and biological implications. *Inland Waters*, **6**(4), 483-498.
- Dils, R., Leaf, S., Robinson, R., Sweet, N. (1999). Phosphorus in the environment-why should recovery be a policy issue? Environment Agency. In *2nd international conference on the recovery of phosphorus from sewage and animal wastes*, 12-13.
- Directive, N., 1991. Council Directive 91/676/EEC of 12 December 1991 concerning the protection of waters against pollution caused by nitrates from agricultural sources. *Official Journal*, **375**(31), 12.
- Directive, 2001. 2455/2001/EC of the European Parliament and of the Council of 20 November 2001 establishing the list of priority substances in the field of water policy and amending Directive 2000/60/EC. *Official Journal of the European Communities*, **15**, 1-5.
- Dodds, W.K., Bouska, W.W., Eitzmann, J.L., Pilger, T.J., Pitts, K.L., Riley, A.J., Schloesser, J.T. and Thornbrugh, D.J. (2009). Eutrophication of US freshwaters: analysis of potential economic damages.
- Dorioz, J.M., Wang, D., Poulenard, J., Trevisan, D. (2006). The effect of grass buffer strips on phosphorus dynamics – critical review and synthesis as a basis for application in agricultural landscapes in France. *Agriculture, Ecosystems and Environment* **117**, 4-21.
- Downing, J.A., Prairie, Y.T., Cole, J.J., Duarte, C.M., Tranvik, L.J., Striegl, R.G., McDowell, W.H., Kortelainen, P., Caraco, N.F., Melack, J.M., Middelburg, J.J. (2006). The global abundance and size distribution of lakes, ponds, and impoundments. *Limnology and Oceanography*, **51**(5), 2388-2397.
- Dupas, R., Delmas, M., Dorioz, J.M., Garnier, J., Moatar, F., Gascuel-Odoux, C. (2015). Assessing the impact of agricultural pressures on N and P loads and eutrophication risk. *Ecological Indicators*, 48, 396-407.
- Earle, E., Blacklocke, S. (2008). Master plan for water framework directive activities in Ireland leading to River Basin Management Plans. *Desalination* **226**, 134-142.
- Edwards, M., Richardson, A.J. (2004). Impact of climate change on marine pelagic phenology and trophic mismatch. *Nature*, **430**(7002), 881.
- Edwards, K.F., Thomas, M.K., Klausmeier, C.A., Litchman, E. (2016). Phytoplankton growth and the interaction of light and temperature: A synthesis at the species and community level. *Limnology and Oceanography*, **61**(4), 1232-1244.
- EEA, (2016). Freshwater quality. [Online] Available at: https://www.eea.europa.eu/soer-
- 2015/europe/freshwater [Accessed 10/08/2017].
- EEA, (2010). *Ecosystem accounting and the cost of biodiversity losses*. The case of coastal Mediterranean wetlands. EEA Technical report No 3/2010, Copenhagen. 91.

- Egemose, S., de Vicente, I., Reitzel, K., Flindt, M.R., Andersen, F.O., Lauridsen, T.L., Søndergaard, M., Jeppesen, E., Jensen, H.S. (2011). Changed cycling of P, N, Si, and DOC in Danish Lake Nordborg after aluminum treatment. *Canadian Journal of Fisheries and Aquatic Sciences*, **68**, 842–856.
- Elliott, J.A., Thackeray, S.J., Huntingford, C., Jones R.G. (2005). Combining a regional climate model with a phytoplankton community model to predict future changes in phytoplankton in lakes. *Freshwater Biology*, **50**: 1404–1411.
- Emerson, H.J., Gillmor, D.A. (1999). The Rural Protection Scheme of the Republic of Ireland. *Land Use and Policy*, **16**, 235-245.
- EPA (2016). Corine datasets. [Online] Available at: 522 http://www.epa.ie/soilandbiodiverrsity/soils/land/corine/datasets/#VvhN630rLIU. [Accessed 03 February 2016].
- EPA (2016). Drinking Water Report for Public Water Supplies 2015. EPA, Ireland.
- EPA (2012). Water quality in Ireland 2010-2012. EPA, Ireland.
- EPA (2012). Environmental Analysis of Food Harvest 2020: Response of the Environmental Protection Agency. EPA, Ireland.
- EPA (2014). Inland and Marine Waters. EPA, Ireland.
- EPA (2015). Water quality in Ireland-water quality of lakes. EPA, Ireland.
- European Standards 2006. *Water quality Guidance standard on the enumeration of phytoplankton using inverted microscopy (Utermohl technique) (EVS-EN 15204: 2006).* European Committee for Standardization, pp.40.
- Faithfull, C., Huss, M., Vrede, T., Karlsson, J., Bergström, A.K. (2011). Transfer of bacterial production based on labile carbon to higher trophic levels in an oligotrophic pelagic system. *Canadian Journal of Fisheries and Aquatic Sciences*, 69(1), 85-93.
- Fealy, R. M., Buckley, C., Mechan, S., Melland, A., Mellander, P. E., Shortle, G., Wall, D., Jordan, P. (2010). The Irish Agricultural Catchments Programme: catchment selection using spatial multi - criteria decision analysis. *Soil Use and Management*, 26(3), 225-236.
- Fee, E. J. (1976). The vertical and seasonal distribution of chlorophyll in lakes of the Experimental Lakes Area, northwestern Ontario: Implications for primary production estimates. *Limnology and Oceanography*, **21**, 767–783.
- Fee, E.J., Hecky, R.E., Kasian, S.E.M., Cruikshank, D.R. (1996). Effects of lake size, water clarity, and climatic variability on mixing depths in Canadian Shield lakes. *Limnology* and Oceanography, 41(5), 912-920.
- Feehan, J., Gillmor, D.A., Culleton, N. (2005). Effects of agri-environment scheme on farmland biodiversity in Ireland. Agriculture, Ecosystems and Environment, 107, 275-286.
- Felip, M., Catalan, J. (2000). The relationship between phytoplankton biovolume and chlorophyll in a deep oligotrophic lake: decoupling in their spatial and temporal maxima. *Journal of Plankton Research*, **22**(1), 91-106.
- Finn, J.A., Ó hUallacháin, D. (2012). A Review of Evidence on the Environmental Impact of Irelands Rural Environment Protection Scheme (REPS). *Biology and Environment: Proceedings of the Royal Irish Academy*, **112**(B), 1–24.
- Firbank, L., Bradbury, R.B., McCracken, D.I., Stoate, C. (2012). Delivering multiple ecosystem services from enclosed farmland in the UK. Agriculture, Ecosystems & Environment, 166, 65-75
- Foley, B., Jones, I.D., Maberly, S.C., Rippey, B. (2012). Long-term changes in oxygen depletion in a small temperate lake: effects of climate change and eutrophication. *Freshwater Biology*, 57, 278–289
- Follett, R.F. (2008). Transformation and transport processes of nitrogen in agricultural systems. *Nitrogen in the Environment: Sources, Problems and Management. Elsevier, The Nederlands*, 19-50.

- Follett, J.R., Follett, R.F., Herz, W.C. (2010). Environmental and human impacts of reactive nitrogen. In: *Advances in Nitrogen Management for Water Quality, ed. JA Delgado and RF Follett*, 1-37.
- Foster, G.R., Young, R.A., Neibling, W.H. (1985). Sediment composition for nonpoint source pollution analyses. *Transactions of the ASAE-American Society of Agricultural Engineers*, 28.
- Free, G., Little, R., Tierney, D., Donnelly, K., Caroni, R. (2006). A reference based typology and ecological assessment system for Irish Lakes. Preliminary Investigations. Final Report. EPA, Ireland.
- Free, G., Tierney, D., Little, R., Kelly, F.L., Kennedy, B., Plant, C., Trodd, W., Wynne, C., Caroni, R., Byrne, C. (2016). Lake ecological assessment metrics in Ireland: relationships with phosphorus and typology parameters and the implications for setting nutrient standards. *Biology and Environment: Proceedings of the Royal Irish Academy*, **116**(3), 191-204).
- Galeone, D.G. (2000). Preliminary effects of streambank fencing of pasture land on the quality of surface water in a small watershed in Lancaster County, Pennsylvania. US Department of the Interior, US Geological Survey.
- Galeone, D.G., Low, D.J., Brightbill, R.A. (2006). Effects of streambank fencing of nearstream pasture land on a small watershed in Lancaster County, Pennsylvania, 1993-2001.
- Gardner, W.S., Cavaletto, J.F., Bootsma, H.A., Lavrentyev, P.J., Troncone, F. (1998). Nitrogen cycling rates and light effects in tropical Lake Maracaibo, Venezuela. *Limnology and Oceanography*, **43**(8), 1814-1825.
- Geider, R.J., Moore, C.M., Ross, O.N. (2009). The role of cost–benefit analysis in models of phytoplankton growth and acclimation. *Plant Ecology & Diversity*, **2**(2), 165-178.
- Ger, K.A., Urrutia-Cordero, P., Frost, P.C., Hansson, L.A., Sarnelle, O., Wilson, A.E., Lürling, M. (2016). The interaction between cyanobacteria and zooplankton in a more eutrophic world. *Harmful algae*, 54, 128-144.
- Gleeson, E., McGrath, R., Treanor, M. (2013). *Ireland's Climate: The Road Ahead* Met Éireann, Dublin.
- Gran, H.H., Braarud, T. (1935). A quantitative study of the phytoplankton in the Bay of Fundy and the Gulf of Maine (including observations on hydrography, chemistry and turbidity). *Journal of the Biological Board of Canada*, **1**(5), 279-467.
- Grantz, E.M., Haggard, B.E., Scott, J.T. (2014). Stoichiometric imbalance in rates of nitrogen and phosphorus retention, storage, and recycling can perpetuate nitrogen deficiency in highly-productive reservoirs. *Limnology and Oceanography*, **59**, 2203–2216.
- Gregor, J., Maršálek, B. (2004). Freshwater phytoplankton quantification by chlorophyll a: a comparative study of in vitro, in vivo and in situ methods. *Water Research*, **38**(3), 517-522.
- Greene, S., Taylor, D., McElarney, Y. R., Foy, R. H., Jordan, P. (2011). An evaluation of catchment-scale phosphorus mitigation using load apportionment modelling. *Science of the Total Environment*, **409**(11), 2211-2221.
- Gutierrez, M.F., Devercelli, M., Brucet, S., Lauridsen, T.L., Søndergaard, M., Jeppesen, E., (2016). Is recovery of large-bodied zooplankton after nutrient loading reduction hampered by climate warming? A long-term study of shallow hypertrophic Lake Søbygaard, Denmark. *Water*, 8(8), 341.
- Haddaway, N.R., Brown, C., Eggers, S., Josefsson, J., Kronvang, B., Randall, N., Uusi-Kämppä, J. (2016). The multifunctional roles of vegetated strips around and within agricultural fields. A systematic map protocol. *Environmental Evidence*, 5(1), 18.
- Hamilton, D. P., O'Brien, K. R., Burford, M. A., Brookes, J. D. (2010). Vertical distributions of chlorophyll in deep, warm monomictic lakes. *Aquatic Sciences-Research Across Boundaries*, **72**, 295-307.

- Harding, J.P. and Smith, W.A., 1974. A Key to the British Freshwater Biological Association Scientific Publication.
- Harris, T.D., Smith, V.H., Graham, J.L., Van de Waal, D.B., Tedesco, L.P., Clercin, N. (2016). Combined effects of nitrogen to phosphorus and nitrate to ammonia ratios on cyanobacterial metabolite concentrations in eutrophic Midwestern USA reservoirs. *Inland Waters*, 6(2), 199-210.
- Hatfield, J.L., Follett, R.F. (Eds.). (2008). *Nitrogen in the Environment*. Access Online via Elsevier.
- Havens, K., Paerl, H., Phlips, E., Zhu, M., Beaver, J., Srifa, A. (2016). Extreme weather events and climate variability provide a lens to how shallow lakes may respond to climate change. *Water*, 8(6), 229.
- Haygarth, P.M., Jarvis, S.C. (1999). Transfer of phosphorus from agricultural soils. In: *Advances Agronomy* **66**, 195-249.
- Haynes, R.J., Williams, P.H. (1993). Nutrient cycling and soil fertility in the grazed pasture ecosystem. In: *Advances in Agronomy*, **49**, 119-199.
- Hayes, N.M., Vanni, M.J., Horgan, M.J., Renwick, W.H. (2015). Climate and land use interactively affect lake phytoplankton nutrient limitation status. *Ecology*, 96(2), 392-402.
- Heaney, S. I., Talling, J. F. (1980). Dynamic aspects of dinoflagellate distribution patterns in a small productive lake. *Journal of Ecology* **68**, 75-94.
- Heathwaite, H.L., Johnes, P.J., Peters, N.E. (1996). Trends in Nutrients and water quality. *Hydrology Processes*, **10**, 263-293
- Heathwaite, A.L., Griffiths, P., Parkinson, R.J. (1998). Nitrogen and phosphorus in runoff from grassland with buffer strips following application of fertilizers and manures. *Soil Use and Management*, 14(3), 142-148.
- Heckrath, G., Brookes, P. C., Poulton, P. R., Goulding, K. W. T. (2008). Phosphorus leaching from soils containing different phosphorus concentrations in the Broadbalk experiment. *Journal of Environmental Quality* 24(5), 904-910.
- Helsel, D.R., Hirsch, R.M. (1992), Statistical methods in water resources (Vol. 49), Elsevier
- Hepperle, D. (2005). Opticount- Software for the enumeration of plankton. at: http://science.domix.de
- Hiriart-Baer, V.P., Smith, R.E. (2005). The effect of ultraviolet radiation on freshwater planktonic primary production: The role of recovery and mixing processes. *Limnology and oceanography*, **50**(5), 1352-1361.
- Hjerppe, T., Seppälä, E., Väisänen, S. and Marttunen, M. (2017). Monetary assessment of the recreational benefits of improved water quality–description of a new model and a case study. *Journal of Environmental Planning and Management*, **60**(11), 1944-1966.
- Hoffmann, C.C., Kjaergaard, C., Uusi-Kamppa, J., Hansen, H.C.B., Krovang, B. (2009), Phosphorus retention in riparian buffers; Review of their efficiency, *Journal of Environmental Quality*, 38(5), 1942-1955
- Hooda, P.S., Edwards, A.C., Anderson, H.A., Miller, A., (2000). A review of water quality concerns in livestock farming areas. *The Science of the Total Environment* 250, 143-167.
- House of the Oireachtas (2011), Dáil Éireann Debate Vol. 743 No. 2. [Online] Available at: http://oireachtasdebates.oireachtas.ie/debates%20authoring/debateswebpack.nsf/takes/ dail2011101200109?opendocument. [Accessed 20 February 16]
- Huber-Pestalozzi, G. (1983). Das phytoplankton des Süβwasser. Die Binningewässer. Schweizerbartsche Verlagsbuchhandlung.
- Huisman, J., Sharples, J., Stroom, J.M., Visser, P.M., Kardinaal, W.E.A., Verspagen, J.M., Sommeijer, B. (2004). Changes in turbulent mixing shift competition for light between phytoplankton species. *Ecology*, 85(11), 2960-2970.

- Huot, Y., Babin, M. (2010). Overview of fluorescence protocols: theory, basic concepts, and practice. In *Chlorophyll a fluorescence in aquatic sciences: Methods and applications* (pp.31-74). Springer Netherlands.
- Huser, B.J., Futter, M., Lee, J.T., Perniel, M. (2016). In-lake measures for phosphorus control: the most feasible and cost-effective solution for long-term management of water quality in urban lakes. *Water research*, **97**, 142-152.
- Hutchinson, E. (1957). A treatise on limnology vol 1. Wiley, New York, 1015
- Hutchinson, G.E., Löffler, H. (1956). The thermal classification of lakes. *Proceedings of the National Academy of Sciences*, **42**(2), 84-86.
- Ibelings, B.W., Kroon, B., Mur, L.R. (1994). Acclimation of photosystem II in a cyanobacterium and a eukaryotic green alga to high and fluctuating photosynthetic photon flux densities, simulating light regimes induced by mixing in lakes. *New Phytologist*, **128**(3), 407-424.
- Ide, J.I., Haga, H., Chiwa, M., Otsuki, K. (2008). Effects of antecedent rain history on particulate phosphorus loss from a small forested watershed of Japanese cypress (Chamaecyparis obtusa). *Journal of Hydrology*, **352**(3), 322-335.
- Irwin, A.J., Finkel, Z.V., Schofield, O.M., Falkowski, P.G. (2006). Scaling-up from nutrient physiology to the size-structure of phytoplankton communities. *Journal of plankton research*, **28**(5), 459-471.
- Istvánovics, V. (2009). Eutrophication of Lakes and Reservoirs. *Encyclopedia of Inland Waters*, 157-165.
- Jackson, L.J., Lauridsen, T.L., Søndergaard, M., Jeppesen, E.A., (2007). comparison of shallow Danish and Canadian lakes and implications for climate change. *Freshwater Biology*, 52, 1782–1792.
- Jackson, M.L. (2016). *Phytoplankton and nutrient dynamics with a focus on nitrogen form in the Anacostia River, in Washington, D.C. and Wst Lake, in Hangzhou, China.* PhD Thesis.
- James, R. T., Havens, K., Zhu, G. W., Qin, B. Q. (2009). Comparative analysis of nutrients, chlorophyll and transparency in two large shallow lakes (Lake Taihu, P. R. China and Lake Okeechobee, USA). *Hydrobiologia*, 627(1), 211-231.
- James, R.T., Havens, K., Zhu, G., Qin, B. (2009). Comparative analysis of nutrients, chlorophyll and transparency in two large shallow lakes (Lake Taihu, PR China and Lake Okeechobee, USA). *Hydrobiologia*, **627**(1), 211-231.
- Jarvie, H. P., Sharpley, A. N., Withers, P. J., Scott, J. T., Haggard, B. E., Neal, C. (2013). Phosphorus mitigation to control river eutrophication: Murky waters, inconvenient truths, and "postnormal" science. *Journal of Environmental Quality* 42(2), 295-304.
- Jellison, R., Romero, J., Melack, J.M. (1998). The onset of meromixis during restoration of Mono Lake, California: Unintended consequences of reducing water diversions. *Limnology and Oceanography*, **43**(4), 706-711.
- Jennings, E., Mills, P., Jordan, P., Jenses, J.P., Søndergaard, M., Barr, A., Glasgow, G., Irvine, K. (2003). Eutrophication from Agricultural Sources, Seasonal Patterns & Effects of Phosphorus. *Environmental Protection Agency*, Ireland.
- Jennings, E., Allott, N., Lenihan, D., Quirke, B., Taylor, D., Twomey, H. (2013). Drivers of long-term trends and seasonal changes in total phosphorus loads to a mesotrophic lake in the west of Ireland. *Marine and Freshwater Research*, 64(5), 413-422.
- Jennings, E., Jones, S., Arvola, L., Staehr, P.A., Gaiser, E., Jones, I.D., Weathers, K.C., Weyhenmeyer, G.A., Chiu, C.Y., De Eyto, E. (2012). Effects of weather-related episodic events in lakes: an analysis based on high-frequency data. *Freshwater Biology*, 57(3), 589-601.
- Jennings, E., Allott, N., Pierson, D.C., Schneiderman, E.M., Lenihan, D., Samuelsson, P., Taylor, D. (2009). Impacts of climate change on phosphorus loading from a grassland catchment: Implications for future management. *Water research*, 43(17), 4316-4326.

- Jensen, J.P., Jeppesen, E., Olrik, K., Kristensen, P. (1994). Impact of nutrients and physical factors on the shift from cyanobacterial to chlorophyte dominance in shallow Danish lakes. *Canadian Journal of Fisheries and Aquatic Sciences*, **51**(8), 1692-1699.
- Jeppesen, E., Meerhoff, M., Holmgren, K., González-Bergonzoni, I., Teixeira-de Mello, F., Declerck, S.A., De Meester, L., Søndergaard, M., Lauridsen, T.L., Bjerring, R., Conde-Porcuna, J.M. (2010). Impacts of climate warming on lake fish community structure and potential effects on ecosystem function. *Hydrobiologia*, 646(1), 73-90.
- Jeppesen, E., Mehner, T., Winfield, I.J., Kangur, K., Sarvala, J., Gerdeaux, D., Rask, M., Malmquist, H.J., Holmgren, K., Volta, P., Romo, S. (2012). Impacts of climate warming on the long-term dynamics of key fish species in 24 European lakes. *Hydrobiologia*, **694**(1), 1-39.
- Jeppesen, E., Kronvang, B., Meerhoff, M., Søndergaard, M., Hansen, K.M., Andersen, H.E., Lauridsen, T.L., Liboriussen, L., Beklioglu, M., Özen, A., Olesen, J.E. (2009). Climate change effects on runoff, catchment phosphorus loading and lake ecological state, and potential adaptations. *Journal of Environmental Quality*, **38**(5), 1930-1941.
- Jeppesen, E., Meerhoff, M., Davidson, T.A., Trolle, D., Sondergaard, M., Lauridsen, T.L., Beklioglu, M., Brucet Balmaña, S., Volta, P., González-Bergonzoni, I., Nielsen, A. (2014). Climate change impacts on lakes: an integrated ecological perspective based on a multi-faceted approach, with special focus on shallow lakes. *Journal of Limnology*, 73, 84-107.
- Jeppesen, E., Sondergaard, M., Jensen, J.P., Lauridsen, T.L., Liboriussen, L., Hansen, R.B. (2005). Lake responses to reduced nutrient loading – an analysis of contemporary longterm data from 35 case studies. *Freshwater Biology*, **50**, 1747–1771.
- Jeppesen, E., Kronvang, B., Olesen, J. E., Audet, J., Søndergaard, M., Hoffmann, C.C., Andersen, H.E., Lauridsen, T.L., Liboriussen, L., Larsen, S.E., Beklioglu, M., Meerhoff, M., O[°] zen, A., O[°], zkan, K. (2011). Climate change effects on nitrogen loading from cultivated catchments in Europe: implications for nitrogen retention, ecological state of lakes and adaptation. *Hydrobiologia*, 663, 1–21.
- Jeppesen, E., Jensen, J.P., Søndergaard, M., Lauridsen, T., Landkildehus, F. (2000). Trophic structure, species richness and biodiversity in Danish lakes: changes along a phosphorus gradient. *Freshwater biology*, 45(2), 201-218.
- Jeppesen, E., Jensen, J.P., Søndergaard, M., Lauridsen, T., Pedersen, L.J., Jensen, L. (1997). Top-down control in freshwater lakes: the role of nutrient state, submerged macrophytes and water depth. In *Shallow Lakes* '95, 151-164. Springer, Dordrecht.
- Jiao, Y., Whalen, J.K., Hendershot, W.H. (2006). No-tillage and manure applications increase aggregation and improve nutrient retention in a sandy-loam soil. *Geoderma*, **134**(1), 24-33.
- Johansson, K.S., Vrede, T., Lebret, K., Johnson, R.K. (2013). Zooplankton feeding on the nuisance flagellate Gonyostomum semen. *PLoS One*, **8**(5), 62557.
- Jones, J., Brett, M.T. (2014). *Lake nutrients, eutrophication, and climate change. Global Environmental Change.* Netherlands, Springer, 273-279.
- Jordan, P., Rippey, B. (2003). Lake sedimentary evidence of phosphorus, iron and manganese mobilisation from intensively fertilised soils. *Water Research*, **37**(6), 1426-1432
- Jordan, C., Smith, R.V. (2005). Methods to predict the agricultural contribution to catchment nitrate loads: designation of nitrate vulnerable zones in Northern Ireland. *Journal of Hydrology*, **304**(1), 316-329.
- Jordan, P., Menary, W., Daly, K., Keily, G., Morgan, G., Byrne, P., Moles, R. (2005). Patterns processes of phosphorus transfer from Irish grassland soils to rivers-integration of laboratory and catchment studies. *Journal of Hydrology* **304**, 20–34.
- Jordan, P., Arnscheidt, J., McGrogan, H.J., McCormick, S. (2007) Characterising phosphorus transfers in rural transfers using a continuous bank-side analyser. *Hydrology and Earth System Science Discussions* **11** (1), 372-381.

- Jordan, P., Haygarth, P. M., Shortle, G., Harris, R. C. (2012). Catchment science and policy for agriculture and water quality. *Environmental Science & Policy* 24, 1-3.
- Jørgensen, S. E. (2009). *Applications in ecological engineering*. Netherlands: Elsevier B.V Timing of riverine export of nitrate and phosphorus from agricultural watersheds in Illinois: Implications for reducing nutrient loading to the Mississippi River.
- Kamykowski, D., Yamazaki, H. (1997). A study of metabolism influenced orientation in the diel vertical migration of marine dinoflagellates. *Limnology and Oceanography*, 42, 1189-1202.
- Kay, A.L., Davies, H.N., Bell, V.A., Jones, R.G. (2009), Comparison of uncertainty sources for climate change impacts: flood frequency in England, *Climatic Change*, 92(1-2), 41-63
- Kc, M.L. (2007). GIS-based statistical, geostatistical, and stochastic analyses of nitrate contamination in the Cimarron Terrace Aquifer in Oklahoma. ProQuest.
- Keane, T. (1986). Climate, weather and Irish agriculture. Meteorological Service.
- Kelly, M. (2009). Introduction. *Biology and Environment: Proceedings of the Royal Irish Academy*, **109B**, 127-129
- Kiely, G., Morgan, G., Moles, R., Byrne, P., Jordan, P., Daly, K., Doody, D., Tunney, H., Kurz, I., Bourke, D., O'Reilly, C., Ryan, D., Holden, N., Jennings, E., Irvine, K., Carton, O. (2007). Eutrophication from Agriculture Sources: Pathways for Nutrient Loss with Emphasis on Phosphorus.
- King, L., Barker, P., Jones, R.I. (2000). Epilithic algal communities and their relationship to environmental variables in lakes of the English Lake District. *Freshwater Biology*, **45**(4), 425-442.Kirillin, G. (2010). Modeling the impact of global warming on water temperature and seasonal mixing regimes in small temperate lakes. *Boreal Environ. Res*, **15**(2), 279-293.
- Kirillin, G., Shatwell, T. (2016). Generalized scaling of seasonal thermal stratification in lakes. *Earth-Science Reviews*, **161**, 179-190.
- Kleeberg, A., Kozerski, H.P. (1997). Phosphorus release in lake Groâer Müggelsee and its implications for lake restoration. *Hydrobiologia*, **342–343**, 9–26.
- Kleinman, P.J.A., Sharpley, A.N., Buda, A.R., McDowell, R.W., Allen, A.L. (2011). Soil controls of phosphorus in runoff: Management barriers and opportunities. *Canadian Journal of Soil Science*. **91**, 329–338.
- Kosten, S., Huszar, V.L.M., Bécares, E., Costa, L.S., Van Donk, E., Hansson, L.A., Jeppesen, E., Kruk, C., Lacerot, G., Mazzeo, N. (2012). Warmer climate boosts cyanobacterial dominance in shallow lakes. *Global. Change. Biology*, 18, 118–126.
- Kramer, S.B., Reganold, J.P., Glover, J.D., Bohannan, B.J., Mooney, H.A. (2006). Reduced nitrate leaching and enhanced denitrifier activity and efficiency in organically fertilized soils. *Proceedings of the National Academy of Sciences of the United States of America*, 103(12), 4522-4527
- Krovang, B., Bechmann, M., Lundekvam, H., Behrendt, H., Rubaek, G., Schoumans, O., Syversen, N., Anderson, H., Hoffman, C. (2005). Phosphorus losses from agricultural areas in river basins: effect and uncertainties of targeted mitigation measures. *Journal of Environmental Quality* 34, 2129-2144.
- Kumar, S., Meena, R.S., Lal, R., Yadav, G.S., Mitran, T., Meena, B.L., Dotaniya, M.L. and Ayman, E.S. (2018). Role of legumes in soil carbon sequestration. In *Legumes for Soil Health and Sustainable Management* (109-138). Springer, Singapore.
- Kurz, M.J., Eichhorn, G., Accomazzi, A., Grant, C., Demleitner, M., Henneken, E., Murray, S.S. (2005). The effect of use and access on citations. *Information Processing & Management*, **41**(6), 1395-1402
- Leach, T.H., Beisner, B.E., Carey, C.C., Pernica, P., Rose, K.C., Huot, Y., Brentrup, J.A., Domaizon, I., Grossart, H.P., Ibelings, B.W., Jacquet, S. (2018). Patterns and drivers of

deep chlorophyll maxima structure in 100 lakes: The relative importance of light and thermal stratification. *Limnology and Oceanography*, **63**(2), 628-646.

- Lebret, K., Fernández, M., Hagman, C.H., Rengefors, K., Hansson, L.A. (2012). Grazing resistance allows bloom formation and may explain invasion success of Gonyostomum semen. *Limnology and Oceanography*, **57**(3), 727-734.
- Legrand, A., Schütz, K.E., Tucker, C.B. (2011), Using water to cool cattle: Behavioural and physiological changes associated with voluntary use of cow showers, *Journal of Dairy Science*, **94**(7), 3376-3386
- Lewandowska, A.M., Striebel, M., Feudel, U., Hillebrand, H., Sommer, U. (2015). The importance of phytoplankton trait variability in spring bloom formation. *ICES Journal of Marine Science*, **72**(6), 1908-1915.
- Lewis, C., Rafique, R., Foley, N., Leahy, P., Morgan, G., Albertson, J., Kumar, S., Kiely, G. (2013). Seasonal exports of phosphorus from intensively fertilised nested grassland catchments. *Journal of Environmental Sciences*, 25(9), 1847-1857.
- Lewis Jr, W.M. (1983). A revised classification of lakes based on mixing. *Canadian Journal* of Fisheries and Aquatic Sciences, **40**(10), 1779-1787.
- Line, D.E. (2003), Changes in a stream's physical and biological conditions following livestock exclusion, *Transactions of the ASAE*, **46**(2), 287-293
- Linnane, S., Jordan, S., McCarthy, V., Jennings, E., Carson, A., Sweeny, N., Wynne, C., McDonald, B. (2011). National Source Protection Pilot Project at the Churchill and Oram Group Water Scheme Co., Monaghan Final Report. *Centre for Freshwater and Environmental Studies, Dundalk Institute of Technology*

Litchman, E., Klausmeier, C.A. (2008). Trait-based community ecology of phytoplankton. *Annual review of ecology, evolution, and systematics*, **39**, 615-639.

- Livingstone, D.M., Kernan, M. (2009). Regional coherence and geographical variability in the surface water temperatures of Scottish Highland lochs. *Advances in Limnology*, 367-378.
- Lloyd, C.E.M., Freer, J.E., Johnes, P.J., Collins, A.L. (2016). Using hysteresis analysis of high-resolution water quality monitoring data, including uncertainty, to infer controls on nutrient and sediment transfer in catchments. *Science of The Total Environment*, 543, 388-404.
- Lucas, L.V., Koseff, J.R., Monismith, S.G., Cloern, J.E., Thompson, J.K. (1999). Processes governing phytoplankton blooms in estuaries. II: The role of horizontal transport. *Marine Ecology Progress Series*, **187**, 17-30.
- Lund, J.W.G. (1967). Eutrophication. Nature, 214, 557-558
- Luhtala, H., Tolvanen, H. (2013). Optimizing the use of Secchi depth as a proxy for euphotic depth in coastal waters: An empirical study from the Baltic Sea. *ISPRS International Journal of Geo-Information*, 2(4), 1153-1168.
- Lürling, M., Mackay, E., Reitzel, K., Spears, B.M. (2016). Editorial–A critical perspective on geo-engineering for eutrophication management in lakes.
- Lynch, M., Shapiro, J. (1981). Predation, enrichment, and phytoplankton community structure 1. *Limnology and oceanography*, **26**(1), 86-102.
- Maberly, S.C. (1996). Diel, episodic and seasonal changes in pH and concentrations of inorganic carbon in a productive lake. *Freshwater Biology*, **35**(3), 579-598.
- Macintosh, K. A., Jordan, P., Cassidy, R., Arnscheidt, J., Ward, C. (2011). Low flow water quality in rivers; septic tank systems and high-resolution phosphorus signals. *Science of the Total Environment* **412**, 58-65.
- Mackay, E.B., Maberly, S.C., Pan, G., Reitzel, K., Bruere, A., Corker, N., Douglas, G., Egemose, S., Hamilton, D., Hatton-Ellis, T., Huser, B., Li, W., Meis, S., Moss, B., Lürling, M., Phillips, G., Yasseri, S., Spears, B.M. (2014). Geoengineering in lakes: welcome attraction or fatal distraction? *Inland Waters*, 4, 349-356.

- Margalef, R. (1969). A manual on the methods for measuring primary production in aquatic environment including a chapter on bacteria. Blackwell Scientific Publications.
- Markensten, H., Moore, K., Persson, I. (2010). Simulated lake phytoplankton composition shifts toward cyanobacteria dominance in a future warmer climate. *Ecological Applications*, **20**(3), 752-767.
- Marra, J. (1997). Analysis of diel variability in chlorophyll fluorescence. *Journal of Marine Research*, **55**(4), 767-784.
- Martinez, E., Antoine, D., D'Ortenzio, F., de Boyer Montégut, C. (2011). Phytoplankton spring and fall blooms in the North Atlantic in the 1980s and 2000s. *Journal of Geophysical Research: Oceans*, **116**(C11).
- Matthews, T., Mullan, D., Wilby, R.L., Broderick, C., Murphy, C. (2016). Past and future climate change in the context of memorable seasonal extremes. *Climate Risk Management*, **11**, 37-52.
- May, L., Spears, B.M., Dudley, B.J. and Hatton-Ellis, T.W. (2009). The importance of nitrogen limitation in the restoration of Llangorse Lake, Wales, UK. *Journal of Environmental Monitoring*, **12**(1), 338-346.
- May, L. and Spears, B.M. eds., 2015. Loch Leven: 40 years of scientific research: Understanding the links between pollution, climate change and ecological response (Vol. 218). Springer.
- May, L. and Wallace, R.L. (2019). An examination of long-term ecological studies of rotifers: comparability of methods and results, insights into drivers of change and future research challenges. *Hydrobiologia*, 844(1), 129-147.
- McCarthy, V., Rafferty, P., Linnane, S., Orr, A., Flynn, R. (2012). A field study assessing the impact of on-site wastewater treatment systems on surface water quality in a Co. Monaghan catchment. In: Proceedings of international symposium on domestic wastewater treatment and disposal systems. Trinity College Dublin. http://www.epapictaural.com/nav/sWwater12.php
- McCarthy, V., Donohue, I., Irvine, K. (2006). Field evidence for stoichiometric relationships between zooplankton and N and P availability in a shallow calcareous lake. *Freshwater Biology*, **51**(9), 1589-1604.
- McCauley, E. (1984). The estimation of the abundance and biomass of zooplankton in samples. *A manual on methods for the assessment of secondary productivity in fresh waters*, **17**, 228-265.
- McConnell, D.A., Doody, D.G., Elliott, C.T., Matthews, D.I., Ferris, C.P. (2013), The impact of herbage re-growth interval on phosphorus losses in runoff post slurry application, *Agriculture, Ecosystems and Environment*, **178**, 100–108
- McDowell, R.W., Srinivasan, M.S. (2009). Identifying critical source areas for water quality:
 2. Validating the approach for phosphorus and sediment losses in grazed headwater catchments. *Journal of Hydrology*, **379**, 68-80.
- McDowell, R.W., Sharpley, A.N., Chalmers, A.T. (2002). Land use and flow regime effects on phosphorus chemical dynamics in the fluvial sediment of the Winooski River, Vermont. *Ecological Engineering*. 18, 477–487.
- McKague, K., Reid., K., Simpson, H. (2005) *Environmental Impact of Nitrogen use in Agriculture*. Ministry of Agriculture, Food and Rural Affairs (November).
- McKee, D., Atkinson, D., Collings, S.E., Eaton, J.W., Gill, A.B., Harvey, I., Hatton, K., Heyes, T., Wilson, D., Moss, B. (2003). Response of freshwater microcosm communities to nutrients, fish, and elevated temperature during winter and summer. *Limnology and Oceanography*, 48, 707–722.
- McKergow, L.A., Weaver, D.M., Prosser, I.P., Grayson, R.B., Reed, A.E. (2003). Before and after riparian management: sediment and nutrient exports from a small agricultural catchment, Western Australia. *Journal of Hydrology*, **270**(3), 253-272.

- McQueen, D.J., Johannes, M.R., Post, J.R., Stewart, T.J., Lean, D.R. (1989). Bottom-up and top-down impacts on freshwater pelagic community structure. *Ecological Monographs*, **59**(3), 289-309.
- Meals, D.W., Dressing, S.A., Davenport, T.E. (2010). Lag time in water quality response to best management practices: a review. *Journal of environmental quality*, **39**(1), 85-96.
- Meehl, G.A., Stocker, T.F., Collins, W.D., Friedlingstein, A.T., Gaye, A.T., Gregory, J.M., Kitoh, A., Knutti, R., Murphy, J.M., Noda, A. and Raper, S.C. (2007). Global climate projections.
- Melland, A.R., Jordan, P., Murphy, P.N., Mellander, P.E., Buckley, C., Shortle, G. (2014). Land use: catchment management. In: *Encyclopedia of agriculture and food systems*. (NK Van Alfen, ed.) pp. 98–113, Elsevier: San Diego, CA.
- Mellander, P. E., Melland, A. R., Jordan, P., Wall, D. P., Murphy, P. N., Shortle, G. (2012). Quantifying nutrient transfer pathways in agricultural catchments using high temporal resolution data. *Environmental Science & Policy* 24, 44-57.
- Mellard, J. P., Yoshiyama, K., Litchman, E., Klausmeier, C. A. (2011). The vertical distribution of phytoplankton in stratified water columns. *Journal of Theorethical Biology*, 269, 16-30.
- Miao, S.Y., DeLaune, R.D., Jugsujinda, A. (2006). Influence of sediment redox conditions on release/solubility of metals and nutrients in a Louisiana Mississippi River deltaic plain freshwater lake. *Science of Total Environment*, **371**, (1–3), 334–343.
- Mignot, A., Claustre, H., D'Ortenzio, F., Xing, X., Poteau, A., Ras, J. (2011). From the shape of the vertical profile of in vivo fluorescence to Chlorophyll-a concentration. *Biogeosciences*, 8(8), 2391.
- Miller, J., Chanasyk, D., Curtis, T., Entz, T., Willms, W. (2010). Influence of streambank fencing with a cattle crossing on riparian health and water quality of the Lower Little Bow River in Southern Alberta, Canada. *Agricultural Water Management* 97, 247-258.
- Miller, J., Chanasyk, D., Curtis, T., Entz, T., Willms, W. (2011). Environmental quality of Lower Little Bow River riparian zone along an unfenced reach with off-stream watering. *Agricultural Water Management* 98, 1505-1515.
- Modenutti, B., Balseiro, E., Callieri, C., Queimalin^os, C., Bertoni, R. (2004). Increase in photosynthetic efficiency as a strategy of planktonic organisms exploiting deep lake layers. *Freshwater Biology*, **49**, 160-169.
- Modenutti, B., Albariño, R., Navarro, M.B., Villanueva, V.D., Souza, M.S., Trochine, C., Laspoumaderes, C., Cuassolo, F., Mariluán, G., Buria, L., Balseiro, E. (2010). Structure and dynamic of food webs in Andean North Patagonian freshwater systems: organic matter, light and nutrient relationships. *Ecología Austral*, 20(2), 95-114.
- Monaghan County Council, (2012). Personal communication.
- Molina-Navarro, E., Andersen, H.E., Nielsen, A., Thodsen, H., Trolle, D. (2018). Quantifying the combined effects of land use and climate changes on stream flow and nutrient loads: A modelling approach in the Odense Fjord catchment (Denmark). *Science of the Total Environment*, 621, 253-264.
- Montagne, D., Cornu, S., LeForestier, L., Cousin, I. (2009). Soil drainage as an active agent of recent soil evolution: a review. *Pedosphere*, **19**(1), 1-13. Moss, B., Kosten, S., Meerhof, M., Battarbee, R., Jeppesen, E., Mazzeo, N., Havens, K., Lacerot, G., Liu, Z., De Meester, L. and Paerl, H. (2011). Allied attack: climate change and eutrophication. *Inland Waters*, **1**, 101–105.
- Moreno-Ostos, E., Cruz-Pizarro, L., Basanta, A., George, D.G. (2009). The influence of windinduced mixing on the vertical distribution of buoyant and sinking phytoplankton species. *Aquatic ecology*, **43**(2), 271-284.
- Mortensen, E. (1994). Lambertus Lijklema)-Biological structure and trophic interaction (Prof. Stephen T. Threlkeld)-Submerged macrophytes: dynamics and role in the lake ecosystem (Ass. Prof. Kaj Sand Jensen)-Comparison of freshwater and brackish lake

ecosystems (Prof. Brian Moss)-Modelling the impact of nutrients, the biological structure and the recovery process (Dr Marten Scheffer). *Hydrobiologia*, **275**, 276.

- Moss, B., Jeppesen, E., Søndergaard, M., Lauridsen, T. L., Liu, Z. (2013). Nitrogen, macrophytes, shallow lakes and nutrient limitation: resolution of a current controversy? *Hydrobiologia*, **710**, 3–21.
- Moss, B., Stephen, D., Alvarez, C. (2003). The determination of ecological status in shallow lakes – a tested system (ECOFRAME) for implementation of the European Water Framework Directive. *Aquatic Conservation-Marine and Freshwater Ecosystems*, 13, 507–549.
- Moss, R.H., Edmonds, J.A., Hibbard, K.A., Manning, M.R., Rose, S.K., Van Vuuren, D.P., Carter, T.R., Emori, S., Kainuma, M., Kram, T. and Meehl, G.A. (2010). The next generation of scenarios for climate change research and assessment. *Nature*, **463**(7282), 747.
- Moss, B., Kosten, S., Meerhoff, M., Battarbee, R.W., Jeppesen, E., Mazzeo, N., Havens, K., Lacerot, G., Liu, Z., De Meester, L. Paerl, H. (2011). Allied attack: climate change and eutrophication. *Inland waters*, **1**(2), 101-105.
- Moustaka-Gouni, M., Michaloudi, E., Sommer, U. (2014). Modifying the PEG model for Mediterranean lakes–no biological winter and strong fish predation. *Freshwater biology*, **59**(6), 1136-1144.
- National Federation of Group Water Scheme, 2016. Annual Report. Available at www.nfgws.ie
- Neal, C., Jarvie, H.P., Withers, P.J.A., Mortimer, R.J.G., Krom, M.D. (2010). The strategic importance of wastewater sources to pollutant phosphorus levels in English rivers and to environmental management for rural, agricultural and urban catchments. *Science of the Total Environment* 408, 1485–1500.
- Noges, P., Noges, T., Ghiani, M., Sena, F., Fresner, R., Friedl, M., Mildner, J. (2011). Increased nutrient loading and rapid changes in phytoplankton expected with climate change in stratified South European lakes: sensitivity of lakes with different trophic state and catchment properties. *Hydrobiologia*, **667**(1), 255-270.
- Nolan, G., Gillooly, M., Whelan, K. (2010). Irish Ocean Climate and Ecosystem Status Report 2009.
- Nürnberg, G.K. (2009). Assessing internal phosphorus load problems to be solved. Lake and Reservoir Management, **25**, 419-32.
- O'Dwyer, B., Crockford, L., Jordan, P., Hislop, L., Taylor, D. (2013). A palaeolimnological investigation into nutrient impact and recovery in an agricultural catchment. *Journal of environmental management*, **124**, 147-155.
- Olsen, S., Jeppesen, E., Moss, B., O[°] zkan, K., Beklioglu, M., Feuchtmayr, H., Sagrario, M.G., Wei, L., Larsen, S., Lauridsen T. L., Søndergaard, M. (2015a). Factors influencing nitrogen processing in lakes: an experimental approach. *Freshwater Biology*, **60**, 646–662.
- Olsen, S., Chan, F., Li, W., Zhao, S., Søndergaard, M., Jeppesen, E. (2015b). Strong impact of nitrogen loading on submerged macrophytes and algae: a long-term mesocosm experiment in a shallow Chinese lake. *Freshwater Biology*, **60**, 1525–1536
- O'Neil, J., Davis, T., Burford, M., Gobler, C. (2012). The rise of harmful cyanobacteria blooms: the potential roles of eutrophication and climate change. *Harmful Algae*, 14, 312–334.
- Padisák, J., Hajnal, E., Krienitz, L., Lakner, J., Üveges, V. (2010). Rarity, ecological memory, rate of floral change in phytoplankton—and the mystery of the Red Cock. In *Fifty years* after the 'Homage to Santa Rosalia'': Old and new paradigms on biodiversity in aquatic ecosystems (45-64). Springer, Dordrecht.
- Padisák, J., Scheffler, W., Koschel, R., Krienitz, L. (2004). Seasonal patterns and interannual variability of phytoplankton in Lake Stechlin. *Berichte des IGB*, **2004**(20), 105-116.

- Paerl, H.W., Scott, J.T., McCarthy, M.J., Newell, S.E., Gardner, W.S., Havens, K.E., Hoffman, D.K., Wilhelm, S.W., Wurtsbaugh, W.A. (2016). It takes two to tango: When and where dual nutrient (N & P) reductions are needed to protect lakes and downstream ecosystems. *Environmental Science & Technology*, **50**(20), 10805-10813.
- Paerl, H.W., Paul, V.J. (2012). Climate change: links to global expansion of harmful cyanobacteria. *Water Research*, **46**(5), 1349-1363.
- Paerl, H.W., Huisman, J. (2008). Blooms like it hot. Science, 320(5872), 57-58.
- Paerl, H.W., Fulton, R.S. (2006). *Ecology of harmful cyanobacteria*. *Ecology of Harmful Marine Algae*. Berlin, Springer-Verlag, 95-107.
- Palmer-Felgate, E.J., Jarvie, H.P., Withers, P.J.A., Mortimer, R.J.G., Krom, M.D. (2009). Stream-bed phosphorus in paired catchments with different agricultural land use intensity. *Agriculture, Ecosystems and Environment* 134, 53-66.
- Palmer-Felgate, E.J., Mortimer, R.J.G., Krom, M.D., Jarvie, H.P., Williams, R.J., Spraggs, R.E., Stratford, C.J. (2011). Internal loading of phosphorus in a sedimentation pond of a treatment wetland: effect of a phytoplankton crash. *Science of the Total Environment* 409, 2222–2232.
- Pauli, H.R. (1989). A new method to estimate individual dry weights of rotifers. *Hydrobiologia*, **186**(1), 355-361.
- Peeters, F., Straile, D., Lorke, A., Livingstone, D.M. (2007). Earlier onset of the spring phytoplankton bloom in lakes of the temperate zone in a warmer climate. *Global Change Biology*, **13**(9), 1898-1909.
- Perga, M.E., Gerdeaux, D. (2006). Seasonal variability in the δ 13 C and δ 15 N values of the zooplankton taxa in two alpine lakes. *Acta Oceologica*, **30**(1), 69-77.
- Pinker, R., Zhao, M., Wang, H., Wood, W.F. (2010). International journal of remote sensing Vol.31, No.19, 5221-5237
- Platt, T., Broomhead, D.S., Sathyendranath, S., Edwards, A.M., Murphy, E.J. (2003). May. Phytoplankton biomass and residual nitrate in the pelagic ecosystem. In *Proceedings of* the Royal Society of London A: Mathematical, Physical and Engineering Sciences 459 (2033), 1063-1073.
- Pociecha, A., Higgins, T., McCarthy, K. (2010). A preliminary study on the plankton assemblages of Lough Derg (Ireland) during a winter-spring season. *Oceanological and Hydrobiological Studies*, **39**(3), 145-154.
- Pontin, R.M. (1978). A key to the freshwater planktonic and semi-planktonic Rotifera of the British Isles (Vol. 38). Hyperion Books.
- Porter, J.H., Nagy, E., Kratz, T.K., Hanson, P., Collins, S.L., Arzberger, P. (2009). New eyes on the world: advanced sensors for ecology. *BioScience*, **59**(5), 385-397.
- Proctor, C.W., Roesler, C.S. (2010). New insights on obtaining phytoplankton concentration and composition from in situ multispectral Chlorophyll fluorescence. *Limnology and Oceanography: Methods*, 8(12), 695-708.
- Ramstack, J.M., Fritz, S.C., Engstrom, D.R., Heiskary, S.A. (2003). The application of a diatom-based transfer function to evaluate regional water-quality trends in Minnessota since 1970. *Journal of Paleolimnology*, **29**(1), 79-94.
- R Core Team. 2015. R: A language and environment for statistical computing. R Foundation for Statistical Computing, Vienna (Austria). ISBN 3-900051-07-0. <u>https://www.R-project.org/</u>.
- Rafferty, P. (2015). Investigation into the effects of on-site wastewater treatment systems on surface and subsurface water quality in a rural catchment. PhD Thesis.
- Ranalli, A.J., Macalady, D.L. (2010). The importance of the riparian zone and in-stream processes in nitrate attenuation in undisturbed and agricultural watersheds–A review of the scientific literature. *Journal of Hydrology*, **389**(3-4), 406-415.

- Read JS, Hamilton, DP, Jones ID, Muraoka K, Winslow LA, Kroiss R, Wu CH, Gaiser E. 2011. Derivation of lake mixing and stratification indices from high-resolution lake buoy data. *Environmental Modelling and Software*, 26, 1325-1336.
- Reichenberger, S., Bach, M., Skitschak, A., Frede, H. (2007). Mitigation strategies to reduce pesticides inputs into ground and surface water and their effectiveness: a review. *Science of the Total Environment*, **384**, 1-35.
- Reynolds, C.S. (1997). Vegetation processes in the pelagic: a model for ecosystem theory (Vol. 9). Oldendorf/Luhe: Ecology Institute.
- Reynolds, C.S. (2006). The ecology of phytoplankton. Cambridge University Press.
- Reynolds, C.S., Huszar, V., Kruk, C., Naselli-Flores, L., Melo, S. (2002). Towards a functional classification of the freshwater phytoplankton. *Journal of plankton research*, 24(5), 417-428.
- Reynolds, C.S. (1984). *The ecology of freshwater phytoplankton*. Cambridge University Press.
- Rigosi, A., Carey, C.C., Ibelings, B.W., Brookes, J.D. (2014). The interaction between climate warming and eutrophication to promote cyanobacteria is dependent on trophic state and varies among taxa. *Limnology Oceanography*, **59**, 99–114.
- Riley, G.A. (1942). The relationship of vertical turbulence and spring diatom flowerings. *Journal of Marine Research*, **5**(1), 67-87.
- Rimmer, A., Gal, G., Opher, T., Lechinsky, Y., Yacobi, Y.Z. (2011). Mechanisms of longterm variations in the thermal structure of a warm lake. *Limnology and Oceanography*, 56(3), 974-988.
- Roberts, W.M., Stutter, M.I., Haygarth, P.M. (2012), Phosphorus retention and remobilization in vegetated buffer strips: a review, *Journal of Environmental Quality*, **41**(2), 389-399
- Robertson, G.P., Vitousek, P.M. (2009). Nitrogen in agriculture: balancing the cost of an essential resource. *Annual Review of Environment and Resources*, **34**, 97-125.
- Rozemeijer, J.C., Klein, J., Broers, H.P., vanTol-Leenders, T.P., van der Grift, B. (2014),
 Water quality status and trends in agricultural-dominated headwaters; a national network for assessing the effectiveness of national and European manure legislation in The Netherlands, *Environmental Monitoring and Assessment*, **186**, 8981-8995
- Rueda, F., Schladow, G. (2009). Mixing and stratification in lakes of varying horizontal length scales: Scaling arguments and energy partitioning. *Limnology and Oceanography*, **54**(6), 2003-2017.
- Saunders, D.L., Kalff, J. (2001). Nitrogen retention in wetlands, lakes and rivers. *Hydrobiologia*, **443**(1), 205-212.
- Schabhüttl, S., Hingsamer, P., Weigelhofer, G., Hein, T., Weigert, A., Striebel, M. (2013). Temperature and species richness effects in phytoplankton communities. *Oecologia*, 171(2), 527-536.
- Schindler, D.W. (2006). Recent advances in the understanding and management of eutrophication. *Limnology and Oceanography* **51**, 356-363.
- Schoumans, O.F., Chardon, W.J., Bechmann, M.E., Gascuel-Odoux, C., Hofman, G., Kronvang, B., Rubæk, G.H., Ulén, B. and Dorioz, J.M. (2014). Mitigation options to reduce phosphorus losses from the agricultural sector and improve surface water quality: A review. *Science of the Total Environment*, **468**, 1255-1266.
- Schulte, R.P.O., Melland, A.R., Fenton, O., Herlihy, M., Richards, K., Jordan, P. (2010). Modelling soil phosphorus decline: Expectations of Water Framework Directives policies. *Environmental Science and Policy*, **13**, 472-484.
- Scotese, K.C. (2008). A Diatom Phosphorus Inference Model for 30 Freshwater Lakes in NE Ohio and NW Pennsylvania. PhD Thesis, Cleveland State University, Ohio, USA.
- Scourfield, D.J., Harding, J.P. (1994). British freshwater cladocera. A key. FBA scientific publication, 5.

- Seybold, C.A. (1994). Polyacrylamide review: Soil conditioning and environmental fate. *Communications in Soil Science & Plant Analysis*, **25**(11-12), 2171-2185.
- Sharpley, A.N., Daniel, T.C., Edwards, D.R. (1993). Phosphorus movement in the landscape. *Journal of Production Agriculture* **6**(4), 492-500.
- Sharpley, A., Jarvie, H.P., Buda, A., May, L., Spears, B., Kleinman, P. (2013). Phosphorus legacy: overcoming the effects of past management practices to mitigate future water quality impairment. *Journal of Environmental Quality* 42(5), 1308-1326.
- Shatwell, T., Adrian, R., Kirillin, G. (2016). Planktonic events may cause polymictic-dimictic regime shifts in temperate lakes. *Scientific reports*, **6**, 24361.
- Sheffield, R.E., Mostaghimi, S., Vaughan, D.H., Collins Jr, E.R., Allen, V.G. (1997). Offstream water sources for grazing cattle as a stream bank stabilization and water quality BMP. *Transactions of the ASAE*, 40(3), 595-604.

Shepherd, G., Buresh, R.J., Gregory, R.J. (2001). Inorganic soil nitrogen distribution in relation to soil properties in smallholder maize fields in the Kenya highlands. *Geoderma*, **101**, 87–103.

- Shore, M., Jordan, P., Mellander, P.E., Kelly-Quinn, M., Wall, D.P., Murphy, P.N.C., Melland, A.R. (2014). Evaluating the critical source area concept of phosphorus loss from soils to water-bodies in agricultural catchments. *Science of the Total Environment*, **490**, 405-415.
- Simmonds, B., Wood, S.A., Özkundakci, D., Hamilton, D.P. (2015). Phytoplankton succession and the formation of a deep chlorophyll maximum in a hypertrophic volcanic lake. *Hydrobiologia*, 745(1), 297-312.
- S.I. No. 101 of 2009. Statutory Instruments EC (Good Agricultural Practice for the Protection of Waters) Regulation 2009.
- S.I. No. 31 of 2014. Statutory Instruments EC (Good Agricultural Practice for Protection of Waters) Regulations 2014.
- S.I. No 605 of 2017, Statutory Instruments EC (Good Agricultural Practice for Protection of Waters) Regulations 2017.
- Smith, R.V., Jordan, C., Annett, J.A. (2005). A phosphorus budget for Northern Ireland: Inputs to inland and coastal waters. *Journal of Hydrology* **304**, 193-202.
- Smith, V.H., Schindler, D.W. (2009). Eutrophication science: Where do we go from here? *Trends in Ecology and Evolution* **24**, 201–207.
- Smith, V.H., Wood, S.A., McBride, C.G., Atalah, J., Hamilton, D.P., Abell, J. (2016). Phosphorus and nitrogen loading restraints are essential for successful eutrophication control of Lake Rotorua, New Zealand. *Inland Waters*, 6(2), 273-283.
- Snucins, E., John, G. (2000). Interannual variation in the thermal structure of clear and colored lakes. *Limnology and Oceanography*, **45**(7), 1639-1646.

Sommer, U., Adrian, R., De Senerpont Domis, L., Elser, J.J., Gaedke, U., Ibelings, B., Jeppesen, E., Lürling, M., Molinero, J.C., Mooij, W.M., Van Donk, E. (2012). Beyond the Plankton Ecology Group (PEG) model: mechanisms driving plankton succession. *Annual Review of Ecology, Evolution, and Systematics*, 43, 429-448.

- Søndergaard, M., Jeppesen, E., Lauridsen, T.L., Skov, C., Van Nes, E.H., Roijackers, R., Lammens, E., Portielje, R.O.B. (2007). Lake restoration: successes, failures and longterm effects. *Journal of Applied ecology*, 44(6), 1095-1105.
- Søndergaard, M., Jensen, J., Jeppesen, E. (2001). Retention and internal loading of phosphorus in shallow eutrophic lakes. *The Scientific World* **1**, 427-442.
- Søndergaard, M., Jeppesen, E., Jensen, J. (1999). Internal phosphorus loading in shallow Danish lakes. *Hydrobiologia* **408/409**, 145-152.
- Søndergaard, M., Bjerring, R., Jeppesen, E. (2013). Persistent internal phosphorus loading during summer in shallow eutrophic lakes. *Hydrobiologia*, **710**(1), 95-107.

- Søndergaard, M., Lauridsen, T.L., Johansson, L.S., Jeppesen, E. (2017). Nitrogen or phosphorus limitation in lakes and its impact on phytoplankton biomass and submerged macrophyte cover. *Hydrobiologia*, **795**(1), 35-48.
- Søndergaard, M., Jensen, J.P., Jeppesen, E. (2003). Role of sediment and internal loading of phosphorus in shallow lakes. *Hydrobiologia*, **506–509**, 135–145.
- Søndergaard, M., Jensen J.P., Jeppesen, E. (2005). Seasonal response of nutrients to reduced phosphorus loading in 12 Danish lakes. *Freshwater Biology*, **50**, 1605–1615.
- Sommer, U., Adrian, R., De Senerpont Domis, L., Elser, J.J., Gaedke, U., Ibelings, B., Jeppesen, E., Lürling, M., Molinero, J.C., Mooij, W.M., Van Donk, E. (2012). Beyond the Plankton Ecology Group (PEG) model: mechanisms driving plankton succession. *Annual Review of Ecology, Evolution, and Systematics*, 43, 429-448.
- Sommer, U., Gliwicz, Z.M., Lampert, W., Duncan, A. (1986). The PEG-model of seasonal succession of planktonic events in fresh waters. *Arch. Hydrobiol*, **106**(4), 433-471.
- Soranno, P.A., Cheruvelil, K.S., Wagner, T., Webster, K.E., Bremigan, M.T. (2015). Effects of land use on lake nutrients: the importance of scale, hydrologic connectivity, and region. *PloS one*, **10**(8), 0135454.
- Spears, B.M., Lürling, M., Yasseri, S., Castro-Castellon, A.T., Gibbs, M., Meis, S., McDonald, C., McIntosh, J., Sleep, D., Van Oosterhout, F. (2013). Lake responses following lanthanum-modified bentonite clay (Phoslock®) application: an analysis of water column lanthanum data from 16 case study lakes. *Water Research*, 47(15), 5930-5942.
- Spears, B.M., Carvalho, L., Perkins, R., Kirika, A., Paterson, D.M. (2012). Long-term variation and regulation of internal phosphorus loading in Loch Leven. *Hydrobiologia*, 681(1), 23-33.
- Standing Committee of Analysts (1983) The determination of chlorophyll a in aquatic environments 1980. HMSO.
- Stella, A. B., Sebastian, D., Herwig, S., Gabriele, T., Miriam, R., Angelika, W., Achim, W., Christoph, G. J., Maren, S. (2007). Water temperature and mixing depth affect timing and magnitude of events during spring succession of the plankton. *Oecologia*, 150(4), 643-654.
- Sterner, R.W. (2008). On the phosphorus limitation paradigm for lakes. *International Review* of Hydrobiology, **93**, 433–445.
- Stich, H.B., Brinker, A. (2010). Oligotrophication outweighs effects of global warming in a large, deep, stratified lake ecosystem. *Global Change Biology*, **16**(2), 877-888.
- Stoate, C., Idl, A.B., Beja, P., Boatman, N.D., Herzon, I., van Doorn, A., de Snoo, G.R., Rakosy, L., Ramwell, C. (2009). Ecological impacts of early 21st century agricultural change in Europe – A review. *Journal of Environmental Management* 91, 22–46.
- Stutter, M.I., Langan, S.J., Cooper, R.J. (2008). Spatial contributions of diffuse inputs and within-channel processes to the form of stream water phosphorus over storm events. *Journal of Hydrology*, **350**(3), 203-214.
- Stutter, M.I., Chardon, W.J., Kronvang, B. (2012), Riparian buffer strips as a multifunctional management tool in agricultural landscapes: introduction, *Journal of Environmental Quality*, **41**(2), 297-303
- Suggett, D.J., Prášil, O., Borowitzka, M.A. (2010). *Chlorophyll a fluorescence in aquatic sciences: methods and applications* (pp. 427-433). Dordrecht, The Netherlands: Springer.
- Sun Loh, P., Molot, L.A., Nowak, E., Nürnberg, G.K., Watson, S.B., Ginn, B. (2013). Evaluating relationships between sediment chemistry and anoxic phosphorus and iron release across three different water bodies. *Inland Waters*, 3(1), 105-118.
- Sverdrup, H.U. (1953). On vernal blooming of phytoplankton. *Conseil Exp. Mer*, **18**, 287-295.

- Sweeney, J. (2014). Regional weather and climates of the British Isles-Part 6: Ireland, *Weather*, **69**, 20-27.
- Tappin, A.D., Mankasingh, U., McKelvie, I.D., Worsfold, P.J. (2013), Temporal variability in nutrient concentrations and loads in the River Tamar and its catchment (SW England) between 1974 and 2004, *Environmental Monitoring and Assessment*, 185(6), 4791– 4818
- Taylor, D., Dalton, C., Leira, M., Jordan, P., Irvine, K., Bennion, H., McGee, E., León-Vintró, L. (2002). Identification of reference status for Irish lake typologies using palaeolimnological methods and techniques. *IN-SIGHT EPA/ERTDI Project*
- Taylor, H. M., Klepper, B. (1975). Water uptake by cotton root systems: an examination of assumptions in the single root model. *Soil Science*, **120**(1), 57-67.
- Teagasc (2011), Summer Grazing Guidelines. [Online] Available at: http://www.teagasc.ie/publications/2011/876/J3_Part2.pdf [Accessed 16 February 16]
- Teissier, S., Peretyatko, A., De Backer, S., Triest, L. (2012). Strength of phytoplankton– nutrient relationship: evidence from 13 biomanipulated ponds. *Hydrobiologia*, **689**(1), 147-159.
- Terry, J.A., Benskin, C.M.H., Eastoe, E.F. and Haygarth, P.M. (2014). Temporal dynamics between cattle in-stream presence and suspended solids in a headwater catchment. *Environmental Science: Processes & Impacts*, 16(7), 1570-1577
- Thomas, Z.P. (2002). *The effects on water quality of restricting cattle access to a Georgia Piedmont stream* (Doctoral dissertation, uga).
- Trolle, D., Nielsen, A., Rolighed, J., Thodsen, H., Andersen, H.E., Karlsson, I.B., Refsgaard, J.C., Olesen, J.E., Bolding, K., Kronvang, B., Søndergaard, M. (2015). Projecting the future ecological state of lakes in Denmark in a 6 degree warming scenario. *Climate Research*, 64(1), 55-72.
- Kragh, T., Sand-Jensen, K., Petersen, K., Kristensen, E. (2017). Fast phosphorus loss by sediment resuspension in a re-established shallow lake on former agricultural fields. *Ecological Engineering*, **108**, 2-9.
- Ulen, B.M., Bechmann, M., Folster, J., Jarvie, H.P., Tunney, H. (2007). Agriculture as a phosphorus source for eutrophication in the northwest European countries, Norway, Sweden, United Kingdom, Ireland. *Soil Use and Management* **23**, 5-15.
- Vaccari, D.A. (2009). Phosphorus: a looming crisis. *Scientific American* **300**, 54-59.
- Van den Pol-van Dasselaar, A., Aarts, H., De Vliegher, A., Elgersma, A., Reheul, D., Reijneveld, J., Verloop, J., Hopkins, A. (2015). Grassland and forages in high output dairy farming systems. *Grassland Science in Europe, Wageningen Academic Publishers, Wageningen, the Netherlands.*
- Van Drecht, G., Bouwman, A.F., Harrison, J., Knoop, J.M. (2009). Global nitrogen and phosphate in urban wastewater for the period 1970 to 2050. *Global Biogeochem*. Cycles 23, GB0A03, http://dx.doi.org/10.1029/2009gb003458.
- Vanni, M.J. (2010). Preface: when and where do fish have strong effects on stream ecosystem processes. In *American Fisheries Society Symposium* (Vol. 73, pp. 531-538).
- VanRensburg, T.M., Murphy, E., Rocks, P. (2009). Commonage land and farmer uptake of the rural environment protection scheme in Ireland. *Land use and Policy*, **26**, 345-355.
- Varela, R. A., Cruzado, A., Tintore, J. (1994). A simulation analysis of various biological and physical factors influencing the deep-chlorophyll maximum structure in oligotrophic areas. *Journal of Marine Systems*, 5, 143-157.
- Velthof, G.L., Oudendag, D., Witzke, H.P., Asman, W.A.H., Klimont, Z., Oenema, O. (2009). Integrated assessment of nitrogen losses from agriculture in EU-27 using MITERRAEUROPE. *Journal of Environmental Quality*, **38**(2), 402-417.
- Venohr, M., Donohue, I., Fogelberg, S., Arheimer, B., Irvine, K., Behrendt, H. (2005). Nitrogen retention in a river system and the effects of river morphology and lakes. *Water Science & Technology*, **51**(3-4), 19-29.

- Verburg, P., Hecky, R.E., Kling, H. (2003). Ecological consequences of a century of warming in Lake Tanganyika. *Science*, **301**(5632), 505-507.
- Verma, S., Bhattarai, R., Bosch, N.S., Cooke, R.C., Kalita, P.K., Markus, M. (2015). Climate change impacts on flow, sediment and nutrient export in a Great Lakes watershed using SWAT. *CLEAN–Soil, Air, Water*, 43(11), 1464-1474.
- Verpoorter, C., Kutser, T., Seekell, D.A., Tranvik, L.J. (2014). A global inventory of lakes based on high-resolution satellite imagery. *Geophysical Research Letters*, **41**(18), 6396-6402.
- Vidon, P., Campbell, M.A., Gray, M. (2008). Unrestricted cattle access to streams and water quality in till landscape of the Midwest. *Agricultural Water Management*, 95(3), 322-330.
- Vidon, P., Allan, C., Burns, D., Duval, T.P., Gurwick, N., Inamdar, S., Lowrance, R., Okay, J., Scott, D., Sebestyen, S. (2010). Hot spots and hot moments in riparian zones: potential for improved water quality management. *JAWRA Journal of the American Water Resources Association*, 46(2), 278-298.
- Vidon, P., Hubbard, H., Cuadra, P., Hennessy, M. (2012). Storm phosphorus concentrations and fluxes in artificially drained landscapes of the US Midwest. *Agricultural Sciences*, 3(4), 474-485.
- Vollenweider, R.A. (1989). Global problems of eutrophication and its control. In *Symp. Biol. Hung* (Vol. **38**, 19-41).
- Wagner, C., Adrian, R. (2009). Cyanobacteria dominance: quantifying the effects of climate change. *Limnology and Oceanography*, **54**, 2460-2468.
- Wall, D., Jordan, P., Melland, A.R., Mellander, P.E., Buckley, C., Reaney, S.M., Shortle, G. (2011). Using the nutrient transfer continuum concept to evaluate the European Union Nitrates Directive National Action Programme. *Environmental Science & Policy*, 14(6), 664-674.
- Wang, L., Flanagan, D.C., Wang, Z., Cherkauer, K.A. (2018). Climate Change Impacts on Nutrient Losses of Two Watersheds in the Great Lakes Region. *Water*, 10(4), 442.
- Wauer, G., Gonsiorczyk, T., Casper, P., Koschel, R. (2005). P-immobilisation and phosphatase activities in lake sediment following treatment with nitrate and iron. *Limnologica-Ecology and Management of Inland Waters*, **35**(1), 102-108.
- Wentz, F.J., Ricciardulli, L., Hilburn, K., Mears, C. (2007). How much more rain will global warming bring? *Science*, **317**(5835), 233-235.
- Wetzel, R.G.: Limnology: lake and river ecosystems, 3rd Edn., Academic Press, San Diego, California, 2001.
- White, B., Matsumoto, K. (2012). Causal mechanisms of the deep chlorophyll maximum in Lake Superior: A numerical modelling investigation. *Journal of Great Lakes Research*, 38, 504-513.
- Whittington, J., Sherman, B., Green, D. and Oliver, R. L. (2000). Growth of *Ceratium hirundinella* in a subtropical Australian reservoir: the role of vertical migration. *Journal of Plankton Research*, 22, 1025–1045. Whitton and Brooke (2003). The Freshwater Algal Flora of the British Isle: An Identification Guide to Freshwater and Terrestrial Algae, Volume 1.
- Wiederholt, R., Johnson, B. (2005). *Phosphorus behaviour in the environment*. North Dakota State, University.
- Wilkinson, S.R., Stuedemann, J.A., Belesky, D.P. (1989). Soil potassium distribution in grazed K-31 tall fescue pastures as affected by fertilization and endophytic fungus infection level. *Agronomy journal*, 81(3), 508-512.
- Williams, G.P. (1989). Sediment concentration versus water discharge during single hydrologic events in rivers. *Journal of Hydrology*, **111**(1-4), 89-106.
- Williamson, C.E., Fischer, J.M., Bollens, S.M., Overholt, E.P., Breckenridge, J.K. (2011). Toward a more comprehensive theory of zooplankton diel vertical migration:

integrating ultraviolet radiation and water transparency into the biotic paradigm. *Limnology and Oceanography*, **56**(5), 1603-1623.

- Winder, M., Sommer, U. (2012). Phytoplankton response to a changing climate. *Hydrobiologia*, **698**(1), 5-16.
- Withers, P.J.A., Edwards, A.C., Foy, R.H. (2001). Phosphorus cycling in UK agriculture and implications for phosphorus loss from soil. *Soil Use and Management* **17** (3), 139-149.
- Withers, P.J.A., Hodgkinson, R.A. (2009). The effect of farming practices on phosphorus transfer to a headwater streams in England. *Agriculture, Ecosystems and Environment* 131, 347-355.
- Wood, S.N. (2006). Generalized Additive Models: An Introduction with R. Chapman and Hall/CRC.
- Woolway, R.I., Maberly, S.C., Jones, I.D., Feuchtmayr, H. (2014). A novel method for estimating the onset of thermal stratification in lakes from surface water measurements. *Water Resources Research*, **50**(6), 5131-5140.
- Woolway, R.I., Meinson, P., Nõges, P., Jones, I.D., Laas, A. (2017). Atmospheric stilling leads to prolonged thermal stratification in a large shallow polymictic lake. *Climatic Change*, **141**(4), 759-773.
- Woolway, R.I., Weyhenmeyer, G.A., Schmid, M., Dokulil, M.T., de Eyto, E., Maberly, S.C., May, L. and Merchant, C.J. (2019). Substantial increase in minimum lake surface temperatures under climate change. *Climatic Change*, 1-14.
- Wu, Y., Wen, Y., Zhou, J., Wu, Y. (2014). Phosphorus release from lake sediments: effects of pH, temperature and dissolved oxygen. *KSCE Journal of Civil Engineering*, 18(1), 323-329.Wu, N., Schmalz, B., Fohrer, N. (2010). Distribution of phytoplankton in a German lowland river in relation to environmental factors. *Journal of Plankton Research*, 33(5), 807-820.
- Wu, Y., Xiang, Y., Wang, J., Zhong, J., He, J., Wu, Q.L. (2010). Heterogeneity of archaeal and bacterial ammonia-oxidizing communities in Lake Taihu, China. *Environmental Microbiology Reports*, 2(4), 569-576.
- Xu, H., Paerl, H.W., Qin, B.Q., Zhu, G.W., Gao, G. (2010). Nitrogen and phosphorus inputs control phytoplankton growth in eutrophic Lake Taihu, China. *Limnology & Oceanography*, 55, 420–432.
- Yan, Z., Han, W., Peñuelas, J., Sardans, J., Elser, J.J., Du, E., Reich, P.B., Fang, J. (2016). Phosphorus accumulates faster than nitrogen globally in freshwater ecosystems under anthropogenic impacts. *Ecology letters*, **19**(10), 1237-1246.
- Yang, J.Y., Drury, C.F., De Jong, R., Huffman, E.C., Yang, X.M., Reid, K. (2013). Sensitivity analysis for nitrogen inputs, nitrogen outputs, and changes in biofuel crop acreages for predicting residual soil nitrogen and nitrate leaching in Canadian agricultural soils. *Ecological Modelling*, 267, 26-38.
- Yu, Q., Wang, H.Z., Li, Y., Shao, J.C., Liang, X.M., Jeppesen, E., Wang, H.J. (2015). Effects of high nitrogen concentrations on the growth of submersed macrophytes at moderate phosphorus concentrations. *Water Research*, **83**, 385–395.
- Vystavna, Y., Hejzlar, J., Kopáček, J. (2017). Long-term trends of phosphorus concentrations in an artificial lake: Socio-economic and climate drivers. *PloS one*, **12**(10), 0186917.Zaied, K.A., El-Mageed, H.A., Fayzalla, E.A., Sharief, A.E., Zehry, A.A. (2009). Enhancement assimilation of nitrite and nitrate containing factory effluents via recombinants induced in Yeast and Bacteria. *Australian Journal of Basic and Applied Sciences*, **3**(1), 17-27
- Zamparas, M. and Zacharias, I. (2014). Restoration of eutrophic freshwater by managing internal nutrient loads. A review. *Science of the Total Environment*, **496**, 551-562.
- Zhu, G., Wang, S., Wang, W., Wang, Y., Zhou, L., Jiang, B., Den Camp, H.J.O., Risgaard-Petersen, N., Schwark, L., Peng, Y., Hefting, M.M. (2013). Hotspots of anaerobic ammonium oxidation at land-freshwater interfaces. *Nature Geoscience*, 6(2), 103.

Zhu, W., Wan, L., Zhao, L. (2010). Effect of nutrient level on phytoplankton community structure in different water bodies. *Journal of Environmental Sciences*, **22**(1), 32-39.

Zuur, A.F., Ieno, E.N., Smith, G.M. (2007), Analysing Ecological Data, New York: Springer

Zuur, A.F., Ieno, E.N., Walker, N.J., Saveliev, A., Smith, G.M. (2009). Mixed Effects Models and Extensions in Ecology with R. New York (NY). Springer. 987-990.